STATION CASINOS INC

Form PRE 14A
March 28, 2005
UNITED STATES
SECURITIES AND EXCHANGE COMMISSION
Washington, D.C. 20549

SCHEDULE 14A INFORMATION

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934

Filed by the Registrant x

Filed by a Party other than the Registrant O

Check the appropriate box:

x Preliminary Proxy Statement

o Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))

o Definitive Proxy Statement o Definitive Additional Materials

o Soliciting Material Pursuant to §240.14a-12

STATION CASINOS, INC.

(Name of Registrant as Specified In Its Charter)

(Name of Person(s) Filing Proxy Statement, if other than the Registrant)

Payment of Filing Fee (Check the appropriate box):

x No fee required.

0

Fee computed on table below per Exchange Act Rules 14a 6(i)(4) and 0-11.

(1) Title of each class of securities to which transaction applies:

(2) Aggregate number of securities to which transaction applies:

(3) Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set

forth the amount on which the filing fee is calculated and state how it was determined):

(4) Proposed maximum aggregate value of transaction:

(5) Total fee paid:

o Fee paid previously with preliminary materials.

Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and

the date of its filing.

(1) Amount Previously Paid:

(2) Form, Schedule or Registration Statement No.:

(3) Filing Party:

(4) Date Filed:

Persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB control number.

STATION CASINOS, INC.

2411 West Sahara Avenue Las Vegas, Nevada (702) 367-2411	
NOTICE OF ANNUAL MEETING OF STOCKHOLDERS To Be Held: May 18, 2005 To Be Held At: Green Valley Ranch Station Casino	

Dear Fellow Stockholder:

You are cordially invited to attend the Annual Meeting of Stockholders of Station Casinos, Inc. to be held on May 18, 2005, at Green Valley Ranch Station Casino, 2300 Paseo Verde, Henderson, Nevada at 10:00 a.m. local time. Your Board of Directors and management look forward to greeting personally those stockholders able to attend.

The attached Notice of Annual Meeting and Proxy Statement describe the formal business to be transacted at the meeting including: the election of two directors to serve for a three year term; approval of the Company s 2005 Stock Compensation Program; and ratification of the appointment of the Company s independent auditors. In addition, a dissident union stockholder, UNITE HERE, an affiliate of the Culinary Workers Union Local 226, has informed your Board that it intends to solicit proxies for three non-binding stockholder proposals at the meeting. The Board urges you to reject the union s self-serving solicitation. Please discard any proxy card the union may send to you.

YOUR VOTE IS IMPORTANT

Your vote is important regardless of the number of shares you own. We urge you to sign, date and mail the Company s white proxy card as soon as possible even if you currently plan to attend the annual meeting. This will not prevent you from voting in person but will assure that your vote is counted if you are unable to attend the meeting. To make certain that your vote is cast on all matters before the meeting, please be certain only to vote the Company s white card or the Company s form of proxy sent to you by your bank or broker.

You are assured that your Board of Directors will continue to act in the best interest of all Company Stockholders.

Thank you for your continued interest and support.

Sincerely, Richard J. Haskins Secretary

IMPORTANT

YOUR VOTE IS IMPORTANT. PLEASE SIGN, DATE AND MAIL YOUR WHITE PROXY CARD AT YOUR EARLIEST CONVENIENCE. NO POSTAGE IS REQUIRED IF MAILED IN THE UNITED STATES. STOCKHOLDERS WITH QUESTIONS OR REQUIRING ASSISTANCE MAY CALL D. F. KING & CO., INC., WHICH IS ASSISTING YOUR COMPANY, TOLL-FREE AT 1-888-887-0082.

STATION CASINOS, INC.
2411 West Sahara Avenue
Las Vegas, Nevada 89102
PROXY STATEMENT
The accompanying proxy is solicited by the board of directors (the Board of Directors) of Station Casinos, Inc., (the Company) to be used at the Annual Meeting of Stockholders on May 18, 2005 (the Annual Meeting) to be held a 10:00 a.m. local time at Green Valley Ranch Station Casino, 2300 Paseo Verde, Henderson, Nevada. This Proxy Statement and the enclosed form of proxy are being sent to stockholders on or about April , 2005.
At the Annual Meeting, stockholders will be asked to consider and vote upon the following matters:
The election of two directors to serve until the 2008 Annual Meeting.
ITEM II Approval of the Company s 2005 Stock Compensation Program.
ITEM III A proposal to ratify the appointment of Ernst & Young LLP as the Company s independent auditors for 2005.
ITEM IV A stockholder proposal from a single stockholder, UNITE HERE, an affiliate of the Culinary Workers Union Local 226, which is a labor union that is attempting to organize our employees (the Union), to urge the Board to lower the voting requirements to amend the Company's Restated Bylaws from the current supermajority to a simple majority.
ITEM V A stockholder proposal from the Union to urge the Board to declassify the Board of Directors and institute annual election of directors.
ITEM VI A stockholder proposal from the Union to urge the Board to submit the Company s Rights Agreement to stockholders for approval.
Any stockholder giving a proxy may revoke it at any time prior to its exercise at the Annual Meeting by giving notice of such revocation either personally or in writing to the Secretary of the Company at the Company s executive offices, by subsequently executing and delivering another proxy or by voting in person at the Annual Meeting.
Your Board of Directors believes that the election of its director nominees, the approval of the Company s 2005 Stock Compensation Program and the ratification of Ernst & Young LLP as the Company s independent auditors for 2005 are in the best interests of the Company and its stockholders. Therefore, your Board of Directors unanimously recommends voting FOR each of the proposals contained in items I, II and III.
The Board of Directors believes that lowering the voting requirement to amend the Company s Restated Bylaws from the current supermajority to a simple majority, that declassifying the Board of Directors and that submitting the Company s Rights Agreement to stockholders for approval are not in the best interest of the Company and its stockholders. Therefore, your Board of Directors unanimously recommends voting AGAINST each of the Union s proposals contained in items IV, V and VI.

VOTING

Shares represented by duly executed and unrevoked proxies in the form of the enclosed white proxy card received by the Board of Directors will be voted at the Annual Meeting in accordance with the specifications made therein by the stockholders, unless authority to do so is withheld. If no specification is made, shares represented by duly executed and unrevoked proxies in the form of the enclosed white proxy card will be voted FOR the election as directors of the nominees listed herein, FOR the approval of the Company s 2005 Stock Compensation Program, FOR the ratification of Ernst & Young LLP as the Company s independent auditors for 2005, AGAINST urging the Board to amend the Company s Restated Bylaws from the current supermajority to a simple majority, AGAINST urging the Board to declassify the Board of Directors, AGAINST urging the Board to submit the Company s Rights Agreement to stockholders for approval and, in the discretion of the persons voting the respective proxies with respect to any other matter that may properly come before the Annual Meeting.

The cost of preparing, assembling and mailing of proxy materials will be borne by the Company. Directors, executive officers and other employees may also solicit proxies but without receiving special compensation. Brokerage houses, nominees, fiduciaries and other custodians will be requested to forward soliciting materials to beneficial owners and will be reimbursed for their reasonable out-of-pocket expenses incurred in sending proxy materials to beneficial owners. The Company has retained D. F. King & Co., Inc. to assist in the solicitation of proxies for an estimated fee of \$\\$, plus reimbursement of certain out-of-pocket expenses.

Only holders of record at the close of business on March 24, 2005 (the Record Date) of the Company s common stock, \$0.01 par value, will be entitled to vote at the Annual Meeting. On the Record Date, there were , , shares of common stock outstanding. Each share of common stock is entitled to one vote on all matters presented at the Annual Meeting.

Vote Required

The election of the director nominees (Item I) requires a plurality of the votes cast in person or by proxy at the Annual Meeting. Approval of the Company s 2005 Stock Compensation Program (Item II) requires the affirmative vote, cast in person or by proxy, of holders of a majority of the outstanding shares. Each of the proposal to ratify Ernst & Young LLP as the Company s independent auditors for 2005 (Item III) and the three stockholder proposals (Items IV, V and VI) requires the affirmative vote of holders of a majority of the voting power represented in person or by proxy and entitled to vote at the Annual Meeting. Under Nevada law and the Company s Restated Bylaws (the Bylaws), shares as to which a stockholder abstains or withholds from voting and shares to which a broker indicates that it does not have discretionary authority to vote (broker non-votes) will be counted as voting thereon and, therefore, will have the same legal effect as a vote against a proposal; however, such shares will have no effect on the election of directors.

The stockholders of the Company have no dissenters or appraisal rights in connection with any of the items to be voted upon at the meeting.

ITEM I

NOMINEES FOR ELECTION OF DIRECTORS

The Articles and Bylaws require that the number of directors on the Board of Directors be not less than three (3) nor more than fifteen (15). Currently, the Board of Directors has fixed the number of directors at six (6). The Board of Directors presently consists of the following persons: Frank J. Fertitta III, Lorenzo J. Fertitta, Lowell H. Lebermann, Jr., James E. Nave D.V.M., Lee S. Isgur and Robert E. Lewis. Mr. Lewis was nominated on May 19, 2004 by the members of the Board of Directors to fill the vacancy created by the resignation of Lynette Boggs McDonald on March 30, 2004. Mr. Lewis s nomination was

recommended by the Company s management and approved by the Governance and Compensation Committee and the Board of Directors. On December 1, 2004, Blake L. Sartini resigned from the Board of Directors. The Board of Directors is divided into three classes. Class I consists of Lowell H. Lebermann, Jr, and Robert E. Lewis whose terms expires in 2006. Class II consists of Lorenzo J. Fertitta and James E. Nave, D.V.M., whose terms expire in 2007. Class III consists of Frank J. Fertitta III and Lee S. Isgur, whose terms expire in 2005. At each annual meeting, the terms of one class of directors expire. Each director nominee is elected to the Board of Directors for a term of three years.

At the Annual Meeting, two directors are to be elected to serve until the 2008 Annual Meeting and until their successors are elected and qualified. Unless authority to vote for directors is withheld in the proxy card, it is the intention of the persons named in the enclosed form of proxy to vote **FOR** the election of the two nominees listed below. The persons designated as proxies will have discretion to cast votes for other persons in the event any nominee for director is unable to serve. At present, it is not anticipated that any nominee will be unable to serve.

The names and certain information concerning the persons to be nominated as directors by the Board of Directors at the Annual Meeting are set forth below. THE BOARD OF DIRECTORS RECOMMENDS THAT YOU VOTE FOR THE ELECTION OF THE NOMINEES LISTED BELOW.

Frank J. Fertitta III. Mr. Fertitta has served as Chairman of the Board of the Company since February 1993 and Chief Executive Officer since July 1992. Mr. Fertitta also served as President of the Company from 1989 until July 2000. He has held senior management positions since 1985, when he was named General Manager of Palace Station. He was elected a director of the Company in 1986, at which time he was also appointed Executive Vice President and Chief Operating Officer.

Lee S. Isgur. Mr. Isgur has served as a director of the Company since December 2003 and serves on the Audit and Governance and Compensation Committees. Mr. Isgur has been the Managing Partner of Corporate Counselors, a research and investment banking consulting firm, since 1997 and has served on the board of directors of eDiets since 1999, where he chairs the Audit and Compensation Committees. From 1994 to 1997, Mr. Isgur was a Managing Director of Jefferies & Company, an investment banking firm. Prior to that, he was a partner at Volpe Welty & Company, a research and investment banking firm.

DIRECTORS AND EXECUTIVE OFFICERS

The following table sets forth the directors and executive officers of the Company. All directors hold their positions until their terms expire and until their respective successors are elected and qualified. Executive officers are elected by and serve at the discretion of the Board of Directors.

Name	Age	Position
Frank J. Fertitta III(*)	43	Chairman of the Board and Chief Executive Officer
Lorenzo J. Fertitta(*)	36	Vice Chairman of the Board and President
Glenn C. Christenson	55	Executive Vice President, Chief Financial Officer, Chief
		Administrative Officer and Treasurer
Richard J. Haskins	41	Executive Vice President, General Counsel and Secretary
Scott M Nielson	47	Executive Vice President and Chief Development Officer
William W. Warner	40	Executive Vice President and Chief Operating Officer
Lee S. Isgur	67	Director
Lowell H. Lebermann, Jr	65	Director
Robert E. Lewis	59	Director
James E. Nave, D.V.M.	60	Director

^(*) Frank J. Fertitta III and Lorenzo J. Fertitta are brothers.

Set forth below are the Class I and Class II directors whose terms do not expire this year together with non-director executive officers of the Company, along with certain information regarding these individuals.

Lorenzo J. Fertitta. Mr. Fertitta was appointed Vice Chairman of the Board of the Company in December 2003. Mr. Fertitta has served as President of the Company since July 2000 and has served as a director since 1991. Mr. Fertitta served as President and Chief Executive Officer of Fertitta Enterprises, Inc. from June 1993 to July 2000, where he was responsible for managing an investment portfolio consisting of marketable securities and real property. From 1991 to 1993, he served as Vice President of the Company. Mr. Fertitta currently serves as Chairman of the Nevada Resort Association and serves as a director of the American Gaming Association. Mr. Fertitta served as a commissioner on the Nevada State Athletic Commission from November 1996 until July 2000.

Lowell H. Lebermann, Jr. Mr. Lebermann has served as a director of the Company since October 1993 and is Chairman of the Governance and Compensation Committee and serves on the Audit Committee. He is Chairman of Centex Beverage, Inc., a wholesale distributor of Miller beer and imported beverages. He is a director of Patton Medical, Inc. and DoggettData, Inc., of Austin, Texas. He is a former director of Valero Energy Corporation, San Antonio and of Franklin Federal Bancorp, Austin (acquired by Norwest) and founding member of the board of directors of the Texas Workers Compensation Fund. From 1993 to 1999, he was a member of the Board of Regents of The University of Texas System. He was a Council Member on the Austin City Council from 1971 to 1977.

Robert E. Lewis. Mr. Lewis has served as a director of the Company since May 2004 and serves on the Audit and Governance and Compensation Committees. Mr. Lewis has served as president of the Nevada Division of Lewis Operating Corp. since December 1999. Mr. Lewis became the president of the Nevada Region of Kaufman and Broad Home Corporation upon the merger of Lewis Homes Management Corp. and Kaufman and Broad Home Corporation in January 1999. He served in that capacity until December 1999. Prior to the merger, Mr. Lewis ran the Nevada operation of the Lewis Homes group of companies and its affiliates for 25 years. He has served as a director for the National Association of Home Builders and as a director and President of the Southern Nevada Home Builders Association. Mr. Lewis is also on the Executive Committee and served as Secretary and Legislative Chairman for the Nevada Development Authority, he is a Past Chairman of the Las Vegas District Council of the Urban Land Institute and serves on the Clark County Community Growth Task Force.

James E. Nave, D.V.M. Dr. Nave has served as a director of the Company since March 2001, is chairman of the Audit Committee and serves on the Governance and Compensation Committee. Dr. Nave has been an owner of the Tropicana Animal Hospital since 1974 and has been the owner and manager of multiple veterinary hospitals since 1976. Dr. Nave has also served on the board of directors of Bank West of Nevada since 1994, where he also serves as Chairman of the Site Committee. Dr Nave has served on the board of directors of Western Alliance Bancorporation since 2003, where he also serves as Chairman of the Investment Committee and as a member of the Audit Committee. Dr. Nave is also the Globalization Liaison Agent for Education and Licensing for the American Veterinary Medical Association and is Chairperson of the National Commission for Veterinary Economics Issues. In addition, Dr. Nave is a member and past President of the Nevada Veterinary Medical Association, the Western Veterinary Conference and the American Veterinary Medical Association. He is also a member of the Clark County Veterinary Medical Association, the National Academy of Practitioners, the American Animal Hospital Association and the Executive Board of the World Veterinary Association. Dr. Nave was a member of the University of Missouri, College of Veterinary Medicine Development Committee from 1984 to 1992. He was also a member of the Nevada State Athletic Commission from 1988 to 1999 and served as its chairman from 1989 to 1992 and from 1994 to 1996.

Glenn C. Christenson. Mr. Christenson was appointed Chief Administrative Officer in March 1997 and has served as Executive Vice President of the Company since February 1994. From 1989 to 1993, he served as Vice President of the Company. He has served as Chief Financial Officer since 1989, as Treasurer since 1992 and as a director of the Company from 1993 to December 2003. Mr. Christenson is a Certified Public Accountant. From 1983 to 1989, he was a partner of the international accounting firm of Deloitte Haskins & Sells (now Deloitte & Touche), where he served as partner-in-charge of audit services for the Nevada practice and National Audit partner for the Hospitality Industry. Mr. Christenson has served on the board of directors of the Nevada Resort Association and was Chairman of the Nevada Resort Association s IRS Liaison Committee. He currently serves as a director of Nevada Community Bank and is a member of its Audit Committee. He is a member of the Board of Trustees of the Nevada Development Authority and is a member of the board of directors for the Nevada State College Foundation Board, the National Center for Responsible Gaming and the Problem Gambling Center.

Scott M Nielson. Mr. Nielson was appointed Chief Development Officer of the Company in July 2004 and has been an Executive Vice President of the Company since June 1994. He served as Chief Legal Officer from March 2002 to July 2004 and General Counsel from 1991 to March 2002. In 1992, he was appointed Secretary of the Company and served in that position until July 2004. From 1991 through June 1994, he served as Vice President of the Company. From 1986 to 1991, Mr. Nielson was in private legal practice as a partner in the Las Vegas firm of Schreck, Jones, Bernhard, Woloson & Godfrey (now Schreck Brignone), where he specialized in gaming law and land use planning and zoning. Mr. Nielson is a member of the American Bar Association, the Nevada Bar Association and the International Association of Gaming Attorneys.

William W. Warner. Mr. Warner was appointed Chief Operating Officer of the Company in July 2004 and has been an Executive Vice President of the Company since March 2002. He served as Chief Development Officer from March 2002 to July 2004, Vice President of Finance of the Company from January 1996 to March 2002 and Director of Finance from August 1993 to January 1996.

Richard J. Haskins. Mr. Haskins was appointed Executive Vice President and Secretary of the Company in July 2004 and has served as General Counsel of the Company since April 2002. He previously served as Assistant Secretary from September 2003 to July 2004, as Vice President and Associate General Counsel from November 1998 to March 2002, and as General Counsel of Midwest Operations from November 1995 to October 1998. From 1990 to October 1995, Mr. Haskins was in private legal practice, most recently as a partner in the Kansas City, Missouri law firm of Rose Brouillette & Shapiro P. C. Mr. Haskins is a member of the American Bar Association, the Kansas Bar Association, the Missouri Bar Association and the Nevada Bar Association.

Meetings of the Board of Directors

The Board of Directors met 12 times during 2004. The Board of Directors has a standing Audit Committee and Governance and Compensation Committee. Each committee is comprised solely of independent directors. None of the members of the Board of Directors attended less than 75% of the meetings of the Board of Directors held, or of the total number of meetings held by all committees of the Board of Directors on which various members served during 2004. On December 1, 2004, Blake L. Sartini resigned as a member of the Board of Directors. The current members of each of the Board of Directors committees are listed below.

All of the members of the Board of Directors were in attendance at the 2004 annual stockholders meeting, which included Frank J. Fertitta III, Lorenzo J. Fertitta, Dr. James E. Nave, Lowell H. Lebermann and Lee S. Isgur. It is the Company s policy that all members of the Board of Directors are expected to attend the annual stockholders meeting.

Director Independence

The Board of Directors has determined that each of Messrs. Lebermann, Nave, Isgur and Lewis are independent directors (as independence is defined in Sections 303.01 (B)(2)(a) and 3 of the New York Stock Exchange (NYSE) Listing Standards). As part of its analysis, the Board of Directors determined that none of Messrs. Lebermann, Nave, Isgur or Lewis has a direct or indirect material relationship with the Company.

Executive Sessions of Non-Management Directors

In order to promote open discussion among the non-management directors, the Board of Directors intends to schedule executive sessions, at least four times each year, in which those directors meet without management participation. The Chairman of each executive session will rotate between the chairpersons of the Company s Audit Committee and Governance and Compensation Committee, depending upon the primary subject matter of the meeting. During 2004, the non-management directors met 6 times without management participation following Audit Committee or Compensation and Governance Committee meetings.

The Audit Committee

The Board of Directors has a separately designated standing Audit Committee that was established in accordance with Section 3(a) (58) (A) of the Securities Exchange Act of 1934. The current members of the Audit Committee are James E. Nave, D.V.M., Chairman, Lowell H. Lebermann, Jr., Lee S. Isgur and Robert E. Lewis. Ms. Boggs McDonald served as a member of the Audit Committee prior to her resignation from the Board of Directors in March 2004. During 2004, the Audit Committee met 9 times.

The Audit Committee, comprised solely of independent directors (as independence is defined in Sections 303.01(B)(2)(a) and (3) of the NYSE listing standards), meets periodically with the Company s independent auditors, management, internal auditors and legal counsel to discuss accounting principles, financial and accounting controls, the scope of the annual audit, internal controls, regulatory compliance and other matters. The Audit Committee also advises the Board of Directors on matters related to accounting and auditing and selects the Company s independent auditors. The independent auditors and the internal auditors have complete access to the Audit Committee without management present to discuss results of their audit and their opinions on adequacy of internal controls, quality of financial reporting and other accounting and auditing matters. The responsibilities of the Audit Committee are outlined in a written charter, which is available on the Company s website at www.stationcasinos.com. Printed copies are also available upon request directed to the Company s Corporate Secretary.

Audit Committee Financial Expert

The Board of Directors has determined that all Audit Committee members are financially literate under the current listing standards of the NYSE. The Board of Directors also determined that Lee S. Isgur qualifies as an audit committee financial expert as defined by the Securities and Exchange Commission rules adopted pursuant to the Sarbanes-Oxley Act of 2002.

The Governance and Compensation Committee

In April 2004, the Board of Directors changed the name of the Human Resources Committee to the Governance and Compensation Committee and expanded the responsibilities of the Committee to include matters relating to the nomination of directors and corporate governance policies in addition to compensation matters. The current members of the Governance and Compensation Committee are Lowell H. Lebermann, Jr., Chairman, James E. Nave, D.V.M., Lee S. Isgur and Robert E. Lewis. Ms. Boggs McDonald served as a member of the Governance and Compensation Committee prior to her resignation from the Board of Directors in March 2004.

The Governance and Compensation Committee, currently comprised solely of independent directors (as independence is defined in Sections 303.01(B)(2)(a) and (3) of the NYSE listing standards), reviews and takes action regarding terms of compensation, employment contracts and pension matters that concern officers and key employees of the Company. The Governance and Compensation Committee also reviews and takes action regarding grants of stock options and restricted shares to employees. The Governance and Compensation Committee is also responsible for matters concerning the nomination of directors and corporate governance policies. The Governance and Compensation Committee also evaluates and recommends candidates for election as directors, makes recommendations concerning the size and composition of the Board of Directors, implements the Company s corporate governance policies and assesses the effectiveness of the Board of Directors. The Governance and Compensation Committee met 17 times during 2004. The responsibilities of the Governance and Compensation Committee are outlined in a written charter, which is available on the Company s website at www.stationcasinos.com. Printed copies are also available upon request directed to the Company s Corporate Secretary.

In evaluating and determining whether to nominate a candidate for a position on the Company s Board of Directors, the Committee will consider the criteria outlined in the Company s corporate governance policy, which include high professional ethics and values, relevant management experience and a commitment to enhancing stockholder value. In evaluating candidates for nomination, the Committee utilizes a variety of methods. The Company regularly assesses the size of the Board of Directors, whether any vacancies are expected due to retirement or otherwise, and the need for particular expertise on the Board of Directors. Candidates may come to the attention of the Committee from current Board of Directors members, stockholders, professional search firms, officers or other persons. The Committee will review all candidates in the same manner regardless of the source of the recommendation. The Committee will consider stockholder recommendations of candidates when the recommendations are properly submitted. Any stockholder recommendations which are submitted under the criteria summarized above should include the candidate s name and qualifications for Board of Directors membership, which must be submitted no later than ninety (90) days prior to the anniversary of the prior annual meeting and should be addressed to:

Corporate Secretary Station Casinos, Inc. 2411 West Sahara Avenue Las Vegas, Nevada 89102

The notice must include all of the information required by the Bylaws, including but not limited to the candidate s name, age, business address, residence address, principal occupation or employment, the number of shares of the Company s common stock beneficially owned by the candidate and information that would be required to solicit a proxy under federal securities law.

Compensation of Directors

Directors who are not directly or indirectly affiliated with the Company receive a monthly fee of \$4,167, a fee of \$1,500 for each board meeting attended, \$1,500 for each committee meeting attended and the committee chairman receives an additional \$1,000 for each committee meeting attended. Such directors are granted 7,500 shares of restricted stock upon their nomination to the Board of Directors and are granted an additional 5,000 shares of restricted stock annually. All of the restricted stock grants made to such directors vest equally over five years. All directors are reimbursed for expenses connected with attendance at meetings of the Board of Directors. All directors are eligible to participate in the Stock Compensation Program. See Stock Compensation Program as described hereinafter.

Governance and Compensation Committee Interlocks and Insider Participation

The current members of the Governance and Compensation Committee are Lowell H. Lebermann, Jr., Chairman, James E. Nave, D.V.M., Lee S. Isgur and Robert E. Lewis. Ms. Boggs McDonald served as a member of the Governance and Compensation Committee prior to her resignation from the Board of Directors in March 2004. Each person is an independent director (as independence is defined in Sections 303 01(B)(2)(a) and (3) of the NYSE listing standards) of the Company.

Section 16(a) Beneficial Ownership Reporting Compliance

Section 16(a) of the Exchange Act requires the Company s executive officers and directors and persons who own more than 10% of the Company s common stock to file reports of ownership on Forms 3, 4 and 5 with the Commission. Executive officers, directors and 10% stockholders are required by the Commission to furnish the Company with copies of all Forms 3, 4 and 5 they file.

Based solely on the Company s review of the copies of such forms it has received, the Company believes that all of its executive officers, directors and greater than 10% beneficial owners complied with all of the filing requirements applicable to them with respect to transactions during 2004, with the exception of one Form 4 transaction for each of Messrs. Lebermann, Nave, Isgur and Lewis in which grants of restricted shares on December 8, 2004 were not reported until February 14, 2005 on their respective Form 5 s.

Legal Proceedings Involving Directors, Officers, Affiliates or Beneficial Owners

No director, officer, affiliate or beneficial owner of the Company, or any associate thereof, is a party adverse to the Company or any of its subsidiaries in any lawsuit nor has a material adverse interest to the Company.

Code of Ethics

The Company has adopted a Code of Business Conduct and Ethics (the Code of Ethics) that applies to all of its directors, officers (including its executive officer and chief financial officer) and employees. The Code of Ethics and any waivers or amendments to the Code of Ethics are available on the Company s website at www.stationcasinos.com. Printed copies are also available upon request directed to the Company s Corporate Secretary.

Corporate Governance Guidelines

The Company has adopted Corporate Governance Guidelines that are available on its website at www.stationcasinos.com. Printed copies are also available upon request directed to the Company s Corporate Secretary.

Stockholder Communications with Directors

Stockholders may communicate with the Board of Directors, including the non-management directors, by sending a letter to the Station Casinos, Inc. Board of Directors c/o Corporate Secretary, 2411 W. Sahara Ave, Las Vegas, Nevada 89102. The Corporate Secretary has the authority to disregard any inappropriate communications or to take other actions with respect to any such inappropriate communications. If deemed an appropriate communication, the Corporate Secretary will submit the correspondence to the Chairman of the Board or to any specific director to whom the correspondence is directed.

PRINCIPAL STOCKHOLDERS OF THE COMPANY

The following table sets forth, as of February 28, 2005, certain information regarding the shares of common stock beneficially owned by each stockholder who is known by the Company to beneficially own in excess of 5% of the outstanding shares of common stock (solely based on information reported on Forms 13G filed with the Securities and Exchange Commission), by each director and named executive officer and by all executive officers and directors as a group.

		Amount and Nature of					
		Beneficial Ownership					
			Acquirable				
		Currently	Within		ercent of		
Name and Address of Beneficial Owner(1)(2)		Owned(3)	60 days	C	lass		
Frank J. Fertitta III		5,914,103	45,000		8.7		
Lorenzo J. Fertitta		4,749,615	30,000		7.1		
Blake L. Sartini (4)		3,847,973			5.7		
Delise F. Sartini (4)		3,790,775			5.6		
FMR Corp (5)		4,393,210			6.5		
Morgan Stanley (6)		3,968,357	10,000		5.9		
Scott M Nielson (7)		732,092	10,000		1.1		
Glenn C. Christenson (8) William W. Warner (9)		406,080	15,000	0.4		A 61	
William W. Warner (9)	0.5	310,797	20,000	9.4	Notices	A-61	
	9.5	Interpretation	A-62				
	9.6	Severability	A-62				
		Entire					
	9.7	Agreement	A-62				
	9.8	Assignment	A-62				
		Parties in					
	9.9	Interest	A-62				
		Failure or					
		Indulgence					
		Not Waiver;					
		Remedies					
	0.10		۸ (۵				
	9.10	Cumulative	A-62				
		Governing					
		Law;					
	9.11	Enforcement	A-62				
	9.12	Counterparts	A-63				
	9.13	Definitions	A-63				
	EXHIBITS						

EXHIBIT A Form of Affiliate Agreement

SCHEDULES

Schedule 2.9

Company Disclosure Schedule

Parent Disclosure Schedule

A-iii

Table of Contents

AGREEMENT AND PLAN OF MERGER (this <u>Agreement</u>), made and entered into June 3, 2007 by and among QIAGEN N.V., a public limited liability company (*Naamloze Vennootschap*) organized in the Kingdom of The Netherlands (<u>Parent</u>), QIAGEN North American Holdings, Inc., a California corporation and wholly owned Subsidiary of Parent (<u>NAH</u>), QIAGEN MERGER SUB, LLC, a Delaware limited liability company and wholly owned Subsidiary of NAH (<u>Merger Sub</u>) and DIGENE CORPORATION, a Delaware corporation (the <u>Company</u>). Parent, NAH, Merger Sub and the Company are sometimes referred to herein each individually as a <u>Party</u> and, collectively, as the <u>Parties</u>. Certain capitalized terms used herein shall have the respective meanings ascribed thereto in Section 9.13.

RECITALS

WHEREAS, the Supervisory Board and Managing Board of Parent, the sole member of Merger Sub and the Boards of Directors of NAH and the Company have each declared it to be advisable and in the best interests of their respective entities and their respective equity holders that Parent acquire the Company in order to advance each of their long-term business interests; and

WHEREAS, the Supervisory Board and Managing Board of Parent, the sole member of Merger Sub and the Boards of Directors of NAH and the Company have each approved this Agreement and the merger of the Company with and into Merger Sub (the $\underline{\underline{Merger}}$), in accordance with the General Corporation Law of the State of Delaware (the $\underline{\underline{DGC}}L$), the Delaware Limited Liability Company Act ($\underline{\underline{DLLCA}}$) and the terms and conditions set forth herein; and

WHEREAS, in furtherance of such transaction, it is proposed that Parent, NAH or Merger Sub shall make an offer in the United States (as such offer may be amended from time to time, the Offer and together with the Merger and the other transactions contemplated by this Agreement, the Transactions) to acquire each issued and outstanding share of Company Common Stock in exchange for, at the election of the holder thereof, cash or Parent Common Shares or a combination of cash and Parent Common Shares, all in accordance with the terms and conditions set forth in this Agreement; and

WHEREAS, for federal income tax purposes, it is intended that (a) the exchange of Company Common Stock for cash and Parent Common Shares pursuant to the Offer and the Merger shall be treated as an integrated transaction and shall qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (<u>Code</u>) and the rules and regulations promulgated thereunder, (b) the exchange of Company Common Stock for Parent Common Shares shall result in no gain recognition to the Company s stockholders pursuant to Section 367 of the Code and the rules and regulations promulgated thereunder, (c) this Agreement constitute a plan of reorganization, and (d) Parent, NAH and the Company will each be a party to such reorganization within the meaning of Section 368(b) of the Code.

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements herein contained, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties hereby agree as follows.

1. THE OFFER

- 1.1 The Offer.
- (a) Provided that (i) this Agreement shall not have been terminated in accordance with Section 8, (ii) none of the events set forth in clauses (i) (v) of Annex I hereto shall have occurred and be continuing (and shall not have been waived by Parent) and (iii) the Company shall have complied with its applicable obligations under Section 1.2, unless otherwise agreed by Parent and the Company, as promptly as reasonably practicable after the public announcement of the execution of this Agreement, Parent shall or shall cause Merger Sub or NAH to commence (as defined in Rule 14d-2 promulgated under the Exchange Act) the Offer to purchase each share of the Company s common stock, par value \$0.01 per share

A-1

Table of Contents

(<u>Company Common Stock</u>), issued and outstanding on the Acceptance Date, in exchange for, at the election of the holder thereof, and subject to Section 1.5(b)(iv), (i) for each share of Company Common Stock with respect to which an election to receive cash has been effectively made and not revoked (a <u>Cash Election</u>), the right to receive in cash from Parent, NAH or Merger Sub, as applicable, without interest, an amount equal to \$61.25 (the <u>Per Share Cash Consideration</u>) or (ii) for each share of Company Common Stock with respect to which an election to receive common shares of Parent, 0.01 par value per share (<u>Parent Common Shares</u>), has been effectively made and not revoked (<u>a Stock El</u>ection), the right to receive from Parent, NAH or Merger Sub, as applicable, 3.545 Parent Common Shares, and any cash to be paid in lieu of fractional Parent Common Shares pursuant to this Agreement (the <u>Per Share Stock Consideration</u> and, together with the Per Share Cash Consideration, as determined in accordance with Section 1.5, the <u>Consideration</u>); and (iii) for each share of Company Common Stock other than shares as to which a Cash Election or a Stock Election has been effectively made and not revoked (<u>Non-Election Shares</u>), the right to receive from Parent such Per Share Stock Consideration and/or Per Share Cash Consideration as is determined in accordance with Section 1.5.

(b) The Offer shall be subject only to (1) the condition that there shall be validly tendered in accordance with the terms of the Offer, prior to the expiration of the Offer, and not withdrawn, a number of shares of Company Common Stock that, together with the shares of Company Common Stock owned by Parent, NAH and Merger Sub, represents, at the expiration of the Offer, at least the Minimum Condition and (2) the other conditions set forth in Annex I hereto. Upon termination of this Agreement, if the Offer has not then been consummated, the Offer shall immediately expire and terminate without any shares of Company Common Stock being purchased thereunder. Parent, NAH and Merger Sub, as applicable, expressly reserve the right to waive any of the conditions to the Offer and to make any change in the terms of or conditions to the Offer; provided, however, that, without the prior written consent of the Company, no change may be made that: (i) decreases the consideration payable in the Offer; (ii) changes the form of consideration payable in the Offer to a form other than cash or Parent Common Shares; (iii) decreases the aggregate amount of Cash Consideration available in the Offer or decreases the relative amount of Cash Consideration available in the Offer; (iv) reduces the number of shares of Company Common Stock sought in the Offer; (v) imposes conditions to the Offer in addition to those set forth above or in Annex I; (vi) modifies or waives the Minimum Condition; (vii) except as provided below, changes the date on which the Offer is scheduled to expire: or (viii) makes any other change that is adverse to the holders of Company Common Stock or to holders that have elected a particular form of Consideration. Notwithstanding the foregoing, Parent, NAH and Merger Sub, as applicable, shall (or in the case of clause (iii) below, shall at their option) extend the Offer for one or more periods (i) beyond the scheduled expiration date, which shall initially be 25 Business Days following the commencement of the Offer, up to the Termination Date, if, at the scheduled or extended expiration date of the Offer, any of the conditions to the Offer shall not have been satisfied or, to the extent permitted, waived, until such conditions are satisfied or, to the extent permitted, waived, (ii) for any period required by any rule, regulation, interpretation or position of the SEC or the staff thereof applicable to the Offer or any period required by applicable Law, or (iii) for an aggregate period of not more than 10 Business Days beyond the latest applicable date that would otherwise be permitted under clause (i) or (ii) of this sentence, if, as of such date, all of the conditions to the Offer have been satisfied or waived, but the number of shares of Company Common Stock validly tendered and not withdrawn pursuant to the Offer equals more than 80%, but less than 90% of the outstanding shares of Company Common Stock on a fully diluted basis; provided, however, that (x) Parent, NAH and Merger Sub, as applicable, shall not be obligated to extend the Offer pursuant to clause (i) of this sentence (but may elect to do so in accordance with this Section 1.1(b), provided that no such extension or series of extensions pursuant to this parenthetical that would extend the offer by more than 10 Business Days in the aggregate may be made without the prior written consent of the Company) if the Minimum Condition is not satisfied at the time such extension would otherwise be required, so long as the conditions set forth in Annex I have been satisfied, and Parent has publicly announced such fact and its intention not to extend the Offer at least two Business Days prior to the date such extension would, but for this proviso, otherwise have been required, (y) in the event Parent, NAH or Merger Sub, as applicable, elects to extend the expiration date pursuant to

A-2

Table of Contents

clause (iii), Parent, NAH and Merger Sub shall be deemed to have irrevocably waived all of the conditions to the Offer set forth in Annex I, and (z) Parent s, NAH s and Merger Sub s, as applicable, obligation to extend the Offer pursuant to clauses (i) or (ii) of this sentence is subject to such Parties respective rights to terminate this Agreement pursuant to Section 8. Except (A) as provided in clause (iii) of the previous sentence and (B) as elected by Parent in its sole discretion to extend the Offer for a period not to exceed 10 Business Days, Parent, NAH and Merger Sub, as applicable, shall not be permitted to extend the Offer without the prior written consent of the Company at the time that all conditions to the Offer have been satisfied or, to the extent permitted, waived. Parent, NAH or Merger Sub, as applicable, shall deliver a written notice to the Company (the <u>Acceptance Notice</u>) of its intention to accept for payment shares of Company Common Stock pursuant to the Offer one Business Day in advance of the proposed Acceptance Date. Subject to the foregoing and upon the terms and subject to the conditions of the Offer, Parent shall or shall cause NAH or Merger Sub to accept for payment or exchange, as promptly as reasonably practicable after the expiration of the Offer, all shares of Company Common Stock validly tendered and not withdrawn pursuant to the Offer.

(c) As soon as reasonably practicable on the date of commencement of the Offer, Parent shall, or Parent shall cause NAH or Merger Sub to, (i) file with the SEC a Tender Offer Statement on Schedule TO relating to the Offer, which shall include an offer to purchase and letter of transmittal/election form and such other ancillary documents as shall be required by applicable Law (together with any amendments or supplements thereto, the Schedule TO; and, together with the Offer Registration Statement and such other documents pursuant to which the Offer will be made, the Offer Documents), (ii) file with the SEC a registration statement on Form F-4 to register the offer and sale of Parent Common Shares pursuant to the Offer (the Offer Registration Statement) and (iii) disseminate the Offer Documents to holders of Company Common Stock. Each of the Company, Parent, NAH and Merger Sub shall use their reasonable efforts to have the Offer Registration Statement declared effective under the Securities Act as promptly as reasonably practicable after the filing thereof with the SEC and to keep the Offer Registration Statement effective as long as necessary to complete the Offer. Each of Parent, NAH and Merger Sub and the Company agree promptly to correct any information provided by it for use in the Offer Documents if and to the extent that such information shall have become false or misleading in any material respect. Parent shall, and Parent shall cause NAH or Merger Sub to, take all steps necessary to cause the Schedule TO and the Offer Registration Statement as so corrected to be filed with the SEC and the other Offer Documents as so corrected to be, at such time as reasonably agreed by Parent and the Company, disseminated to holders of shares of Company Common Stock, in each case as and to the extent required by applicable federal securities Laws. The Company and its counsel shall be given an opportunity to review and comment on the Offer Documents prior to their being filed with the SEC or disseminated to the holders of shares of Company Common Stock. Parent shall, and Parent shall cause NAH or Merger Sub to, provide the Company and its counsel with any comments Parent, NAH and Merger Sub or their counsel may receive from the SEC or its staff with respect to the Offer Documents promptly after the receipt of such comments and shall provide the Company and its counsel an opportunity to participate in the response of Parent, NAH or Merger Sub to such comments.

1.2 Company Action.

(a) The Company hereby consents to the Offer and represents that the Board of Directors of the Company (the <u>Company Board</u>), at a meeting duly called and held, has (i) determined that this Agreement and the Transactions are fair to and in the best interests of the Company and its stockholders, (ii) approved and declared advisable this Agreement and the Transactions in accordance with the DGCL, and (iii) resolved to recommend acceptance of the Offer and, as applicable, adoption of this Agreement by the Company s stockholders (the <u>Recommendation</u>); provided, however, that the Company Board may withdraw, qualify, modify or amend the Recommendation as and only to the extent permitted by Section 5.2. The Company further represents that the Company Board has received the opinion of J.P. Morgan Securities Inc. to the effect that, as of the date of this Agreement, the consideration to be received by the holders of Company Common Stock (other than Parent, NAH, Merger Sub and their respective affiliates) in the Offer and the Merger is, taken

A-3

Table of Contents

together, fair from a financial point of view to such holders, and a copy of such opinion, promptly upon receipt thereof, will be delivered to Parent. The Company hereby consents to the inclusion in the Offer Documents of the Recommendation and the Company shall not withdraw, qualify, modify or amend the Recommendation in any manner adverse to Parent, NAH or Merger Sub except as and only to the extent permitted by Section 5.2.

- (b) The Company hereby agrees to file with the SEC contemporaneously with the commencement of the Offer and disseminate to holders of shares of Company Common Stock a Solicitation/Recommendation Statement on Schedule 14D-9 (together with any amendments or supplements thereto, the <u>Schedule 14D-9</u>) that, subject to Section 5.2, shall include the Recommendation. Parent and its counsel shall be given a reasonable opportunity to review and comment on the Schedule 14D-9 prior to its being filed with the SEC or disseminated to holders of shares of Company Common Stock. The Company agrees to provide Parent and its counsel with any comments that the Company or its counsel may receive from the SEC or its staff with respect to the Schedule 14D-9 promptly after the receipt of such comments and shall provide Parent and its counsel with a reasonable opportunity to participate in the response of the Company to such comments. Each of the Company and Parent agrees promptly to correct any information provided by it for use in the Schedule 14D-9 if and to the extent that it shall have become false or misleading in any material respect. The Company agrees to take all steps necessary to cause the Schedule 14D-9 as so corrected to be filed with the SEC and to be, at such time as reasonably agreed by Parent and the Company, disseminated to holders of shares of Company Common Stock, in each case as and to the extent required by applicable federal securities Laws.
- (c) The Company shall promptly furnish Parent with mailing labels containing the names and addresses of all record holders of shares of Company Common Stock and with security position listings of shares of Company Common Stock held in stock depositories, each as of a recent date, together with all other available listings and computer files containing names, addresses and security position listings of record holders and beneficial owners of shares of Company Common Stock. The Company shall promptly furnish Parent with such additional information, including, without limitation, updated listings and computer files of stockholders, mailing labels and security position listings, and such other assistance in disseminating the Offer Documents to holders of shares of Company Common Stock as Parent may reasonably request. Subject to the requirements of applicable Law, and except for such steps as are necessary to disseminate the Offer Documents and any other documents necessary to consummate the Offer or the Merger, Parent, NAH and Merger Sub shall hold in confidence the information contained in such labels, listings and files, shall use such information only in connection with the Transactions, and, if this Agreement shall be terminated in accordance with Section 8, shall deliver to the Company all copies of such information then in their possession.

1.3 Top-Up Option.

- (a) Subject to the terms and conditions herein, the Company hereby grants to Parent an irrevocable, assignable option (the <u>Top-Up Option</u>) to purchase up to that number of shares of Company Common Stock (the <u>Top-Up Option Shares</u>) equal to the lowest number of shares of Company Common Stock that, when added to the number of shares of Company Common Stock collectively owned by Parent, NAH and Merger Sub and any of their respective affiliates immediately following consummation of the Offer, shall constitute at least 90% of the Fully-Diluted Shares (assuming the issuance of the Top-Up Option Shares), at a purchase price per Top-Up Option Share equal to the Consideration, payable in Parent Common Shares, cash or a demand note at the sole discretion of Parent.
- (b) Parent may, at its election, exercise the Top-Up Option, whether in whole or in part, at any one time after the occurrence of a Top-Up Exercise Event and prior to the Top-Up Termination Date. For the purposes hereof, a <u>Top-Up Exercise Event</u> shall occur upon Parent s acceptance for payment pursuant to the Offer (including, without limitation, any subsequent offering that Parent may elect to extend pursuant to the terms and conditions of this Agreement) of shares of Company Common Stock constituting, together with shares of Company Common Stock owned directly or indirectly by any other affiliates of Parent, less

A-4

Table of Contents

than 90% of the Fully-Diluted Shares, but only if (i) the issuance of the Top-Up Option Shares pursuant thereto would not require the approval of the stockholders of the Company under NASDAQ rules and regulations, or NASDAQ has granted a waiver from any such rule or regulation that is reasonably acceptable to the parties hereto, (ii) there is no other applicable law, rule or regulation that would require the approval of the Company s stockholders for the issuance of the Top-Up Shares, or any such approval shall have been waived, (iii) there are sufficient shares of Company Common Stock available for issuance under the Company s Certificate of Incorporation or available in the treasury of the Company, and (iv) any Parent Common Stock to be issued in connection with the exercise of the Top-Up Option may be issued pursuant to a valid exemption from the registration requirements of the Securities Act or any state securities laws. Upon and after the request of Parent, the Company shall use its reasonable best efforts to obtain any waiver contemplated by clause (i) of the preceding sentence from NASDAQ as promptly as possible after any such request. For the purposes hereof, the Top-Up Termination Date shall occur upon the earliest to occur of (A) the Effective Time, (B) the termination of this Agreement, (C) the date that is ten (10) Business Days after the occurrence of a Top-Up Exercise Event, unless the Top-Up Option has been previously exercised in accordance with the terms and conditions hereof and (D) the date that is ten Business Days after the Top-Up Notice Date, unless the Top-Up Closing shall have previously occurred.

- (c) If Parent wishes to exercise the Top-Up Option, Parent shall send to the Company a written notice (a <u>Top-Up Exercise Notice</u>, and the date of receipt of which notice is referred to herein as the <u>Top-Up Notice Date</u>), specifying the place for the closing of the purchase and sale of shares of Company Common Stock pursuant to the Top-Up Option (the <u>Top-Up Closing</u>) and a date not earlier than one Business Day nor later than ten Business Days after the Top-Up Notice Date for the Top-Up Closing. The Company shall, promptly after receipt of the Top-Up Exercise Notice, deliver a written notice to Parent confirming the number of Top-Up Option Shares and the aggregate purchase price therefor.
- (d) At the Top-Up Closing, subject to the terms and conditions of this Agreement, (i) the Company shall deliver to Parent a certificate or certificates evidencing the applicable number of Top-Up Option Shares; *provided* that the obligation of the Company to deliver Top-Up Option Shares upon the exercise of the Top-Up Option is subject to the condition that no provision of any applicable law or regulation and no judgment, injunction, order or decree shall prohibit the exercise of the Top-Up Option or the delivery of the Top-Up Option Shares in respect of any such exercise and (ii) Parent shall purchase each Top-Up Option Share from the Company for the Consideration described above. Payment by Parent of the Consideration for the Top-Up Option Shares may be made, at Parent s option, by delivery of (A) immediately available funds by wire transfer to an account designated by the Company, (B) a demand note issued by Parent in customary form and in a principal face amount equal to the purchase price for the Top-Up Option Shares or (C) Parent Common Shares.
- (e) Upon the delivery by Parent to the Company of the Top-Up Exercise Notice, and the tender of the Consideration described in Section 1.3(d), Parent shall be deemed to be the holder of record of the Top-Up Option Shares issuable upon that exercise, notwithstanding that the stock transfer books of the Company shall then be closed or that certificates representing those Top-Up Option Shares shall not then be actually delivered to Parent or the Company shall have failed or refused to designate the bank account described in Section 1.3(d).
- (f) Parent shall pay all expenses, and any and all federal, state and local stock transfer taxes and other related charges, that may be payable in connection with the preparation, issuance and delivery of stock certificates under this Section 1.3.
- (g) Certificates evidencing Top-Up Option Shares delivered hereunder may include legends legally required, including a legend in substantially the following form:

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY BE REOFFERED OR SOLD ONLY IF SO REGISTERED OR IF AN EXEMPTION FROM SUCH REGISTRATION IS AVAILABLE.

A-5

Table of Contents

1.4 Directors.

- (a) Effective upon the Acceptance Date, Parent shall be entitled to designate the number of directors, rounded up to the next whole number, on the Company Board that equals the product of the (i) total number of directors on the Company Board (giving effect to the election of any additional directors pursuant to this Section 1.4) and (ii) the percentage that the number of shares of Company Common Stock beneficially owned by Parent, NAH or Merger Sub (including shares of Company Common Stock accepted for payment or exchange) bears to the total number of shares of Company Common Stock outstanding, and the Company shall take all action necessary to cause Parent s designees to be elected or appointed to the Company Board, including increasing the number of directors, and seeking and accepting resignations of incumbent directors. At such time, the Company will also use its best efforts to cause individuals designated by Parent to constitute the number of members, rounded up to the next whole number, on (i) each committee of the Company Board and (ii) each board of directors of each Subsidiary of the Company identified by Parent (and each committee thereof) that represents the same percentage as such individuals represent on the Company Board, in each case only to the extent permitted by applicable Law. Notwithstanding the provisions of this Section 1.4, the parties hereto shall use their respective best efforts (including by reducing the number of directors that Parent may designate pursuant to the first sentence of this paragraph (a), but in no event to less than a majority of the directors on the Company Board) to ensure that at least two of the members of the Company Board shall, at all times prior to the Effective Time, be directors of the Company who were directors of the Company on the date hereof (the <u>Continuing Directors</u>); provided that if there shall be in office fewer than two Continuing Directors for any reason, the Company Board shall cause a person designated by the remaining Continuing Director to fill such vacancy who shall be deemed to be a Continuing Director for all purposes of this Agreement, or if no Continuing Directors then remain, the other directors of the Company then in office shall designate two persons to fill such vacancies who will not be officers or employees or affiliates of the Company, Parent, NAH or Merger Sub or any of their respective Subsidiaries and such persons shall be deemed to be Continuing Directors for all purposes of this Agreement.
- (b) The Company s obligations to appoint Parent s designees to the Company Board shall be subject to Section 14(f) of the Exchange Act and Rule 14f-1 promulgated thereunder. The Company shall promptly take all actions, and shall include in the Schedule 14D-9 such information with respect to the Company and its officers and directors, as Section 14(f) and Rule 14f-1 require in order to fulfill its obligations under this Section, so long as Parent shall have provided to the Company on a timely basis in writing and be solely responsible for any information with respect to itself, NAH, Merger Sub and their respective nominees, officers, directors and affiliates required by Section 14(f) and Rule 14f-1.
- (c) Following the election or appointment of Parent s designees pursuant to Section 1.4(a) and until the Effective Time, the approval of a majority of the Continuing Directors shall be required to authorize any termination of this Agreement by the Company, any amendment of this Agreement requiring action by the Company Board, any extension of time for performance of any obligation or action hereunder by Parent, NAH or Merger Sub, any waiver of compliance with any of the agreements or conditions contained herein for the benefit of the Company, any amendment of the certificate of incorporation or by-laws of the Company, and any other action of the Company hereunder which adversely affects the holders of shares of Company Common Stock (other than Parent, NAH or Merger Sub).
- 1.5 Allocation of Consideration; Election Procedures.
- (a) Allocation. Notwithstanding anything in this Agreement to the contrary, the maximum number of shares of Company Common Stock (the <u>Cash Election Number</u>) to be exchanged for Per Share Cash Consideration pursuant to the Offer shall be 55% of the number of whole shares of Company Common Stock that are tendered and accepted for payment or exchange pursuant to the Offer. The number of shares of Company Common Stock to be exchanged for Per Share Stock Consideration pursuant to the Offer (the <u>Stock Election Number</u>) shall be equal to the number of whole shares of Company Common Stock that are tendered and accepted for payment or exchange pursuant to the Offer less the Cash Election Number.

A-6

Table of Contents

- (b) Election Procedures.
- (i) Subject to allocation and proration in accordance with the provisions of this Section 1.5, each record holder of shares of Company Common Stock issued and outstanding immediately prior to the Election Deadline (A) may specify in a request made in accordance with the provisions of this Section 1.5(b) (1) the number of shares of Company Common Stock owned by such holder with respect to which such holder desires to make a Cash Election and (2) the number of shares of Company Common Stock owned by such Holder with respect to which such Holder desires to make a Stock Election or (B) may indicate that such holder has no preference as to the receipt of Per Share Cash Consideration or Per Share Stock Consideration for such shares of Company Common Stock (a Non-Election). Shares of Company Common Stock in respect of which a Non-Election is made (Non-Election Shares) shall be deemed by Parent, in its sole and absolute discretion, subject to Section 1.5(b)(iv)-(vii), to be, in whole or in part, shares of Company Common Stock in respect of which Cash Elections or Stock Elections have been made.
- (ii) Elections pursuant to Section 1.5(b)(i) shall be made on a form and with such other provisions to be reasonably agreed upon by the Company and Parent (a Form of Election) to be provided by the Exchange Agent for that purpose to holders of record of shares of Company Common Stock, together with appropriate transmittal materials, at the time of mailing to holders of record of shares of Company Common Stock of the Offer Documents. Elections shall be made by mailing to the Exchange Agent a duly completed Form of Election. To be effective, a Form of Election must be (x) properly completed, signed and submitted to the Exchange Agent at its designated office, by 5:00 p.m., on the Acceptance Date, (which date shall be publicly announced by Parent as soon as reasonably practicable but in no event less than five trading days prior to the Acceptance Date (the Election Deadline). The Company shall use its commercially reasonable efforts to make a Form of Election available to all persons who become holders of record of shares of Company Common Stock between the date of mailing described in the first sentence of this Section 1.5(b)(ii) and the Election Deadline. Parent shall determine, in its reasonable discretion, which authority it may delegate in whole or in part to the Exchange Agent, whether Forms of Election have been properly completed, signed and submitted or revoked. Neither Parent nor the Exchange Agent will be under any obligation to notify any person of any defect in a Form of Election submitted to the Exchange Agent. A holder of shares of Company Common Stock that tenders shares of Company Common Stock in the Offer and does not submit an effective Form of Election prior to the Election Deadline shall be deemed to have made a Non-Election.
- (iii) An election may be revoked with respect to any shares of Company Common Stock, but only by written notice received by the Exchange Agent prior to the Election Deadline. Upon any such revocation, unless a duly completed Form of Election is thereafter submitted in accordance with Section 1.5(b)(i), such shares of Company Common Stock shall be Non-Election Shares.
- (iv) In the event that the aggregate number of shares of Company Common Stock in respect of which Cash Elections have been made (collectively, the <u>Cash Election Shares</u>) exceeds the Cash Election Number (such excess the <u>Excess Cash Election Shares</u>), then:
- 1. all shares in respect of which Stock Elections have been made (the <u>Stock Election Shares</u>) shall be exchanged for Per Share Stock Consideration,
- 2. all Non-Election Shares shall be deemed converted into Stock Election Shares and exchanged for Per Share Stock Consideration,
- 3. all Excess Cash Election Shares shall be deemed converted into Stock Election Shares, on a pro-rata basis for each record holder of Cash Election Shares so that the number of Excess Cash Election Shares so converted, when added to the other Stock Election Shares, shall equal as closely as reasonably practicable the Stock Election Number, and all such Excess Cash Election Shares so converted shall be exchanged for Per Share Stock Consideration; and

A-7

Table of Contents

- 4. all Cash Election Shares not deemed converted into Stock Election Shares pursuant to 3 above shall be exchanged for Per Share Cash Consideration.
- (v) In the event that the aggregate number of Stock Election Shares exceeds the Stock Election Number (such excess the <u>Excess Stock Election Shares</u>), then:
- 1. all Cash Election Shares shall be exchanged for Per Share Cash Consideration,
- 2. all Non-Election Shares shall be deemed converted into Cash Election Shares and exchanged for Per Share Cash Consideration,
- 3. all Excess Stock Election Shares shall be deemed converted into Cash Election Shares, on a pro-rata basis for each record holder of Stock Election Shares, so that the number of Excess Stock Election Shares so converted, when added to the other Cash Election Shares, shall equal as closely as reasonably practicable the Cash Election Number, and all such Excess Stock Election Shares so converted shall be exchanged for Per Share Cash Consideration, and
- 4. all Stock Election Shares not deemed converted into Cash Election Shares pursuant to 3 above shall be exchanged for Per Share Stock Consideration.
- (vi) In the event that neither clause (iv) nor clause (v) of this Section 1.5(b) is applicable, Non-Election Shares shall be deemed converted into Stock Election Shares on a pro rata basis for each record holder of Non-Election Shares such that the total number of Stock Election Shares equals the Stock Election Number and any remaining Non-Election Shares shall be deemed converted into Cash Election Shares, and (x) all Cash Election Shares and all Non-Election Shares which are deemed converted into Cash Election Shares shall be exchanged for Per Share Stock Election Shares and all Non-Election Shares which are deemed converted into Stock Election Shares shall be exchanged for Per Share Stock Consideration.
- (vii) The Exchange Agent, in consultation with Parent and the Company, shall make all computations to give effect to this Section 1.5(b).
- 1.6 Exchange of Certificates. The procedures for exchanging Company Certificates representing outstanding shares of Company Common Stock for the Consideration pursuant to the Offer are set forth below.
- (a) Exchange Agent. Prior to the Acceptance Date, Parent shall designate a bank or trust company reasonably acceptable to the Company to act as Exchange Agent in connection with the Offer (the <u>Exchange Agent</u>).
- (b) Exchange Fund. No later than the Acceptance Date, as a condition to acceptance of the shares of Company Common Stock tendered in the Offer, Parent shall deposit, or shall cause to be deposited, with the Exchange Agent, for the benefit of the holders of Company Common Stock, for exchange in accordance with the terms of the Offer, (i) the Offer Cash Pool plus an estimate of any cash to be paid in lieu of fractional Parent Common Shares pursuant to this Agreement and (ii) certificates representing the Parent Common Shares issuable to holders of Company Common Stock in the Offer (such cash and certificates for Parent Common Shares, together with any dividends or distributions with respect thereto, being hereinafter referred to as, the Exchange Fund) and shall give irrevocable instructions to the Exchange Agent to make cash payments and deliver the Parent Common Shares in exchange for the shares of Company Common Stock. The Exchange Agent shall, pursuant to such irrevocable instructions, make cash payments and deliver the Parent Common Shares contemplated to be issued pursuant to this Agreement out of the Exchange Fund.
- (c) Distributions on Parent Common Shares. For purposes of determining entitlement to dividends or other distributions declared on Parent Common Shares, holders of Company Common Stock who have validly tendered and not withdrawn such shares pursuant to the Offer shall be deemed to be record holders of Parent Common Shares as of the Acceptance Date, notwithstanding the fact that certificates representing such shares have not yet been issued or delivered to tendering stockholders (or, if applicable, appropriate book-entries have not yet been made).

A-8

Table of Contents

2. THE MERGER

- 2.1 *The Merger*. Subject to the consummation of the Offer, in accordance with the DGCL, the DLLCA and the terms and conditions of this Agreement, the Company shall be merged with and into Merger Sub. From and after the Effective Time, the separate corporate existence of the Company shall cease and Merger Sub, as the surviving entity in the Merger, shall continue its existence under the Laws of the State of Delaware as a wholly owned Subsidiary of NAH. Merger Sub as the surviving entity after the Merger is hereinafter sometimes referred to as the Surviving Company.
- 2.2 Closing. Unless this Agreement shall have been terminated pursuant to the provisions of Section 8, and subject to the satisfaction or waiver, as the case may be, of the conditions set forth in Section 7, the closing of the Merger (the Closing) shall take place at a time and on a date to be mutually agreed upon by the Parties (the Closing Date), which date shall be no later than the fifth Business Day after all the conditions set forth in Section 7 (excluding conditions that, by their nature, cannot be satisfied until the Closing, it being understood that the occurrence of the Closing shall remain subject to the satisfaction or waiver of such conditions) shall have been satisfied or waived, unless another time and/or date is agreed to in writing by Parent and the Company. The Closing shall take place at the offices of Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C., One Financial Center, Boston, MA 02111, unless another place is agreed to in writing by the Parties. For purposes of this Agreement,

 Business Day shall mean any day other than Saturday, Sunday or a legal holiday on which banks are permitted to be closed in New York, New York.
- 2.3 Filing of Certificate of Merger. Subject to the provisions of this Agreement, at the Closing, the Parties shall cause the Merger to become effective by causing the Surviving Company to execute and file in accordance with the DGCL and the DLLCA a certificate of merger with the Secretary of State of the State of Delaware (the <u>Certificate of Merger</u>). The Merger shall become effective upon such filing, or at such later date and time as is agreed to by Parent and the Company and set forth in the Certificate of Merger (the <u>Effective Time</u>).
- 2.4 Effect of the Merger. Upon the Closing, the Merger shall have the effects set forth in this Agreement and in Section 259 of the DGCL and Section 18-209 of the DLLCA.
- 2.5 Conversion of Shares.
- (a) Subject to this Section 2, at the Effective Time, each share of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than Excluded Shares and Dissenting Shares) shall, by virtue of the Merger and without any action on the part of Parent, NAH, Merger Sub, the Company or the holder thereof, be converted into and shall become the right to receive the Consideration in accordance with the provisions of this Section 2, and each holder of a Company Certificate shall cease to have any rights with respect thereto other than, at the election of the holder thereof, (i) for each share of Company Common Stock with respect to which an election to receive cash has been effectively made and not revoked (a Merger Cash Election), the right to receive in cash, without interest, an amount equal to the Per Share Cash Consideration, (ii) for each share of Company Common Stock with respect to which an election to receive Parent Common Shares has been effectively made and not revoked (a Merger Stock Election), the right to receive the Per Share Stock Consideration, and (iii) for each share of Company Common Stock other than shares as to which a Merger Cash Election or a Merger Stock Election has been effectively made and not revoked (Merger Non-Election Shares), the right to receive from Parent such Per Share Stock Consideration and/or Per Share Cash Consideration as is determined in accordance with this Section 2.5.
- (b) *Allocation*. Notwithstanding anything in this Agreement to the contrary, the maximum number of shares of Company Common Stock (the <u>Merger Cash Election Number</u>) to be converted into the right to receive Per Share Cash Consideration pursuant to the Merger shall be equal to 55% of the number of shares of Company Common Stock (other than Excluded Shares and Dissenting Shares) issued and outstanding immediately prior to the Effective Time. The number of shares of Company Common Stock to be converted

A-9

Table of Contents

into the right to receive Per Share Stock Consideration pursuant to the Merger (the <u>Merger Stock Election Number</u>) shall be equal to the number of shares of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than Excluded Shares and Dissenting Shares) less the Merger Cash Election Number.

- (c) Election Procedures.
- (i) Any Merger Cash Election or Merger Stock Election shall be made on a form furnished by Parent for that purpose (a Merger Form of Election), which form may be part of the letter of election and transmittal delivered to former holders of shares of Company Common Stock promptly following the Effective Time. Holders of record who hold shares of Company Common Stock as nominees, trustees or in other representative capacities may submit multiple Merger Form of Election on behalf of their respective beneficial holders. To be effective, a Merger Form of Election must be properly completed, signed and submitted to the Exchange Agent at its designated office, within 45 calendar days after the Closing Date (the Merger Election Deadline). Parent shall determine, in its reasonable discretion, which authority it may delegate in whole or in part to the Exchange Agent, whether Merger Forms of Election have been properly completed, signed and submitted or revoked. Neither Parent nor the Exchange Agent will be under any obligation to notify any person of any defect in a Merger Form of Election submitted to the Exchange Agent. A holder of shares of Company Common Stock that does not submit an effective Merger Form of Election prior to the Merger Election Deadline shall be deemed to have made a Merger Non-Election.
- (ii) Subject to allocation and proration in accordance with the provisions of this Section 2.5, each record holder of shares of Company Common Stock (other than Excluded Shares and Dissenting Shares) issued and outstanding immediately prior to the Effective Time (A) may specify in a request made in accordance with the provisions of this Section 2.5 (1) the number of shares of Company Common Stock owned by such holder with respect to which such holder desires to make a Merger Cash Election and (2) the number of shares of Company Common Stock owned by such Holder with respect to which such Holder desires to make a Merger Stock Election or (B) may indicate that such holder is making a Merger Non-Election for such shares of Company Common Stock. Merger Non-Election Shares shall be deemed by Parent, in its sole and absolute discretion, subject to Section 2.5(c)(iii)-(v), to be, in whole or in part, shares of Company Common Stock in respect of which Merger Cash Elections or Merger Stock Elections have been made.
- (iii) In the event that the aggregate number of shares of Company Common Stock in respect of which Merger Cash Elections have been made (collectively, the <u>Merger Cash Election Shares</u>) exceeds the Merger Cash Election Number (such excess, the <u>Excess Merger Cash Elec</u>tion <u>Shares</u>), then:
- 1. all shares in respect of which Merger Stock Elections have been made (the <u>Merger Stock Election Shares</u>) shall be converted into the right to receive Per Share Stock Consideration,
- 2. all Merger Non-Election Shares shall be deemed converted into Merger Stock Election Shares and shall be converted into the right to receive Per Share Stock Consideration,
- 3. all Excess Merger Cash Election Shares shall be deemed converted into Merger Stock Election Shares on a pro-rata basis for each record holder of Merger Cash Election Shares, so that the number of Excess Merger Cash Election Shares so converted, when added to the other Merger Stock Election Shares, shall equal as closely as reasonably practicable the Merger Stock Election Number, and all such Excess Merger Cash Election Shares so converted shall be converted into the right to receive Per Share Stock Consideration, and
- 4. all Merger Cash Election Shares not deemed converted into Merger Stock Election Shares pursuant to 3 above shall be converted into the right to receive Per Share Cash Consideration.

A-10

Table of Contents

- (iv) In the event that the aggregate number of Merger Stock Election Shares exceeds the Merger Stock Election Number (such excess, the <u>Excess Merger Stock Election Shares</u>), then:
- 1. all Merger Cash Election Shares shall be converted into the right to receive Per Share Cash Consideration,
- 2. all Merger Non-Election Shares shall be deemed converted into Merger Cash Election Shares and shall be converted into the right to receive Per Share Cash Consideration.
- 3. all Excess Merger Stock Election Shares shall be deemed converted into Merger Cash Election Shares, on a pro-rata basis for each record holder of Merger Stock Election Shares, so that the number of Excess Merger Stock Election Shares so converted, when added to the other Merger Cash Election Shares, shall equal as closely as reasonably practicable the Merger Cash Election Number, and all such shares of Excess Company Common Stock so converted shall be converted into the right to receive the Per Share Cash Consideration, and
- 4. all Merger Stock Election Shares not deemed converted into Merger Cash Election Shares pursuant to 3 above shall be converted into the right to receive Per Share Stock Consideration.
- (v) In the event that neither clause (iii) nor clause (iv) of this Section 2.5 is applicable, Merger Non-Election Shares shall be deemed Merger Stock Election Shares on a pro rata basis for each record holder of Merger Non-Election Shares such that the total number of Merger Stock Election Shares equals the Merger Stock Election Number and any remaining Merger Non-Election Shares shall be deemed Merger Cash Election Shares, and (x) all Merger Cash Election Shares and all Merger Non-Election Shares in respect of which Merger Cash Elections are deemed to have been made shall be converted into the right to receive Per Share Cash Consideration, and (y) all Merger Stock Election Shares and all Merger Non-Election Shares in respect of which Merger Stock Elections are deemed to have been made shall be converted into the right to receive Per Share Stock Consideration.
- (vi) The Exchange Agent, in consultation with Parent and the Company, shall make all computations to give effect to this Section 2.5.
- 2.6 Exchange of Certificates. The procedures for exchanging Company Certificates representing outstanding shares of Company Common Stock for the Consideration pursuant to the Merger are set forth below.
- (a) Merger Exchange Fund. Immediately prior to the Effective Time, Parent shall deposit, or shall cause to be deposited, with the Exchange Agent, for the benefit of the holders of Company Common Stock, for exchange in accordance with the terms of the Merger, (i) the Merger Cash Pool plus an estimate of any cash to be paid in lieu of fractional Parent Common Shares pursuant to the Merger and (ii) certificates representing the Parent Common Shares issuable to holders of Company Common Stock in the Merger (such cash and certificates for Parent Common Shares, together with any dividends or distributions with respect thereto, being hereinafter referred to as, the Merger Exchange Fund). The Exchange Agent shall, pursuant to irrevocable instructions, make cash payments and deliver the Parent Common Shares contemplated to be issued pursuant to this Agreement out of the Merger Exchange Fund.
- (b) Exchange Procedures. Upon surrender of a Company Certificate for cancellation to the Exchange Agent, together with a letter of transmittal, duly completed and validly executed in accordance with the instructions thereto and such other documents as may reasonably be required pursuant to such instructions, the holder of such Company Certificate shall be entitled to receive promptly in exchange therefor (A) a certificate representing the number of whole Parent Common Shares that such holder has the right to receive as part of the Consideration and (B) a check for the cash that such holder is entitled to receive as part of the Consideration, including the Per Share Cash Consideration and cash in lieu of fractional shares. Until so surrendered, each outstanding Company Certificate shall be deemed from and after the Effective Time for all corporate purposes, to evidence the right to receive upon such surrender the Consideration. Any portion of the Parent Common Shares and cash deposited with the Exchange Agent pursuant to this Agreement, which remains undistributed to the holders of the shares of Company Common Stock for 12 months after the

A-11

Table of Contents

Closing shall be delivered to Parent, upon demand, and any holders of shares of Company Common Stock who have not theretofore complied with this Section 2.6 shall thereafter be entitled to receive from Parent payment of the Consideration to be issued in exchange therefor and any cash in lieu of fractional shares, dividends or distributions with respect to Parent Common Shares to which such holders may be then entitled.

- 2.7 Payment for Parent Common Shares. The Parent Common Shares issued as Per Share Stock Consideration shall, for purposes of the Laws of The Netherlands, be paid for by the holder of the Company Common Stock through the subscription described below. By virtue of the Merger or the tendering of Company Common Stock in the Offer, as applicable, and without any further action on the part of the holders of Company Common Stock, each holder of Company Common Stock shall be deemed to have subscribed under this Agreement for the Parent Common Shares that are to be issued to such holder in accordance with this Agreement. In accordance with the Laws of The Netherlands, each holder of Company Common Stock who receives Parent Common Shares as part of such holder s Consideration shall be obligated to pay up the Parent Common Shares so issued to such holder in accordance with this Agreement. This payment obligation on the part of such holders of Company Common Stock (which, solely for the purpose of satisfying this obligation to pay up the Parent Common Shares, shall be equal to the par value of such Parent Common Shares) shall be fully satisfied and paid by such holders of Company Common Stock by Parent s assumption of such holder s right to receive compensation (which, solely for the purpose of satisfying this obligation to compensate the holders of Company Common Stock, shall also be equal to the par value of such Parent Common Shares) from NAH for the cancellation of the Company Common Stock, or exchange of such Company Common Stock in connection with the consummation of the Offer, followed by a set off of this right to receive compensation against the obligation to pay up the Parent Common Shares. It is understood that, if the Merger or the Offer is consummated, the stockholders of the Company are entitled to receive the Parent Common Shares directly from Parent, NAH or Merger Sub and that, if the Merger or the Offer is consummated, such stockholders will thereupon become third party beneficiaries of this Agreement solely for this purpose. For the avoidance of doubt, (x) in no event shall the amount of the Consideration to be received by each holder of Company Common Stock pursuant to this Agreement be reduced by the obligation to pay up the Parent Common Shares and (y) NAH shall be obligated to deliver the Consideration to holders of Company Common Stock in accordance with, and subject to the terms of, this Agreement; provided, however, that the obligations of NAH to deliver the Consideration under this Agreement may be assigned, in whole or in part, to Parent or any of Parent s wholly-owned direct or indirect Subsidiaries.
- 2.8 Certificate of Formation and Operating Agreement of the Surviving Company. At the Effective Time, (a) the Certificate of Formation of Merger Sub (the <u>Certificate of Formation</u>) immediately prior to the Closing shall continue in its then current form until amended as provided therein and under the DLLCA and (b) the Operating Agreement of Merger Sub (the <u>Operating Agreement</u>) immediately prior to the Closing shall continue in its then current form until amended as provided therein and under the DLLCA.
- 2.9 Directors and Officers. Subject to the requirements of Law, the persons listed on Schedule 2.9 shall be the officers of the Surviving Company immediately after the Closing, each to hold office in accordance with the Certificate of Formation and the Operating Agreement, in each case until their respective successors are duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Certificate of Formation and the Operating Agreement.
- 2.10 *Cancellation of Shares*. At the Effective Time, each share of Company Common Stock owned by Parent or any direct or indirect wholly owned Subsidiary of Parent immediately prior to the Effective Time (collectively, <u>Excluded Shares</u>), shall be canceled and extinguished without any conversion thereof or payment therefor.
- 2.11 *No Liability*. To the extent permitted by applicable Law, none of the Exchange Agent, Parent, NAH, Merger Sub or the Surviving Company shall be liable to a holder of shares of Company Common Stock for any Parent Common Shares or any amount of cash properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Law or retained by Parent.

A-12

Table of Contents

- 2.12 Dissenting Shares. Notwithstanding anything to the contrary in this Agreement, any shares of Company Common Stock outstanding immediately prior to the Effective Time and held by a person who has not voted in favor of the Merger or consented thereto in writing and who has demanded appraisal for such shares in accordance with the DGCL (the Dissenting Shares) shall not be converted into a right to receive the Consideration, unless such holder fails to perfect or withdraws or otherwise loses its rights to appraisal or it is determined that such holder does not have appraisal, or if it is determined that such holder does not have appraisal rights, such shares shall be treated as if they had been converted as of the Effective Time into the right to receive the Consideration. The Company shall give Parent, NAH and Merger Sub prompt notice of any demands received by the Company for appraisal of shares, and Parent, NAH and Merger Sub shall have the right to participate in all negotiations and proceedings with respect to such demands except as required by applicable Law. The Company shall not, except with prior written consent of Parent, make any payment with respect to, or settle or offer to settle, any such demands, unless and to the extent required to do so under applicable Law.
- 2.13 Taking of Necessary Action; Further Action. If, at any time and from time to time after the Closing, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest in the Surviving Company full right, title, interest and possession of all properties, assets, rights, privileges, powers and franchises of the Company and Merger Sub, the officers and directors of the Surviving Company shall be and are fully authorized, in the name of and on behalf of any of the Company, Merger Sub or the Surviving Company, to take, or cause to be taken, all such lawful and necessary action as is not inconsistent with this Agreement.
- 2.14 Distributions With Respect to Unexchanged Shares. No dividends or other distributions declared or made after the Closing with respect to Parent Common Shares with a record date after the Closing will be paid to the holder of any unsurrendered shares of Company Common Stock with respect to the Parent Common Shares represented thereby, and no cash in lieu of fractional Parent Common Shares shall be paid to any such holder, until the holder of record of the Company Certificate representing such Company Common Stock shall surrender such Company Certificate. Subject to Law, following surrender of any such Company Certificate, there shall be paid to the holder of record of such Company Certificate Parent Common Shares representing the whole number of Parent Common Shares to be issued in exchange therefor, the applicable Per Share Cash Consideration, any cash in lieu of fractional Parent Common Shares and any dividends or other distributions with a record date after the Closing theretofore paid with respect to such whole number of Parent Common Shares.
- 2.15 Transfers of Ownership. If any certificate for Parent Common Shares or check is to be issued in a name other than that in which the Company Certificate surrendered in exchange therefor is registered, it shall be a condition of the issuance thereof that the Company Certificate so surrendered shall be properly endorsed and otherwise in proper form for transfer, accompanied by all documents reasonably required to evidence and effect such transfer, and that the stockholder requesting such exchange shall have paid to Parent, or any agent designated by Parent, any transfer or other taxes required by reason of the issuance of a certificate for Parent Common Shares or a check in any name other than that of the registered holder of the certificate surrendered, or established to the reasonable satisfaction of Parent or any agent designated by it that such tax has been paid or is otherwise not payable.
- 2.16 Withholding of Tax. Parent, Surviving Company or the Exchange Agent will be entitled to deduct and withhold from the Consideration otherwise payable pursuant to this Agreement to any holder of shares of Company Common Stock such amounts as Parent (or any Affiliate, including the Surviving Company) or the Exchange Agent are required to deduct and withhold with respect to the making of such payment under the Code or any provision of federal, state, local or foreign tax Law. To the extent that amounts are so withheld by Parent, Surviving Company or the Exchange Agent, such withheld amounts (i) shall be remitted by Parent, the Surviving Company or the Exchange Agent, as the case may be, to the applicable Governmental Authority and (ii) shall be treated for all purposes of this Agreement as having been paid to the holder of the shares of Company Common Stock in respect of whom such deduction and withholding were made by Parent.

A-13

Table of Contents

- 2.17 *Lost Certificates*. If any Company Certificate is lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Company Certificate to be lost, stolen or destroyed, and an indemnification against loss in customary form, the Exchange Agent shall issue in exchange for such lost, stolen or destroyed Company Certificate the Consideration deliverable in respect thereof pursuant to this Agreement.
- 2.18 Company Equity Awards.
- (a) At the Effective Time, all outstanding Company Equity Awards under the Amended and Restated Directors Equity Compensation Plan (the <u>Cancelled Equity Awards</u>) will be terminated or cancelled, as the case may be, in accordance with the terms of such plan and the agreements entered into thereunder. Prior to the Effective Time, the Company shall give any notice required by the Amended and Restated Directors Equity Compensation Plan, which notice shall have been provided to Parent for its review prior to delivery, to holders of Cancelled Equity Awards thereunder of (i) the acceleration in full of the vesting of such Cancelled Equity Awards, effective as of a date determined by the Company on or prior to the date of the Effective Time and (ii) the termination or cancellation, as the case may be, upon the Closing of any unexercised Cancelled Equity Awards.
- (b) At the Effective Time, each outstanding Company Equity Award described in Section 3.3(e) of the Company Disclosure Schedule other than Cancelled Equity Awards (the <u>Assumed Equity Awards</u>), whether vested or unvested, shall be assumed by Parent and shall become an award with respect to, on the same terms and conditions as were applicable under the applicable Company Equity Plan immediately prior to the Effective Time, the number of Parent Common Shares determined as follows:
- (i) the number of Parent Common Shares subject to each Assumed Equity Award shall be determined by multiplying the number of shares of Company Common Stock that were subject to such Assumed Equity Award immediately prior to the Effective Time by the Option Conversion Ratio (as defined below), and rounding the resulting number down to the nearest whole number of Parent Common Shares; and
- (ii) the per share exercise price for the Parent Common Shares issuable upon exercise of each Assumed Equity Award shall be determined by dividing the per share exercise price of Company Common Stock subject to such Assumed Equity Award, as in effect immediately prior to the Effective Time, by the Option Conversion Ratio, and rounding the resulting exercise price up to the nearest whole cent;

provided, however, that the exercise price and the number of Parent Common Shares subject to each Company Equity Award shall be determined in a manner consistent with the requirements of Section 409A of the Code to the extent applicable; and provided, further, that in the case of any Company Equity Award to which Section 422 of the Code applies, the option price, the number of shares subject to such Company Equity Award and the terms and conditions of exercise of such Company Equity Award shall be determined in accordance with the foregoing, subject to such adjustments as are necessary in order to satisfy the requirements of Section 424(a) of the Code.

- (c) Any restriction on the exercise of any Assumed Equity Awards assumed by Parent shall continue in full force and effect and the term, exercisability, vesting schedule and other provisions of such Assumed Equity Awards shall otherwise remain unchanged as a result of the assumption of such Assumed Equity Awards, in each case except to the extent otherwise provided in any stock option, change in control or in any retention agreement in effect on the date of this Agreement. The Option Conversion Ratio shall be equal to 3.545.
- (d) Following the Effective Time, Parent shall deliver to the participants in the Company Equity Plans holding Assumed Equity Awards an appropriate notice setting forth such participants rights pursuant to the Assumed Equity Awards, as provided in this Section 2.18.
- (e) Parent shall take all corporate action necessary to reserve for issuance a sufficient number of Parent Common Shares for delivery upon exercise of the Assumed Equity Awards assumed in accordance with this Section 2.18. Within five (5) Business Days following the Effective Time, Parent shall file a registration statement on Form S-8 (or any successor form) with respect to the Parent Common Shares subject to such

A-14

Table of Contents

Assumed Equity Awards held by persons who become employees or consultants of the Surviving Company and shall use its reasonable best efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as such Assumed Equity Awards remain outstanding.

- (f) The Company and Parent shall each take all corporate and other action reasonably necessary to cause all dispositions of equity securities of the Company (including Assumed Equity Awards) or acquisitions of equity securities of Parent (including any Assumed Equity Awards to acquire Parent Common Shares that may be granted by Parent) by each individual who (i) is a director or officer of the Company or (ii) at the Effective Time will become a director or officer of Parent, to be exempt under Rule 16b-3 of the Exchange Act.
- 2.19 Adjustments to Per Share Stock Consideration. The number of Parent Common Shares issuable in connection with the Offer and the Merger and the Option Conversion Ratio (and, in the case of any of the matters described in this Section 2.19 with respect to Company Common Stock, the Per Share Cash Consideration) shall each be appropriately adjusted, at any time and from time to time, to fully reflect the effect of any stock split or combination (splitsing van aandelen, samenvoeging van (verschillende klassen van) aandelen), recapitalization (omzetting reserves in aandelenkapitaal), stock dividend, reclassification, redenomination, adjustment of par value, exchange of shares, reorganization, recapitalization or other like change with respect to Parent Common Shares or, if permitted by the terms of Section 5.1(f), Company Common Stock, as the case may be, occurring (or for which a record date occurs) during the Interim Period.
- 2.20 No Fractional Shares. No certificate or scrip representing fractional Parent Common Shares shall be issued in connection with the Offer or the Merger, and such fractional share interests will not entitle the owner thereof to vote or to any other rights of a stockholder of Parent. Each holder of shares of Company Common Stock exchanged pursuant to the Offer or the Merger who would otherwise be entitled to receive a fraction of a share of Parent Common Shares (after taking into account all Company Certificates delivered by such holder) shall receive from Parent, in lieu thereof, cash (without interest) in an amount equal to such fractional part of a Parent Common Share multiplied by the Closing Average. For purposes of this Agreement, the Closing Average shall be the average closing sale price per share of Parent Common Shares (rounded up to the nearest cent) on the NASDAQ Global Select Market (NGSM) for the ten (10) consecutive trading days ending on the second-to-last trading day immediately prior to the Acceptance Date, in the case of the Offer or the Effective Time, in the case of the Merger.

3. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedule provided by the Company to Parent on the date hereof, or in a reference to a Company SEC Report filed and publicly available prior to the date of this Agreement referred to in such disclosure schedule, provided by the Company to Parent on the date hereof (the <u>Company Disclosure Schedule</u>), the Company represents and warrants to Parent that the statements contained in this Section 3 are true and correct. The Company Disclosure Schedule shall be arranged in paragraphs corresponding to the numbered and lettered paragraphs contained in this Section 3, and the disclosure in any paragraph shall qualify (A) the corresponding paragraph of this Section 3, and (B) the other paragraphs of this Section 3 to the extent that it is readily apparent from a reading of the Company Disclosure Schedule, without reference to anything other than the Company Disclosure Schedule, that it also qualifies or applies to such other paragraphs. As used in this Agreement, a <u>Company Material Adverse Effect</u> means any change, event or effect that is materially adverse to the business, assets (including intangible assets), financial condition, or results of operations of the Company and its Subsidiaries, taken as a whole, or to the Company Sability to consummate the transactions contemplated in this Agreement, provided that none of the following shall constitute a Company Material Adverse Effect:

(a) general economic conditions worldwide, in the United States, or in any nation or region in which the Company or any of its Subsidiaries has a substantial presence or operations, provided that such conditions do not disproportionately affect the Company and its Subsidiaries relative to other industry participants;

A-15

Table of Contents

- (b) any acts of terrorism not directed at the Company or any of its Subsidiaries or any outbreak of war, provided that such conditions do not disproportionately affect the Company and its Subsidiaries relative to other industry participants;
- (c) the public announcement by the Parties of this Agreement, the pendency of the Merger and the other transactions contemplated hereby, or any action taken which is required by this Agreement or specifically requested by Parent;
- (d) factors generally affecting the industries or markets in which the Company and its Subsidiaries operate, provided that such factors do not disproportionately affect the Company and its Subsidiaries relative to other industry participants;
- (e) changes in Law or GAAP or the interpretation thereof, provided that such changes do not disproportionately affect the Company and its Subsidiaries relative to other industry participants; or
- (f) a decline in the trading price or change in trading volume of the Company Common Stock; provided that this clause will not exclude any underlying change, event, circumstance, development or effect that may have resulted in, or contributed to, a decline in trading price or change in trading volume.
- 3.1 Organization and Qualification. The Company is a corporation duly incorporated, validly existing and in corporate good standing under the Laws of the State of Delaware. The Company is duly qualified or licensed as a foreign corporation to conduct business, and is in corporate good standing, under the Laws of each jurisdiction where the character of the properties owned, leased or operated by it, or the nature of its activities, makes such qualification or licensing necessary, except where the failure to be so qualified, licensed or in good standing, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect. Complete and correct copies of the Company s Certificate of Incorporation and Bylaws, each as in effect on the date of this Agreement, are on file as exhibits to the Company SEC Reports.
- 3.2 Subsidiaries.
- (a) Except as set forth in Section 3.2(a) of the Company Disclosure Schedule, Exhibit 21 to the Company s Annual Report on Form 10-K for the fiscal year ended June 30, 2006 sets forth a complete and correct list of each Subsidiary of the Company as of the date of this Agreement.
- (b) Each material Subsidiary of the Company is a corporation duly organized, validly existing and in corporate good standing (to the extent such concepts are applicable) under the Laws of the jurisdiction of its incorporation, and is duly qualified or licensed as a foreign corporation to conduct business, and is in corporate good standing (to the extent such concepts are applicable), under the Laws of each jurisdiction where the character of the properties and other assets owned, leased or operated by it, or the nature of its activities, makes such qualification or licensing necessary, except where the failure to be so qualified, licensed or in good standing, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect.
- (c) All of the issued and outstanding shares of capital stock of, or other equity interests in, each material Subsidiary of the Company are: (i) duly authorized, validly issued, fully paid, non-assessable (to the extent such concepts are applicable); (ii) owned, directly or indirectly, by the Company (other than directors—qualifying shares in the case of foreign Subsidiaries) free and clear of all liens, claims, security interests, pledges and similar encumbrances (collectively, <u>Liens</u>) and (iii) free of any restriction which prevents the payment of dividends to the Company or any other Subsidiary of the Company, or which otherwise restricts the right to vote, sell or otherwise dispose of such capital stock or other ownership interest, other than restrictions under the Securities Act of 1933, as amended (the <u>Securities Act</u>) and state securities Laws.
- (d) None of the Company s Subsidiaries is required to file any forms, reports or other documents with the U.S. Securities and Exchange Commission (the SEC).

A-16

Table of Contents

- (e) For purposes of this Agreement, the term <u>Subsidiary</u> means, with respect to any Party, any corporation or other organization, whether incorporated or unincorporated, of which (i) such Party (or any other Subsidiary of such Party) is a general partner or (ii) at least a majority of the securities or other equity interests having by their terms ordinary voting power to elect a majority of the Board of Directors or others performing similar functions with respect to such corporation or other organization, is directly or indirectly owned or controlled by such Party or by any one or more of its Subsidiaries, or by such Party and one or more of its Subsidiaries.
- 3.3 Capital Structure.
- (a) The authorized capital stock of the Company as of the date of this Agreement consists of (i) 50,000,000 shares of Company Common Stock and (ii) 1,000,000 shares of Preferred Stock, \$0.10 par value per share (<u>Company Preferred Stock</u>).
- (b) As of the close of business on May 30, 2007: (i) 24,415,240 shares of Company Common Stock were issued and outstanding; (ii) no shares of Company Preferred Stock were issued or outstanding; (iii) no shares of Company Common Stock were held in the treasury of the Company and (iv) 2,060,398 shares of Company Common Stock were duly reserved for future issuance upon the exercise of equity awards made or granted on or prior to the date hereof (the <u>Company Equity Awards</u>) pursuant to the Amended and Restated Equity Incentive Plan, the Amended and Restated Directors Equity Compensation Plan, the Amended and Restated Omnibus Plan and the Amended and Restated 1997 Stock Option Plan (collectively, the <u>Company Equity Plans</u>), 206,689 shares reserved to meet the requirement under the Company Equity Plans for double reservation of all Restricted Stock Unit Awards and Performance Shares Awards made on or after October 26, 2006 which shares, while reserved, are not and shall not be issuable under the Company Equity Plans and 1,400,163 shares of Company Common Stock were duly reserved for future grants or awards made after the date hereof pursuant to the Company Equity Plans. Except as described above, as of the close of business on the last Business Day prior to the date hereof, there were no shares of voting or non-voting capital stock, equity interests or other securities of the Company authorized, issued, reserved for issuance or otherwise outstanding.
- (c) All outstanding shares of Company Common Stock are, and all shares which may be issued pursuant to the Company Equity Plans and the Company Equity Awards will be, when issued against payment therefor in accordance with the terms thereof, duly authorized, validly issued, fully paid and non-assessable, and not subject to, or issued in violation of, any preemptive, subscription or any kind of similar rights.
- (d) There are no bonds, debentures, notes or other indebtedness of the Company having the right to vote (or convertible into securities having the right to vote) on any matters on which stockholders of the Company may vote. Except as described in subsection (b) above, there are no outstanding securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind (contingent or otherwise) to which the Company is a party or bound obligating the Company to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of the Company or obligating the Company to issue, grant, extend or enter into any agreement to issue, deliver or sell any such capital stock or securities. Neither the Company nor any Subsidiary of the Company is subject to any obligation or requirement to provide material funds for, make any guarantee with respect to the obligations of or to make any material investment (in the form of a loan or capital contribution) in, any Person (as defined in Section 3(a)(9) of the Securities Exchange Act of 1934, as amended (the __Exchange Act)).
- (e) Section 3.3(e) of the Company Disclosure Schedule contains a complete and correct list of the holders of all Company Equity Awards outstanding as of May 30, 2007, including, for stock option awards (<u>Company Stock Options</u>) and any other Company Equity Award if applicable: (i) the date of grant; (ii) the exercise price; (iii) the vesting schedule and expiration date and (iv) any terms regarding the acceleration of vesting (other than those requiring acceleration of vesting in connection with the Merger as set forth in the applicable Company Equity Plan).

A-17

Table of Contents

- (f) All grants of Company Equity Awards were made in compliance with the terms of the applicable Company Equity Plan under which such Company Equity Awards were made and with applicable Laws. All Company Equity Awards have been properly accounted for in the Company s financial statements and reported in compliance with GAAP and applicable Tax requirements and the published rules and regulations of the SEC with respect thereto, and there is no basis for any claim that the grant date of any Company Stock Option is inaccurate or the exercise price thereof is improper.
- (g) All of the issued and outstanding shares of Company Common Stock were issued in compliance in all material respects with all applicable federal and state securities Laws.
- (h) There are no outstanding contractual obligations of the Company to repurchase, redeem or otherwise acquire any shares of capital stock (or options to acquire any such shares) or other security or equity interests of the Company, other than rights of forfeiture of Company Common Stock pursuant to Restricted Stock Unit Awards or Performance Shares Awards made pursuant to the Company Equity Plans. Except as described in Section 3.3(e) of the Company Disclosure Schedule, there are no stock-appreciation rights, security-based performance units, phantom stock or other security rights pursuant to which any Person is or may be entitled to receive any payment or other value based on the stock price performance of the Company or to cause the Company or any of its Subsidiaries to file a registration statement under the Securities Act, or which otherwise relate to the registration of any securities of the Company or any of its Subsidiaries.
- (i) Other than as set forth on Section 3.3(i) of the Company Disclosure Schedule, there are no voting trusts, proxies or other agreements, commitments or understandings to which the Company or any of its Subsidiaries or, to the knowledge of the Company, any of the stockholders of the Company is a party or by which any of them is bound with respect to the issuance, holding, acquisition, voting or disposition of any shares of capital stock or other security or equity interest of the Company or any of its Subsidiaries.
- (j) The Company is in compliance in all material respects with the applicable criteria for continued listing of the Company Common Stock on the NGSM, including all applicable corporate governance rules and regulations.
- 3.4 Authority; No Conflict; Required Filings.
- (a) The Company has all requisite corporate power and authority to execute and deliver this Agreement and, subject to the adoption of this Agreement by the affirmative vote of the holders of a majority of the outstanding shares of Company Common Stock in accordance with the DGCL and the Company s Certificate of Incorporation (the Company Requisite Stockholder Approval), and assuming the accuracy of Parent s, NAH s and Merger Sub s representation and warranty set forth in Section 4.19, to perform its obligations hereunder and consummate the Merger and other transactions contemplated hereby. The execution and delivery of this Agreement by the Company and, subject to obtaining the Company Requisite Stockholder Approval, and assuming the accuracy of Parent s, NAH s and Merger Sub s representation and warranty set forth in Section 4.19, the performance by the Company of its obligations hereunder and the consummation by the Company of the Merger and other transactions contemplated hereby, have been duly authorized by all necessary corporate action on the part of the Company.
- (b) This Agreement has been duly executed and delivered by the Company and constitutes a valid and binding obligation of the Company, enforceable against it in accordance with its terms, subject to: (i) the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting the enforcement of creditors—rights generally and (ii) general equitable principles (whether considered in a proceeding in equity or at law) (collectively, the <u>Bankruptcy and Equitable Exceptions</u>).
- (c) The execution and delivery of this Agreement by the Company do not, and the performance by the Company of its obligations hereunder and the consummation by the Company of the Merger and other transactions contemplated hereby will not, conflict with or result in any violation of, or default (with or

A-18

Table of Contents

without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or to a loss of a material benefit, or result in the creation of any Liens in or upon any of the properties or other assets of the Company or any of its Subsidiaries under any provision of: (i) the Certificate of Incorporation or Bylaws of the Company; (ii) subject to the governmental filings and other matters referred to in paragraph (d) below, any (A) permit, license, franchise or Law or (B) judgment, decree or order, in each case applicable to the Company or any of its Subsidiaries, or by which any of their respective properties or assets is bound or (iii) except as set forth in Section 3.4(c) of the Company Disclosure Schedule, any loan or credit agreement, note, bond, mortgage, indenture, contract, agreement, lease or other instrument or obligation to which the Company or any of its Subsidiaries is a party or by which any of their respective properties is bound other than in the case of Subsections (c)(ii) and (iii) those exceptions that would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

- (d) No consent, approval, order or authorization of, or registration, declaration or filing with, any government, governmental, statutory, regulatory or administrative authority, agency, body or commission or any court, tribunal or judicial body, whether federal, state, local or foreign (each, a Governmental Authority) is required by the Company or any of its Subsidiaries in connection with the execution and delivery by the Company of this Agreement or the consummation by the Company of the Merger and other transactions contemplated hereby except for: (i) compliance with any applicable requirements under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the HSR Act); (ii) the filing with the European Commission of a merger notification in accordance with Council Regulation (EC) 139/2004, the E.C. Merger Regulation (the ECMR); (iii) the applicable requirements of the competent authority of any member state of the European Union to which any of the transactions contemplated by this Agreement is referred pursuant to Article 9 of the ECMR; (iv) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware in accordance with the DGCL and appropriate corresponding documents with the appropriate authorities of other states in which the Company is qualified as a foreign corporation to transact business and (v) any notices required under the U.S. Federal Food, Drug, and Cosmetic Act, as amended (the FDA Act), other than those consents, approvals, orders or authorizations of, or registrations, declarations or filings with, any Governmental Authority that would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.
- 3.5 Board Approval; Section 203; Required Vote.
- (a) The Company Board has, at a meeting duly called and held, by a unanimous vote of all directors: (i) approved and declared advisable this Agreement; (ii) determined that the Transactions are advisable, fair to and in the best interests of the Company and its stockholders; (iii) adopted a resolution that includes the Recommendation and (iv) directed that this Agreement be submitted to the stockholders of the Company for their adoption.
- (b) The Company Board has taken all actions so that the restrictions contained in Section 203 of the DGCL applicable to a business combination (as defined therein) will not apply to the execution, delivery or performance of this Agreement or the consummation of the Merger or other transactions contemplated by this Agreement.
- (c) Assuming the accuracy of Parent s, NAH s and Merger Sub s representation and warranty set forth in Section 4.19, the Company Requisite Stockholder Approval is the only vote of the holders of any class or series of capital stock of the Company necessary to adopt this Agreement.
- 3.6 SEC Filings; Sarbanes-Oxley Act.
- (a) Since July 1, 2004, the Company has timely filed all forms, reports and documents required to be filed by the Company with the SEC, including all exhibits required to be filed therewith (including any forms, reports and documents filed after the date hereof, as filed, the Company SEC Reports). The Company SEC Reports: (i) at the time filed complied (or will comply when filed, as the case may be) as to

A-19

Table of Contents

form in all material respects with the applicable requirements of the Securities Act and/or the Exchange Act, as the case may be and (ii) did not at the time they were filed (or, if later filed, amended or superseded, then on the date of such later filing) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements contained therein, in the light of the circumstances under which they were made, not misleading.

- (b) Each of the consolidated financial statements (including, in each case, any related notes thereto) contained in the Company SEC Reports (collectively, the Company Financial Statements), at the time filed and amended, (i) complied or will comply, as the case may be, as to form in all material respects with the applicable published rules and regulations of the SEC with respect thereto; (ii) was or will be prepared in accordance with U.S. generally accepted accounting principles (GAAP) applied on a consistent basis throughout the periods involved, except as may otherwise be indicated in the notes thereto or, in the case of unaudited interim financial statements, as permitted by Form 10-Q promulgated by the SEC and (iii) fairly presented or will fairly present, as the case may be, in all material respects, the consolidated financial position of the Company and its Subsidiaries as of the dates indicated and the consolidated results of operations and cash flows for the periods therein indicated, except, in the case of the unaudited interim financial statements, as permitted by Form 10-Q promulgated by the SEC.
- (c) The management of the Company has established and maintains disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) that are designed to provide reasonable assurance that information (both financial and non-financial) required to be disclosed by Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC and that all such information is accumulated and communicated to Company s management as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the Chief Executive Officer and Chief Financial Officer of Company required under the Exchange Act with respect to such reports. As of March 31, 2007, there were no material weaknesses in Company s or any of the Company Subsidiaries internal controls as contemplated under Section 404 of the Sarbanes-Oxley Act of 2002 (SOX). The Company has disclosed, based on the most recent evaluation of its internal control over financial reporting (as defined in Rule 13a-15(f) of the Exchange Act) by its chief executive officer and chief financial officer, to the Company s auditors and the audit committee of the Company Board (a) any significant deficiencies in the design or operation of its internal control over financial reporting that are reasonably likely to adversely affect the Company s ability to record, process, summarize and report financial information and has identified for the Company s auditors and audit committee of the Company Board any material weaknesses in its internal control over financial reporting and (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting. Since the date of the Company s most recent evaluation of internal control over financial reporting, to the knowledge of the Company, no facts or circumstances have arisen or occurred that would be required to be disclosed to the Company s auditors or the Company s audit committee regarding (x) a significant deficiency in the design or operation of its internal control over financial reporting, (y) a material weakness in its internal control over financial reporting or (z) fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting.
- (d) Neither the Company nor any of its Subsidiaries has outstanding, or has arranged any outstanding, extensions of credit to directors or executive officers within the meaning of Section 402 of SOX. To the Company s knowledge, there are no other violations of the Company s code of conduct, adopted pursuant to NASDAQ Rule 4350(n). The Company has delivered to Parent complete and correct copies of any written complaints, reports or allegations, and a reasonably detailed summary of any verbal complaints, reports or allegations, that have been submitted or made by any party to the Audit Committee of the Company Board since July 1, 2005 pursuant to the procedures established in accordance with Section 10A(m)(4) of the Exchange Act. The Company has delivered to Parent complete and correct copies of any written complaints, reports or allegations, and a reasonably detailed summary of any verbal complaints, reports or allegations,

A-20

Table of Contents

made or submitted by any attorney representing the Company or any of its Subsidiaries to the Company s chief executive officer, any supervising attorney or the Company Board or any committee thereof, including any qualified legal compliance committee, as contemplated by the rules set forth in 17 CFR Part 205. Neither the Company nor any of its Subsidiaries is a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar arrangement, including without limitation any off-balance sheet arrangement (as defined in Item 303(a) of Regulation S-K promulgated by the SEC), where the result, purpose or intended effect of such contract or arrangement is to avoid disclosure of any material transaction involving, or material liabilities of, the Company or any of its Subsidiaries in any Company Financial Statement or Company SEC Report.

- 3.7 Absence of Undisclosed Liabilities. The Company and its Subsidiaries do not have any material liabilities or obligations, whether fixed, contingent, accrued or otherwise, liquidated or unliquidated and whether due or to become due, in each case of a nature required by GAAP to be reflected on a consolidated balance sheet of the Company, other than liabilities (i) reflected or reserved against on the balance sheet contained in the Company s Form 10-Q (the Most Recent Balance Sheet) filed with the SEC on May 9, 2007, (ii) described in Section 3.7(ii) of the Company Disclosure Schedule and (iii) liabilities or obligations incurred since March 31, 2007 (the Most Recent Balance Sheet Date) in the ordinary course of business consistent with past practice, none of which has had or reasonably would be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 3.8 Absence of Certain Changes or Events. Since the Most Recent Balance Sheet Date, the Company and its Subsidiaries have conducted their respective businesses only in the ordinary course of business consistent with past practice, and there has not been: (i) any action, event or occurrence which would reasonably be expected to have a Company Material Adverse Effect or (ii) except as described in Section 3.8 of the Company Disclosure Schedule, any other action, event or occurrence that would have required the consent of Parent pursuant to Section 5.1 had such action, event or occurrence taken place after the execution and delivery of this Agreement.
- 3.9 Agreements, Contracts and Commitments.
- (a) The Company has made available to Parent a complete and correct copy of all contracts and agreements to which the Company is a party as of the date of this Agreement that are material to the business, assets (tangible or otherwise), financial condition or results of operations of the Company and its Subsidiaries, taken as a whole, including without limitation (i) agreements listed on Section 3.9 of the Company Disclosure Schedule, (ii) any agreement, contract or commitment in connection with which or pursuant to which the Company and its Subsidiaries will spend or receive (or are reasonably expected to spend or receive), in the aggregate, more than \$500,000 during the current fiscal year or during the next fiscal year; (iii) any non-competition or other agreement that prohibits or otherwise restricts the Company or any of its Subsidiaries from freely engaging in business anywhere in the world; (iv) any material contract (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) with respect to the Company and its Subsidiaries, and (v) any employment or consulting agreement with any executive officer or other employee of the Company or member of the Company Board earning an annual base salary in excess of \$100,000, other than those that are terminable by the Company or any of its Subsidiaries on no more than thirty (30) days notice without material liability or financial obligation to the Company or any of its Subsidiaries (collectively, the Company Material Contracts).
- (b) Each Company Material Contract is valid, binding and enforceable against the Company and its Subsidiaries in accordance with its terms, subject to the Bankruptcy and Equitable Exceptions. Neither the Company nor any of its Subsidiaries is in breach, or has received in writing any claim that it is in breach, of any of the terms or conditions of any Company Material Contract in such a manner as would permit any other party thereto to cancel or terminate the same or to collect material damages from the Company or any of its Subsidiaries.
- (c)(i) Each Company Material Contract that has not expired or otherwise been terminated in accordance with its terms is in full force and effect and (ii) to the knowledge of the Company, no other party to such contract is in default.

A-21

Table of Contents

- 3.10 Compliance with Laws; Regulatory Matters.
- (a) The Company and its Subsidiaries are conducting and have conducted their business and operations in material compliance with the Federal Food, Drug, and Cosmetic Act (the <u>FD&C Act</u>), 21 U.S.C. §301 et. seq., and all applicable regulations promulgated by the United States Food and Drug Administration (<u>FDA</u>) (collectively, <u>FDA Law and Regulation</u>) and comparable foreign regulatory or Governmental Authorities.
- (b) Except as set forth in Section 3.10(b) of the Company Disclosure Schedule, neither the Company nor its Subsidiaries has received any notice or communication from the FDA alleging noncompliance with any applicable FDA Law and Regulation. The Company and its Subsidiaries are not subject to any enforcement, regulatory or administrative proceedings by the FDA or any comparable foreign regulatory or Governmental Authority and, to the knowledge of the Company, no such proceedings have been threatened. There is no civil, criminal or administrative action, suit, demand, claim, complaint, hearing, investigation, demand letter, warning letter, proceeding or request for information pending against Company or the Company Subsidiaries, and, to the Company s knowledge, the Company and its Subsidiaries have no liability (whether actual or contingent) for failure to comply with any FDA Law and Regulation or any law or regulation of a comparable foreign regulatory or Governmental Authority. There is no act, omission, event, or circumstance of which the Company or its Subsidiaries have knowledge that would reasonably be expected to give rise to or lead to any such action, suit, demand, claim, complaint, hearing, investigation, notice, demand letter, warning letter, proceeding or request for information or any such liability. There has not been any violation of any FDA Law and Regulation or any law or regulation of a comparable foreign regulatory or Governmental Authority by the Company or its Subsidiaries in their product development efforts, submissions, record keeping and reports to FDA or a comparable Governmental Authority that could reasonably be expected to require or lead to investigation, corrective action or enforcement, regulatory or administrative action that would result in a Material Adverse Effect. To the knowledge of Company, there are no civil or criminal proceedings relating to the Company or its Subsidiaries or any Company or Subsidiary employee which involve a matter within or related to the FDA s jurisdiction or the jurisdiction of any comparable Governmental Authority.
- (c) Each of the products and product candidates of the Company and its Subsidiaries is being, and at all times has been, developed, tested, manufactured, labeled, promoted and stored, as applicable, in material compliance with FDA Law and Regulation and requirements of comparable foreign regulatory and Governmental Authorities, including those requirements relating to good manufacturing practice, good laboratory practice and good clinical practice.
- (d) The Company has made available to Parent each annual report filed by any of the Company and its Subsidiaries with the FDA and any similar state or foreign regulatory or Governmental Authority with respect to any products and product candidates of the Company or its Subsidiaries.
- (e) The clinical trials (including any post-marketing studies), animal studies and other preclinical tests conducted by the Company or its Subsidiaries were, and if still pending, are, being conducted in all material respects in accordance with all experimental protocols, informed consents, procedures and controls of the Company and its Subsidiaries and applicable FDA and foreign requirements including, but not limited to, good clinical practices, good laboratory practices, and FDA Law and Regulation including the protection of human subjects. No clinical trials or studies conducted by the Company or its Subsidiaries have been conducted that were not in compliance with the Declaration of Helsinki. Neither the Company nor its Subsidiaries has received any written notice from the FDA or any other regulatory or Governmental Authority requiring the termination or suspension or material modification of any animal study, preclinical study or clinical trial conducted by or on behalf of the Company or any Subsidiary.
- (f) Neither the Company nor any of its Subsidiaries is subject to any pending or, to the knowledge of the Company, threatened investigation by (A) the FDA, (B) Department of Health and Human Services Office of Inspector General or Department of Justice pursuant to the Federal Healthcare Program Anti-Kickback Statute (42 U.S.C. Section 1320a-7b(b) (known as the <u>Anti-Kickback Statute</u>) or the Federal False Claims Act (31 U.S.C. Section 3729), or (C) any equivalent state or foreign Law.

A-22

Table of Contents

- (g) To the knowledge of the Company, no officer, employee or agent of the Company or the Company Subsidiaries, has been convicted of any crime or engaged in any conduct for which such person or entity could be excluded from participating in the federal health care programs under Section 1128 of the Social Security Act or any similar law or regulation.
- (h) The Company is not debarred or otherwise excluded from or restricted in any manner from participation in, any government program (<u>Debarred</u>). To the knowledge of the Company, no officer, employee or agent of the Company has been convicted of any crime or engaged in any conduct for which debarment is mandated or permitted by 21 U.S.C. § 335a. To the knowledge of the Company, no officer, employee or agent of the Company has been convicted of any crime or engaged in any conduct for which such person or entity could be excluded from participating in the federal health care programs under Section 1128 of the Social Security Act or any similar Law. To the knowledge of the Company, neither the Company nor its Subsidiaries employs or uses the services of any individual or entity that is or, during the time when such individual or entity was employed by or providing services to the Company or any of its Subsidiaries, Debarred.
- (i) With respect to all third party manufacturers and suppliers of key raw materials used by the Company or its Subsidiaries (each a <u>Third Party Manufacturer</u>), the Company has inspected all Third Party Manufacturers and to its knowledge, each such Third Party Manufacturer:
- (i) has complied and is complying, in each case in all material respects, with all Laws, including the FD&C Act and any similar state or foreign Laws:
- (ii) has all material permits to perform its obligations as a Third Party Manufacturer and all such permits are in full force and effect; and
- (iii) has not been Debarred from participation in any government program and does not employ or use the services of any individual or entity that is or, during the time when such person or entity was providing services as a Third Party Manufacturer to the Company or any of its Subsidiaries, was Debarred.
- (j) To the knowledge of the Company, no officer, employee or agent of the Company or the Company Subsidiaries has: made any untrue statement of material fact or fraudulent statement to the FDA or any other Governmental Entity; failed to disclose a material fact required to be disclosed to the FDA or any other Governmental Entity; or committed an act, made a statement, or failed to make a statement that would reasonably be expected to provide the basis for the FDA or any other Governmental Entity to invoke its policy respecting Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities, as set forth in 56 Fed.Reg. 46191 (September 10, 1991).
- (k) Neither the Company nor its Subsidiaries has knowingly or willfully solicited, received, paid or offered to pay any remuneration, directly or indirectly, overtly or covertly, in cash or kind for the purpose of making or receiving any referral which violated any applicable anti-kickback or similar Law, including the Anti-Kickback Statute, or any applicable state or foreign anti-kickback Law.
- (1) Section 3.10(1) of the Company Disclosure Schedule lists, as of the date of this Agreement, all written claims or statements (including all correspondence, other than immaterial correspondence, with Governmental Authorities and/or Medicare Administrative Contractors) concerning or relating to any health care program that involves, relates to or alleges (i) any material violation of any applicable rule, regulation, policy or requirement of any such program with respect to any activity, practice or policy of the Company or its Subsidiaries or (ii) any violation of any applicable rule, regulation, policy or requirement of any such program with respect to any claim for payment or reimbursement made by the Company or its Subsidiaries or any payment or reimbursement paid to the Company or its Subsidiaries. Except as set forth on Section 3.10(1) of the Company Disclosure Schedule, there are no such violations nor, to the knowledge of the Company, are there any grounds to reasonably anticipate the commencement of any investigation or inquiry, or the assertion of any claim or demand by, any Governmental Authorities and/or Medicare Administrative Contractors with respect to any of the activities, practices, policies or claims of the Company

A-23

Table of Contents

or its Subsidiaries, or any payments or reimbursements claimed by the Company or its Subsidiaries, in each case concerning or relating to any federal or state government funded health care program. Neither the Company nor its Subsidiaries is, as of the date hereof, subject to any outstanding audit by any Governmental Authority, intermediary or carrier and, to the knowledge of the Company, there are no grounds to anticipate any such audit, except such audits in the ordinary course of review, in the foreseeable future.

- (m) To the knowledge of the Company, neither the Company nor any of its Subsidiaries has submitted any claim for payment in violation of any Laws relating to false claim or fraud, including without limitation the Federal False Claim Act, 31 U.S.C. § 3729, or any applicable state or foreign false claim or fraud Law.
- (n) Neither the Company nor any of its Subsidiaries has failed to comply with any applicable security and privacy standards regarding protected health information under the Health Insurance Portability and Accountability Act of 1996, including the regulations promulgated thereunder (collectively <u>HIPA</u>A), or any applicable state or foreign privacy Laws.
- (o) Neither the Company nor its Subsidiaries nor, to the knowledge of the Company, any of their respective officers, directors or employees, acting in their capacities as such, is or has been involved in any activities which are, or are alleged in writing by any qui tam relator or Governmental Authority to be, prohibited under the federal Medicare and Medicaid statutes, which are specifically defined as 42 U.S.C. §§ 1320a-7, 1320a-7a, 1320a-7b, 1395nn, 18 U.S.C. §1347, § 287, §1001, and § 1035, or the federal TRICARE statute, or the regulations promulgated pursuant to such federal statutes.
- (p) The Company and each of its Subsidiaries has conducted its export transactions in accordance in all material respects with applicable provisions of United States export control Laws, including but not limited to the FD&C Act, Export Administration Act and implementing Export Administration Regulations. Without limiting the foregoing: (i) the Company and each of its Subsidiaries has obtained all material export licenses and other approvals required for its exports of products and technologies from the United States; (ii) the Company and each of its Subsidiaries is in compliance in all material respects with the terms of all applicable material export licenses or other approvals; (iii) there are no pending or, to the knowledge of the Company, threatened claims against the Company or any of its Subsidiaries with respect to such material export licenses or other approvals; (iv) there are no actions, conditions or circumstances pertaining to the Company s or any of its Subsidiary s material export transactions that would reasonably be expected to give rise to any future claims and (v) no consents or approvals will be required in connection with the Merger.
- (q) Each of the Company and its Subsidiaries has submitted to the FDA and all comparable foreign regulatory or Governmental Authorities for and received approval of all material registrations, applications, licenses, requests for exemptions, permits and other regulatory authorizations (the <u>FD&C Act Permits</u>) necessary to conduct the business of the Company and its Subsidiaries as currently conducted. The Company and its Subsidiaries are in material compliance with all such FD&C Act Permits. To the knowledge of the Company, no suspension, revocation, cancellation or withdrawal of the FD&C Act Permits is threatened and there is no basis for believing that such FD&C Act Permits will not be renewable upon expiration or will be suspended, revoked, cancelled or withdrawn.
- 3.11 *Material Permits*. Each of the Company and its Subsidiaries holds all material federal, state, local and foreign governmental licenses, permits, franchises and authorizations necessary for conduct of its business as presently conducted and the ownership and operation of its properties and other assets, including those that are required under all Environmental Laws, in each case (whether under Environmental Laws or otherwise) (such licenses, permits, franchises and authorizations, the <u>Material Permits</u>) The Company and its Subsidiaries are in material compliance with all such Material Permits. To the knowledge of the Company, no suspension, revocation, cancellation or withdrawal of the Material Permits is threatened and there is no basis for believing that such Material Permits or will not be renewable upon expiration or will be suspended, revoked, cancelled or withdrawn.
- 3.12 Litigation and Product Liability. Except as set forth in Section 3.12 of the Company Disclosure Schedule, there is no suit, action, arbitration, claim, governmental or other proceeding before any Governmental

A-24

Table of Contents

Authority pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries. No material product liability claims have been asserted in writing or, to the knowledge of the Company, threatened against the Company in respect of any product or product candidate tested, researched, developed, manufactured, marketed, distributed, or sold by, on behalf of, or in cooperation with the Company. There are currently pending no internal investigations or inquiries being conducted by the Company, the Company Board (or any committee thereof) or any third party at the request of any of the foregoing concerning any financial, accounting, option grant practices, Tax, conflict of interest, illegal activity, fraudulent or deceptive conduct or other misfeasance or malfeasance issues (each, an Internal Investigation). The Company has delivered to Parent complete and correct copies of any written reports, submissions, correspondence, findings or other documentation with respect to any Internal Investigation, or a reasonably detailed summary of any such Internal Investigation to the extent that a written report with respect thereto was not or has not been prepared, whether such Internal Investigation is currently pending or was conducted in the past since July 1, 2004.

3.13 *Restrictions on Business Activities*. There is no agreement, judgment, injunction, order or decree binding upon the Company or any of its Subsidiaries which has the effect of prohibiting or materially impairing (a) any current or future business practice of the Company or any of its Subsidiaries or (b) any acquisition of any Person or property by the Company or any of its Subsidiaries.

3.14 Employee Benefit Matters.

(a) Section 3.14(a) of the Company Disclosure Schedule lists all employee benefit plans (as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (_ERISA_) and all material bonus, stock or other security option, stock or other security purchase, stock or other security appreciation rights, incentive, deferred compensation, retirement or supplemental retirement, severance, golden parachute, vacation, cafeteria, dependent care, medical care, employee assistance program, education or tuition assistance programs, and all material insurance and other similar fringe or employee benefit plans, programs or arrangements for the benefit of, or relating to, any present or former employee or director of the Company or any ERISA Affiliate under which the Company or any ERISA Affiliate has any current liability (together, the _Company Employee Plans_). The Company has made available to Parent correct and complete copies of (where applicable) (i) all plan documents, summary plan descriptions, summaries of material modifications and amendments related to such plans; (ii) the most recent determination or opinion letters received from the Internal Revenue Service (_IRS_); (iii) the most recent Form 5500 Annual Report; (iv) the most recent audited financial statement and actuarial valuation, if applicable, and (v) all material agreements and insurance contracts relating to each such Company Employee Plan. For purposes of this Agreement, _ERISA Affiliate means or any trade or business (whether or not incorporated) which is a member of a controlled group or which is under common control with the Company within the meaning of Section 414 of the Code.

(b) There has been no prohibited transaction, as such term is defined in Section 406 of ERISA and Section 4975 of the Internal Revenue Code of 1986, as amended (<u>Code</u>), with respect to any Company Employee Plan that would reasonably be expected to result in a material liability to the Company or any ERISA Affiliate; (ii) there are no claims pending (other than routine claims for benefits) or, to the knowledge of the Company, threatened against any Company Employee Plan or against the assets of any Company Employee Plan that could result in material liability to the Company or an ERISA Affiliate, nor are there any current or, to the knowledge of the Company, threatened Liens on the assets of any Company Employee Plan; (iii) all Company Employee Plans have been maintained in material compliance with their terms and the requirements prescribed by any and all applicable Laws; (iv) the Company and ERISA Affiliates have performed all material obligations required to be performed by them under, are not in default under or violation of, and the Company has no knowledge of any material default or violation by any other party with respect to, any of the Company Employee Plans; (v) each Company Employee Plan intended to qualify under Section 401(a) of the Code and each corresponding trust exempt under Section 501 of the Code has received or is the subject of a favorable determination or opinion letter from the IRS, and to the knowledge of the Company, nothing has occurred which would reasonably be expected to cause the loss of

A-25

Table of Contents

such qualification or exemption; (vi) all contributions required to be made to any Company Employee Plan pursuant to Section 412 of the Code or otherwise, the terms of the Company Employee Plan or any collective bargaining agreement, have been made and a reasonable amount has been accrued for contributions to each Company Employee Plan for the current plan years to the extent required by GAAP; (vii) other than as set forth on Section 3.14(b)(vii) of the Company Disclosure Schedule, the transactions contemplated herein will not directly or indirectly result in an increase of benefits, acceleration of vesting or acceleration of timing for payment of any benefit to any participant in or beneficiary of any Company Employee Plan; (viii) each Company Employee Plan is maintained outside of the United States has been operated in material conformance with the applicable Laws in the jurisdictions in which such Company Employee Plan is present or operates and, to the extent relevant, the United States and (ix) neither the Company nor any ERISA Affiliate has ever made a complete or partial withdrawal from a Multiemployer Plan (as such term is defined in Section 3(37) of ERISA) resulting in withdrawal liability (as such term is defined in Section 4201 of ERISA), without regard to any subsequent waiver or reduction under Section 4207 or 4208 of ERISA.

- (c) No Company Employee Plan is an employee pension benefit plan (within the meaning of Section 3(2) of ERISA) subject to Title IV of ERISA, and neither the Company nor any ERISA Affiliate has ever partially or fully withdrawn from any such plan. No Company Employee Plan is a Multiemployer Plan or single-employer plan under multiple controlled groups as described in Section 4063 of ERISA, and neither the Company nor any ERISA Affiliate has ever contributed to or had an obligation to contribute, or incurred any liability in respect of a contribution, to any Multiemployer Plan or multiple employer plan (within the meaning of Section 3(40) of ERISA or Section 413 of the Code).
- (d) Each Company Employee Plan that is a group health plan (within the meaning of Section 5000(b)(1) of the Code) has been operated in material compliance with applicable Law, its terms, and with the group health plan continuation coverage requirements of Section 4980B of the Code and Sections 601 through 608 of ERISA (<u>COBRA Coverage</u>), Section 4980D of the Code and Sections 701 through 707 of ERISA, to the extent such requirements are applicable. No Company Employee Plan or written or oral agreement exists which obligates the Company or any ERISA Affiliate to provide retiree medical, health or life benefits, other than COBRA Coverage or as required by any similar applicable Law.
- (e) No Company Employee Plan which is an employee welfare benefit plan under Section 3(1) of ERISA is funded by a trust or is subject to Section 419 or 419A of the Code.
- (f) Except as set forth on Section 3.14(f) of the Company Disclosure Schedule, no payment in connection with the Merger will constitute an excess parachute payment within the meaning of Section 280G of the Code.
- (g) With respect to each Company Employee Plan, (A) other than restrictions under the Code and ERISA, there are no restrictions on the ability of the sponsor of each Company Employee Plan to amend or terminate any Company Employee Plan, the Company has expressly reserved in itself the right to amend, modify or terminate any such Company Employee Plan, or any portion of it, and has made no material written representations which would conflict with or contradict such reservation or right (other than in connection with the grant of outstanding awards and vested rights under such plan) and (B) the Company has satisfied any and all bond coverage requirements of ERISA.
- (h) Other than as set forth on Section 3.14(h) of the Company Disclosure Schedule, no Company Employee Plan or other agreement provides for deferred compensation subject to Section 409A of the Code. The Company Employee Plans listed in Section 3.14(h) of the Company Disclosure Schedule shall be known as Deferred Compensation Plans. No stock option, stock appreciation right or restricted stock award issued by the Company that is currently outstanding is subject to Section 409A of the Code. Each Deferred Compensation Plan is in good faith compliance with Section 409A of the Code and the U.S. Treasury guidance related thereto.

A-26

Table of Contents

- 3.15 Labor and Employment Matters.
- (a)(i) To the knowledge of the Company, there are no material labor grievances pending or, to the knowledge of the Company, threatened in writing between the Company or its Subsidiaries, on the one hand, and any of their respective employees or former employees, on the other hand and (ii) neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement, work council agreement, work force agreement or any other labor union contract applicable to persons employed by the Company or its Subsidiaries, nor, to the knowledge of the Company, are there any activities or proceedings of any labor union to organize any such employees. The Company has not received written notice of any pending or threatened charge, other than any that are immaterial, of (i) an unfair labor practice as defined in the National Labor Relations Act, as amended; (ii) safety violations under the Occupational Safety and Health Act; (iii) wage or hour violations; (iv) discriminatory acts or practices in connection with employment matters or (v) claims by governmental agencies that the Company has failed to comply with any Law relating to employment or labor matters. To the knowledge of the Company, the Company is not currently and has not been the subject of any threatened or actual whistleblower or similar claims by past or current employees or any other persons.
- (b) To the knowledge of the Company, the Company is currently in substantial compliance with all Laws relating to employment, including those related to wages, hours and collective bargaining and is not liable for any arrears of wages or penalties or other sums for failing to comply with any of the foregoing.
- (c)(i) Except as otherwise set forth in Section 3.15(c) of the Company Disclosure Schedule, all contracts of employment to which the Company or any of its Subsidiaries is a party are terminable by the Company or its Subsidiaries on three months or less notice without penalty; (ii) no change of control or other payments to any employee will be triggered as a result of the Merger or this Agreement and (iii) there are no legally binding established practices, plans or policies of the Company or any of its Subsidiaries, requiring the payment of any material amounts or the provision of any material benefits as a result of the termination of employment of any of its employees (whether voluntary or involuntary).
- (d) Since the enactment of the Worker Adjustment and Retraining Notification Act of 1988 (<u>WARN Act</u>), neither the Company nor any of its Subsidiaries has effectuated (i) a plant closing affecting any site of employment or one or more facilities or operating units within any site of employment or facility of the Company or any of its Subsidiaries, or (ii) a mass layoff (as defined in the WARN Act) affecting any site of employment or facility of the Company or any of its Subsidiaries, nor has the Company or any of its Subsidiaries been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar state or local law, in each case that would reasonably be expected to have a Company Material Adverse Effect.
- 3.16 Schedule 14D-9; Information to Be Supplied Proxy Statement.
- (a) Each of the Schedule 14D-9 and the other documents required to be filed by the Company with the SEC in connection with the Offer, the Merger and the other transactions contemplated hereby will comply as to form in all material respects with the requirements of the Exchange Act and the Securities Act, as the case may be. Each of the Schedule 14D-9 and the other documents required to be filed by the Company with the SEC in connection with the Transactions and any of the information supplied or to be supplied by the Company or its Subsidiaries or their representatives for inclusion or incorporation by reference in the Offer Documents, at the time the Offer is commenced and at the Acceptance Date, or, in the case of the Merger Registration Statement (if applicable) will not, on the date of its filing or mailing, at the time of the Company Stockholder Meeting and at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- (b) Notwithstanding the provisions of Section 3.16(a) and (e), no representation or warranty is made by the Company with respect to statements made or incorporated by reference in the Proxy Statement or the Schedule 14D-9 based on information supplied by or on behalf of Parent or Merger Sub for inclusion or

A-27

Table of Contents

incorporation by reference therein or based on information which is not made in or incorporated by reference in such documents but which should have been disclosed pursuant to Section 4.12.

- (c) The information to be supplied by the Company for inclusion (or incorporation by reference, as the case may be) in the European Prospectus shall not, at the time the European Prospectus is made publicly available, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading.
- (d) The information to be supplied by the Company for inclusion (or incorporation by reference, as the case may be) in the Parent Circular shall not, at the time the Parent Circular is made publicly available and at the time of the Parent Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading.
- (e) In the event that a vote of the Company s stockholders is required to consummate the Merger, the proxy or information statement to be sent to the stockholders of the Company in connection with the adoption of this Agreement and, if applicable, the special meeting of stockholders of the Company (the Company Special Meeting) to consider and vote on a proposal to adopt this Agreement (such proxy or information statement, as the same may be amended or supplemented, the Proxy Statement) and the other documents required to be filed by the Company with the SEC in connection with the Merger and the other transactions contemplated hereby will comply as to form in all material respects with the requirements of the Exchange Act and the Securities Act, as the case may be. The Proxy Statement and the other documents required to be filed by the Company with the SEC in connection with the Merger and the other transactions contemplated hereby shall not on the date the Proxy Statement is first mailed to the stockholders of the Company, at the time of the Company Special Meeting (if applicable) or at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading, or omit to state any material fact necessary to correct any statement in any earlier written communication constituting a solicitation of proxies by the Company for the Company Special Meeting which has in the interim become false or misleading in any material respect.

3.17 Properties and Assets.

- (a) Other than properties and assets disposed of by the Company and its Subsidiaries in the ordinary course of business since the Most Recent Balance Sheet Date, the Company and its Subsidiaries have good and valid title to all of their respective material properties, interests in properties and assets, real and personal, reflected on the Most Recent Balance Sheet or acquired since the Most Recent Balance Sheet Date, or, in the case of material leased properties and assets, valid leasehold interests in such properties and assets, in each case free and clean of all Liens, except in each case in this Section 3.17(a) as would not reasonably be expected to have a Company Material Adverse Effect.
- (b) Section 3.17(b) of the Company Disclosure Schedule sets forth a complete and correct list of each parcel of real property owned or leased by the Company or any of its Subsidiaries as of the date of this Agreement and material to the conduct of the business of the Company and its Subsidiaries, taken as a whole (the material leases pursuant to which the Company or any of its Subsidiaries is a tenant of any such real property being hereinafter referred to as the <u>Leases</u>). As of the date of this Agreement (i) the Leases are in full force and effect in accordance with their terms; (ii) the Company is not in default of any of its obligations under the Leases and (iii) to the knowledge of the Company, the landlords under the Leases are not in default of the landlords obligations under the Leases.
- (c) The facilities, property and equipment owned, leased or otherwise used by the Company or any of its Subsidiaries are in a good state of maintenance and repair, free from material defects and in good operating condition (subject to normal wear and tear), and suitable for the purposes for which they are presently used.

A-28

Table of Contents

3.18 Insurance.

- (a) Section 3.18(a) of the Company Disclosure Schedule sets forth a list, as of the date of this Agreement, of each insurance policy that is material to the Company and its Subsidiaries, taken as a whole (the <u>Insurance Policies</u>), and all material claims made under such Insurance Policies since July 1, 2005. All premiums due and payable under the Insurance Policies have been paid on a timely basis and the Company and its Subsidiaries are in compliance in all material respects with all other material terms thereof. Complete and correct copies of the Insurance Policies have been made available to Parent.
- (b) The Insurance Policies are in full force and effect and there are no material claims pending as of the date of this Agreement as to which coverage has been denied by the Company s respective insurer. Since July 1, 2005, all material claims thereunder have been filed in a due and timely fashion.

3.19 Taxes.

- (a) For purposes of this Agreement, a <u>Tax</u> means any and all federal, state, local and foreign taxes, and any assessments and other governmental charges, duties, impositions and liabilities in the nature of a tax, including taxes based upon or measured by gross receipts, income, profits, sales, use and occupation, value added, ad valorem, transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes, together with all interest, penalties and additions imposed with respect to such amounts and any obligations under any agreements or arrangements with any other Person with respect to such amounts and including any liability for Taxes of a predecessor entity.
- (b) Except as set forth on Section 3.19(b) of the Company Disclosure Schedule, each of the Company and its Subsidiaries has timely filed all material federal, state, local and foreign returns, estimates, information statements and reports required to be filed by it (collectively, <u>Returns</u>) relating to any and all Taxes concerning or attributable to the Company or any of its Subsidiaries or to their operations, and all such Returns are complete and correct in all material respects.
- (c) Each of the Company and its Subsidiaries (i) has paid all Taxes it is obligated to pay as reflected on the Returns or otherwise to the extent such payment was legally due and (ii) has withheld all federal, state, local and foreign Taxes required to be withheld with respect to its employees or otherwise, except for any failure to withhold that would not reasonably be expected to have a Company Material Adverse Effect.
- (d) There is no material Tax deficiency proposed in writing or assessed against the Company or any of its Subsidiaries that is not accurately reflected as a liability on the Most Recent Balance Sheet, nor has the Company or any of its Subsidiaries executed any waiver of any statute of limitations on or extending the period for the assessment or collection of any material Tax which waiver or extension is currently in effect.
- (e) Neither the Company nor any of its Subsidiaries has any material liability for unpaid Taxes that has not been properly accrued for under GAAP and reserved for on the Most Recent Balance Sheet, whether asserted or unasserted, contingent or otherwise or which accrued after the Most Recent Balance Sheet Date in the ordinary course of business.
- 3.20 Environmental Matters. Except as set forth on Section 3.20 of the Company Disclosure Schedule:
- (a) The Company is in compliance in all material respects with all Environmental Laws. The Company possesses Material Permits required under all Environmental Laws and is in material compliance with the terms and conditions thereof.
- (b) The Company has not received any written communication, whether from a Governmental Authority or other Person, that alleges that either the Company or any of its Subsidiaries is not in compliance with any Environmental Laws or any Material Permit required under any applicable Environmental Law, or that it is liable under any Environmental Law, or that it is responsible (or potentially responsible) for the remediation of any Materials of Environmental Concern at, on or beneath its facilities or at, on or beneath any land adjacent thereto or any other property.

A-29

Table of Contents

- (c) There are no past or present facts, circumstances, activities, practices or conditions existing at the facilities currently or formerly owned or operated by the Company, including the release of any Materials of Environmental Concern, that would reasonably be expected to give rise to any liability or result in a claim against the Company or any of its Subsidiaries under any Environmental Law or that would interfere with or prevent compliance with any Material Permits.
- (d) Neither the Company nor any of its Subsidiaries has used, generated, manufactured, discharged, assembled, processed, stored, released, disposed of, arranged for the disposal of or transported any Materials of Environmental Concern except in material compliance with all Material Permits and Environmental Laws.
- (e) Neither the Company nor any of its Subsidiaries is a party to any federal, state, local or private litigation, proceedings, administrative action, or, to the knowledge of the Company, is the subject of any investigation involving a demand for damages or other potential liability under any Environmental Laws, and neither the Company nor any of its Subsidiaries has received or, to the knowledge of the Company, is subject to any order or decree of any Governmental Authority relating to a violation of or liability under Environmental Laws.
- (f) To the knowledge of the Company, no underground storage tanks or surface impoundments exist on any property currently or formerly owned or leased by the Company or its Subsidiaries.
- (g) For purposes of this Agreement, the terms <u>release</u> and <u>environment</u> shall have the meaning set forth in the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, <u>Environmental Law</u> shall mean any Law existing and in effect on the date hereof relating to pollution or protection of the environment, health or safety including without limitation any statute or regulation pertaining to the: (i) manufacture, processing, use, distribution, management, possession, treatment, storage, disposal, generation, transportation or remediation of Materials of Environmental Concern; (ii) air, water and noise pollution; (iii) the protection and use of surface water, groundwater and soil; (iv) the release or threatened release into the environment of hazardous substances, or solid or hazardous waste, including emissions, discharges, releases, injections, spills, escapes or dumping of Materials of Environmental Concern; (v) the conservation, management, or use of natural resources and wildlife, including all endangered and threatened species; (vi) aboveground or underground storage tanks, vessels and containers and (vii) abandoned, disposed of or discarded barrels, tanks, vessels, containers and other closed receptacles. <u>Materials of Environmental Concern</u> shall mean any chemical, substance, material, by-product, waste or product defined or regulated under any Environmental Law, and includes without limitation petroleum or petroleum byproducts, medical or infectious waste, radioactive material, asbestos, asbestos-containing material, polychlorinated biphenyls, and hazardous waste.

3.21 Intellectual Property.

- (a) Each of the Company and its Subsidiaries owns, licenses or otherwise possesses the rights to use and license, subject to any existing licenses or other grants of rights to third parties pursuant to agreements previously made available to Parent, all patents (including any registrations, continuations, continuations in part, divisionals, renewals, reexaminations, reissues and applications therefor), copyrights, trademarks, service marks, trade names, Uniform Resource Locators and Internet URLs, designs, slogans, computer programs and other computer software, databases, technology, trade secrets and other confidential information, know-how, processes, formulae, algorithms, models, user interfaces, customer lists, inventions, source codes and object codes, methodologies, architecture, structure, display screens, layouts, development tools, instructions, templates, trade dress, logos and all documentation and media constituting, describing or relating to each of the foregoing, together with all goodwill related to any of the foregoing, in each case as is necessary to conduct their respective businesses as presently conducted (collectively, the Company Intellectual Property Rights).
- (b) Section 3.21(b) of the Company Disclosure Schedule sets forth, with respect to all Company Intellectual Property Rights registered with any Governmental Authority or for which an application has

A-30

Table of Contents

been filed with any Governmental Authority, as of the date of this Agreement, (i) the registration or application number, the date filed and the title, if applicable, of the registration or application and (ii) the names of the jurisdictions covered by the applicable registration or application. The Company has disclosed to Parent each Company Material Contract in effect as of the date of this Agreement containing any ongoing royalty or payment obligations in excess of \$50,000 per annum with respect to Company Intellectual Property Rights that are licensed or otherwise made available to the Company and its Subsidiaries.

- (c)(i) to the knowledge of the Company, all Company Intellectual Property Rights that have been registered with any Governmental Authority are valid and subsisting and (ii) as of the Closing Date, in connection with such registered Company Intellectual Property Rights, all necessary registration, maintenance and renewal fees will have been paid and all necessary documents and certificates will have been filed with the relevant Governmental Authorities.
- (d) Neither the Company nor any of its Subsidiaries is, or will as a result of the consummation of the Merger or other transactions contemplated by this Agreement be, in breach in any material respect of any license, sublicense or other agreement relating to the Company Intellectual Property Rights, or any licenses, sublicenses and other agreements to which the Company or any of its Subsidiaries is a party and pursuant to which the Company or any of its Subsidiaries uses any patents, copyrights (including software), trademarks or other intellectual property rights of or owned by third parties material to the conduct of the business of the Company and its Subsidiaries (the <a href="https://example.com/en-charge-cha
- (e) Neither the Company nor any of its Subsidiaries has been named as a defendant in any suit, action or proceeding which involves a claim of infringement or misappropriation of any Third Party Intellectual Property Right. Except as set forth in Section 3.21(e)(i) of the Company Disclosure Schedule, neither the Company nor any of its Subsidiaries has as of the date of this Agreement received any written notice of any actual or alleged infringement, misappropriation or unlawful or unauthorized use of any Third Party Intellectual Property that has not been addressed or resolved. With respect to its marketed products, the Company does not infringe any Third Party Intellectual Property Rights. With respect to the Company s product candidate identified in Section 3.21(e)(ii) of the Company Disclosure Schedule, to the knowledge of the Company, after the same are marketed, such marketing would not infringe any third party intellectual property rights other than Third Party Intellectual Property Rights.
- (f) Except as set forth on Section 3.21(f) of the Company Disclosure Schedule, as of the date hereof, to the knowledge of the Company, no other Person is infringing, misappropriating or making any unlawful or unauthorized use of any Company Intellectual Property Rights. As of the Closing Date, to the knowledge of the Company, no Person will be infringing, misappropriating or making any unlawful or unauthorized use of any Company Intellectual Property Rights.
- 3.22 *Brokers*. No broker, financial advisor, investment banker or other financial intermediary is entitled to any fee, commission or expense reimbursement in connection with the Merger or other transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Company, other than J.P. Morgan Securities Inc. and any other investment bank who renders a fairness opinion to the Company Board in connection with the Merger.
- 3.23 Certain Business Practices. To the knowledge of the Company, neither the Company, its Subsidiaries or, to the knowledge of the Company, any director, officer, employee, consultant, service provider, or agent of the Company has, in the course of his or her duties on behalf of the Company: (a) used any funds for unlawful contributions, gifts, entertainment or other unlawful payments relating to political activity; (b) made any unlawful payment to any foreign or domestic government official or employee or to any foreign or domestic political party or campaign or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended; (c) consummated any transaction, made any payment, entered into any agreement or arrangement or taken any other action in violation of Section 1128B(b) of the Social Security Act, as amended or (d) made any other unlawful payment.

A-31

Table of Contents

- 3.24 *Government Contracts*. The consummation of the Merger and other transactions contemplated by this Agreement will not result in any Debarment of the Company or the Surviving Company, any of its Subsidiaries or, to the knowledge of the Company, Parent.
- 3.25 *Interested Party Transactions*. Between July 1, 2006 and the date of this Agreement, no event has occurred that would be required to be reported by the Company as a Certain Relationship or Related Transaction pursuant to Item 404 of Regulation S-K.
- 3.26 *Opinion of Financial Advisor*. The Company Board has received the opinion of its financial advisor, J.P. Morgan Securities Inc., dated as of the date of this Agreement, to the effect that, in its opinion, as of such date the consideration to be received by the stockholders of the Company in the Offer and the Merger, together and not separately, is fair, from a financial point of view, to such stockholders. The Company has provided, or will provide, a complete and correct copy of such opinion to Parent solely for informational purposes.

4. REPRESENTATIONS AND WARRANTIES OF PARENT, NAH AND MERGER SUB

Except as set forth in the disclosure schedule provided by Parent to the Company on the date hereof (the <u>Parent Disclosure Schedule</u>), each of Parent, NAH and Merger Sub represents and warrants to the Company that the statements contained in this Section 4 are true and correct. The Parent Disclosure Schedule shall be arranged in paragraphs corresponding to the numbered and lettered paragraphs contained in this Section 4, and the disclosure in any paragraph shall qualify (A) the corresponding paragraph of this Section 4 and (B) the other paragraphs of this Section 4 to the extent that it is readily apparent from a reading of the Parent Disclosure Schedule, without reference to anything other than the Parent Disclosure Schedule, it also qualifies or applies to such other paragraphs. As used in this Agreement, a <u>Parent Material Adverse Effect</u> means any change, event or effect that is materially adverse to the business, assets (including intangible assets), financial condition, or results of operations of Parent and its Subsidiaries, taken as a whole, or on Parent s ability to consummate the transactions contemplated in this Agreement, provided that none of the following shall constitute a Parent Material Adverse Effect:

- (a) general economic conditions worldwide, in the United States, or in any nation or region in which Parent or any of its Subsidiaries has a substantial presence or operations, provided that such conditions do not disproportionately affect Parent and its Subsidiaries relative to other industry participants;
- (b) any acts of terrorism not directed at Parent or any of its Subsidiaries or any outbreak of war, provided that such conditions do not disproportionately affect Parent and its Subsidiaries relative to other industry participants;
- (c) the public announcement by the Parties of this Agreement, the pendency of the Merger and the other transactions contemplated hereby, or any action taken which is required by this Agreement or specifically requested by the Company;
- (d) factors generally affecting the industries or markets in which Parent and its Subsidiaries operate, provided that such factors do not disproportionately affect Parent and its Subsidiaries relative to other industry participants;
- (e) changes in Law or GAAP or the interpretation thereof, provided that such changes do not disproportionately affect Parent and its Subsidiaries relative to other industry participants; or
- (f) a decline in the trading price or change in trading volume of the Parent Common Shares, provided that this clause will not exclude any underlying change, event, circumstance, development or effect that may have resulted in, or contributed to, a decline in trading price or change in trading volume.

A-32

Table of Contents

- 4.1 Organization and Qualification.
- (a) Parent is a Naamloze Vennootschap organized in the Kingdom of The Netherlands. Parent is duly qualified or licensed as a foreign corporation to conduct business, and is in corporate good standing, under the Laws of each jurisdiction that recognizes the concept of good standing where the character of the properties owned, leased or operated by it, or the nature of its activities, makes such qualification or licensing necessary, except where the failure to be so qualified, licensed or in good standing, individually or in the aggregate, would not reasonably be expected to have a Parent Material Adverse Effect.
- (b) Parent has heretofore furnished or otherwise made available to the Company a true, complete and correct copy of its articles of association (the <u>Articles of Association</u>), the charters of the Supervisory and Managing Boards of Parent and any committees thereof, the Articles of Incorporation and Bylaws of NAH and the Certificate of Formation and Operating Agreement, in each case as currently in effect. The Articles of Association, the charters of the Supervisory and Managing Boards of Parent and any committees thereof, the Articles of Incorporation and Bylaws of NAH and the Certificate of Formation and Operating Agreement, as so made available, are in full force and effect and no other organizational documents are applicable to or binding upon Parent, NAH or Merger Sub. Parent is not in violation of any provisions of its Articles of Association in any material respect. NAH is not in violation of any provisions of its Articles of Incorporation or Bylaws in any material respect. Merger Sub is not in violation of any provisions of the Certificate of Formation or Operating Agreement in any material respect.
- 4.2 Subsidiaries.
- (a) Except as set forth on Section 4.2(a) of the Parent Disclosure Schedule, Exhibit 8.1 to Parent s Annual Report on Form 20-F for the fiscal year ended December 31, 2006 sets forth a complete and correct list of each Subsidiary of Parent other than Merger Sub as of the date of this Agreement.
- (b) Each material Subsidiary of Parent is a corporation duly organized, validly existing and in good standing (to the extent such concepts are applicable) under the Laws of the jurisdiction of its organization (which in the case of Merger Sub is the State of Delaware), and is duly qualified or licensed as a foreign corporation to conduct business, and is in good standing (to the extent such concepts are applicable), under the Laws of each jurisdiction where the character of the properties and other assets owned, leased or operated by it, or the nature of its activities, makes such qualification or licensing necessary, except where the failure to be so qualified, licensed or in good standing, individually or in the aggregate, would not reasonably be expected to have a Parent Material Adverse Effect.
- (c) All of the issued and outstanding shares of capital stock of, or other equity interests in, each material Subsidiary of Parent are: (i) duly authorized, validly issued, fully paid, non-assessable (to the extent such concepts are applicable); (ii) owned, directly or indirectly, by Parent (other than directors—qualifying shares in the case of foreign Subsidiaries) free and clear of all Liens and (iii) free of any restriction which prevents the payment of dividends to Parent or any other Subsidiary of Parent, or which otherwise restricts the right to vote, sell or otherwise dispose of such capital stock or other ownership interest other than restrictions under the Securities Act and state securities Law.
- (d) None of Parent s Subsidiaries is required to file any forms, reports or other documents with the SEC.
- 4.3 Capital Structure.
- (a) The authorized capital stock of Parent as of the date of this Agreement consists of (i) 260,000,000 Parent Common Shares, (ii) 40,000,000 Financing Preference Shares (Parent Financing Preference Shares) and 300,000,000 Preference Shares (Parent Preference Shares).
- (b) As of the close of business on May 30, 2007: (i) 150,624,527 Parent Common Shares were issued and outstanding; (ii) no shares of Parent Preference Stock were issued or outstanding; (iii) Parent Common Shares were held in the treasury of Parent; (iv) 12,081,368 Parent Common Shares (the Parent Option/Award Shares) were duly reserved for future issuance upon the exercise of stock options and stock awards

A-33

Table of Contents

granted on or prior to the date hereof pursuant to Parent s option and incentive plans and 5,012,063 Parent Common Shares were duly reserved for future issuance upon the exercise of employee stock options and stock awards available for grant after the date hereof pursuant to Parent s option and incentive plans; (v) 26,862,490 shares (the <u>Parent Convertible Note Shares</u>) of Parent Common Shares were duly reserved for issuance in connection with the convertible notes due 2024 and 2026 and (vi) Parent Preference Shares equal to the number of Parent Common Shares outstanding from time to time less one share issuable to SPAQ pursuant to the Option Agreement dated August 2, 2004.

- (c) All outstanding Parent Common Shares and applicable Parent Option/Award Shares are, and any Parent Option/Award Shares, Parent Warrant Shares, Parent Convertible Note Shares, and Parent Preference Shares will be, if and when issued in accordance with the terms of the underlying securities described in Section 4.3(b), and all Parent Common Shares to be issued in connection with the Merger will be, when issued in accordance with the terms hereof, duly authorized, validly issued, fully paid and non-assessable, and not subject to, or issued in violation of, any preemptive, subscription or any kind of similar rights.
- (d) As of the date of this Agreement, except as described in subsection (b) above or as described in Section 4.3(d) of the Parent Disclosure Schedule, there are no bonds, debentures, notes or other indebtedness of Parent having the right to vote (or convertible into securities having the right to vote) on any matters on which stockholders of Parent may vote. As of the date of this Agreement, except as described in subsection (b) above or as described in Section 4.3(d) of the Parent Disclosure Schedule, there are no outstanding securities, options, warrants, calls, rights, commitments, agreements, arrangements or undertakings of any kind (contingent or otherwise) to which Parent is a party or bound obligating Parent to issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock or other voting securities of Parent or obligating Parent to issue, grant, extend or enter into any agreement to issue, deliver or sell any such capital stock or securities. As of the date of this Agreement, except as set forth in Section 4.3(d) of the Parent Disclosure Schedule, neither Parent nor any Subsidiary of Parent is subject to any obligation or requirement to provide material funds for or to make any material investment (in the form of a loan or capital contribution) in any Person.
- (e) All grants of Parent Stock Options were made in compliance with the terms of the applicable Parent Stock Option Plan under which such Parent Stock Options were made and with applicable Laws. All Parent Stock Options have been properly accounted for in the Parent statements and reported in compliance with GAAP and applicable Tax requirements and the published rules and regulations of the SEC with respect thereto, and there is no basis for any claim that the grant date of any Parent Stock Option is inaccurate or the exercise price thereof is improper.
- (f) All of the issued and outstanding Parent Common Shares were issued in compliance in all material respects with all applicable federal and state securities Laws.
- (g) Except as set forth on Section 4.3(g) of the Parent Disclosure Schedule there are no voting trusts, proxies or other agreements, commitments or understandings to which the Parent or any of its Subsidiaries or, to the knowledge of the Parent, any of the shareholders of the Parent is a party or by which any of them is bound with respect to the issuance, holding, acquisition, voting or disposition of any shares of capital stock or other security or equity interest of the Parent or any of its Subsidiaries.
- (h) The execution, delivery and performance of this Agreement by Parent, NAH and Merger Sub, do not and will not trigger the rights of *Stichting Preferente Aandelen QIAGEN* (SPAQ) to purchase Parent Preference Shares pursuant to the Option Agreement between SPAQ and Parent dated August 2, 2004.
- 4.4 Authority; No Conflict; Required Filings.
- (a) Each of Parent, NAH and Merger Sub has all necessary power and authority to execute and deliver this Agreement and to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution, delivery and performance of this Agreement by each of Parent, NAH and Merger Sub and the consummation by each of Parent, NAH and Merger Sub of the Merger and the other

A-34

transactions contemplated hereby have been duly and validly authorized by all necessary action by each of the Supervisory and Managing Boards of Parent (including approval of the issuance of the Parent Common Shares up to a maximum number sufficient to effect the Offer and the Merger and to exclude preemptive right in respect of such issuance and any other related matters pursuant to this Agreement), the Board of Directors of NAH and the sole member of Merger Sub, and no other proceedings by or on the part of Parent, NAH or Merger Sub are necessary to authorize this Agreement, to perform their respective obligations hereunder, or to consummate the transactions contemplated hereby (other than (i) adoption by Parent s stockholders of a valid resolution or resolutions in relation to the transactions contemplated by this Agreement, including valid adoption or approval, as the case may be, of the decisions to (A) enter into this Agreement pursuant to Section 2:107a Dutch Civil Code, (B) delegate the authority to issue Parent Common Shares to the Supervisory Board of Parent up to a maximum number sufficient to effect the Merger and to exclude pre-emptive rights in respect of such issuance and any related matters pursuant to this Agreement, with each of the decisions under clauses (A) and (B) being conditional upon the valid adoption of the other decisions set out therein, by the vote of a majority in voting power present in person or by proxy at the Parent Stockholders Meeting and entitled to vote on the matter (such vote, the Parent Requisite Stockholder Approval), (ii) the adoption of this Agreement by NAH as the sole member of Merger Sub (which adoption shall occur immediately after the execution and delivery of this Agreement) and (iii) the filing with the Secretary of State of the State of Delaware of the Certificate of Merger as required by the DGCL. This Agreement has been duly and validly executed and delivered by Parent, Merger Sub and NAH and, assuming due authorization, execution and delivery hereof by the Company, constitutes a legal, valid and binding obligation of each of Parent, NAH and Merger Sub enforceable against each of Parent, NAH and Merger Sub in accordance with its terms, subject to the Bankruptcy and Equitable Exceptions.

- (b) The execution, delivery and performance of this Agreement by Parent, NAH and Merger Sub, do not and will not: (i) conflict with or violate the Articles of Association of Parent, the Articles of Incorporation or Bylaws of NAH or the Certificate of Formation or Operating Agreement; (ii) conflict with or violate the memorandum or articles of association, certificate of incorporation, Bylaws or comparable constituent documents of any of the Subsidiaries of Parent; (iii) assuming that all consents, approvals, authorizations, declarations and permits contemplated by clauses (i) through (ix) of subsection (c) below have been obtained, and all filings described in such clauses have been made, conflict with or violate any Law applicable to Parent or any of its Subsidiaries or by which its or any of their respective properties are bound or (iv) other than as set forth in Section 4.4 of the Parent Disclosure Schedule, result in any breach or violation of or constitute a default (or an event which with notice or lapse of time or both would become a default) or result in the loss of a benefit under, require any consent or approval under, or give rise to any right of termination, cancellation, amendment, acceleration or other alteration in the rights under, any agreement to which Parent or any of its Subsidiaries or its or any of their respective properties are bound.
- (c) The execution, delivery and performance of this Agreement by each of Parent, NAH and Merger Sub and the consummation of the transactions contemplated hereby by each of Parent, NAH and Merger Sub do not and will not require any consent, approval, authorization, declaration or permit of, action by, filing with or notification to, any Governmental Authority, except for: (i) the applicable requirements of the Securities Act, the Exchange Act and the rules and regulations promulgated thereunder; (ii) the applicable requirements of the Financial Markets Supervision Act (*Wet op het financieel toezicht*)(the <u>FMS</u>A), the Trade Register Act (*Handelsregisterwet 1996*), and the rules and regulations promulgated thereunder; (iii) the applicable requirements of the HSR Act and the rules and regulations promulgated thereunder; (iv) the applicable requirements of state securities and blue sky Laws; (v) the applicable requirements of the Frankfurt Stock Exchange, the German Federal Financial Supervisory Authority (*Bundesanstalt für Finanzdienstleistungsaufsicht*) (the <u>BaFin</u>), the Authority for the Financial Market (*Autoriteit Financiële Markten*) (the <u>AFM</u>) and the NGSM; (vi) the filing with the Secretary of State of the State of Delaware of the Certificate of Merger as required by the DGCL; (vii) the filing with the European Commission of a merger notification in accordance with the ECMR (viii) the applicable requirements of the competent authority of any member state of the European Union to which any of the transactions contemplated by this

A-35

Table of Contents

Agreement is referred pursuant to Article 9 of the ECMR and (ix) the applicable requirements of the German Securities Trading Act (*Wertpapierhandelsgesetz*), the Securities Prospectus Act (*Wertpapierprospektgesetz*), and the Stock Exchange Act (*Börsengesetz*), and the rules and regulations promulgated thereunder.

- 4.5 SEC Filings; Financial Statements; Sarbanes-Oxley Act.
- (a) Since January 1, 2004, Parent has timely filed all forms, reports and documents required to be filed by Parent with the SEC, including all exhibits required to be filed therewith (including any forms, reports and documents filed after the date hereof, as filed, the Parent SEC Reports (i) at the time filed complied (or will comply when filed, as the case may be) as to form in all material respects with the applicable requirements of the Securities Act and/or the Exchange Act, as the case may be and (ii) did not at the time they were filed (or, if later filed, amended or superseded, then on the date of such later filing) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements contained therein, in the light of the circumstances under which they were made, not misleading.
- (b) Each of the consolidated financial statements (including, in each case, any related notes thereto) contained in the Parent SEC Reports (collectively, the Parent Financial Statements), at the time filed and amended, (i) complied or will comply, as the case may be, as to form in all material respects with the applicable published rules and regulations of the SEC with respect thereto, (ii) was or will be prepared in accordance with GAAP applied on a consistent basis throughout the periods involved except as may otherwise be indicated in the notes thereto and (iii) fairly presented or will fairly present, as the case may be, in all material respects, the consolidated financial position of Parent and its Subsidiaries as at the dates indicated and the consolidated results of operations and cash flows for the periods therein indicated, except, in the case of the unaudited interim financial statements for the absence of footnotes and normal year-end adjustments.
- (c) The management of Parent has established and maintains disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) that are designed to provide reasonable assurance that information (both financial and non-financial) required to be disclosed by Parent in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC and that all such information is accumulated and communicated to Parent s management as appropriate to allow timely decisions regarding required disclosure and to make the certifications of the Chief Executive Officer and Chief Financial Officer of Parent required under the Exchange Act with respect to such reports. As of December 31, 2006, there were no material weaknesses in Parent s or any of Parent Subsidiaries internal controls as contemplated under SOX. Parent has disclosed, based on the most recent evaluation of its internal control over financial reporting (as defined in Rule 13a-15(f) of the Exchange Act) by its chief executive officer and chief financial officer, to Parent s auditors and the audit committee of Parent s Supervisory Board (a) any significant deficiencies in the design or operation of its internal control over financial reporting that are reasonably likely to adversely affect Parent s ability to record, process, summarize and report financial information and has identified for Parent s auditors and audit committee of Parent s Supervisory Board any material weaknesses in its internal control over financial reporting and (b) any fraud, whether or not material, that involves management or other employees who have a significant role in Parent s internal control over financial reporting. Since the date of Parent s most recent evaluation of internal control over financial reporting, to the knowledge of Parent, no facts or circumstances have arisen or occurred that would be required to be disclosed to Parent s auditors or Parent s audit committee regarding (x) a significant deficiency in the design or operation of its internal control over financial reporting, (y) a material weakness in its internal control over financial reporting or (z) fraud, whether or not material, that involves management or other employees who have a significant role in Parent s internal control over financial reporting.
- (d) Neither Parent nor any of its Subsidiaries has outstanding, or has arranged any outstanding, extensions of credit to directors or executive officers within the meaning of Section 402 of SOX. To

A-36

Table of Contents

Parent s knowledge, there are no other violations of Parent s code of conduct, adopted pursuant to NASDAQ Rule 4350(n). Parent has delivered to Parent complete and correct copies of any written complaints, reports or allegations, and a reasonably detailed summary of any verbal complaints, reports or allegations, that have been submitted or made by any party to the Audit Committee of Parent s Supervisory Board since July 1, 2005 pursuant to the procedures established in accordance with Section 10A(m)(4) of the Exchange Act. Parent has delivered to the Company complete and correct copies of any written complaints, reports or allegations, and a reasonably detailed summary of any verbal complaints, reports or allegations, made or submitted by any attorney representing Parent or any of its Subsidiaries to Parent s chief executive officer, any supervising attorney or Parent s Supervisory Board or any committee thereof, including any qualified legal compliance committee, as contemplated by the rules set forth in 17 CFR Part 205. Neither Parent nor any of its Subsidiaries is a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar arrangement, including without limitation any off-balance sheet arrangement (as defined in Item 303(a) of Regulation S-K promulgated by the SEC), where the result, purpose or intended effect of such contract or arrangement is to avoid disclosure of any material transaction involving, or material liabilities of, Parent or any of its Subsidiaries in any Parent Financial Statement or Parent SEC Report.

- 4.6 Absence of Undisclosed Liabilities. Parent and its Subsidiaries do not have any material liabilities or obligations, whether fixed, contingent, accrued or otherwise, liquidated or unliquidated and whether due or to become due, in each case of a nature required by GAAP to be reflected on a consolidated balance sheet of Parent, other than: (i) liabilities reflected or reserved against on the balance sheet contained in Parent s Form 20-F (the Parent s Most Recent Balance Sheet) filed with the SEC on April 2, 2007; (ii) liabilities or obligations incurred since December 31, 2006 (the Parent s Most Recent Balance Sheet Date) in the ordinary course of business consistent with past practice, none of which has had or reasonably would be expected to have, individually or in the aggregate, a Material Adverse Effect and (iii) liabilities described in Section 4.6 of the Parent Disclosure Schedule.
- 4.7 Absence of Certain Changes or Events. Since the Parent s Most Recent Balance Sheet Date, Parent and its Subsidiaries have conducted their respective businesses only in the ordinary course of business consistent in all material respects with past practice, and there has not been any action, event or occurrence which or would reasonably be expected to have, a Parent Material Adverse Effect.
- 4.8 Compliance with Law; Regulatory Matters.
- (a) The Parent and its Subsidiaries are conducting and have conducted their business and operations in material compliance with applicable provisions of the FDA Law and Regulation and comparable foreign regulatory or Governmental Authorities.
- (b) Except as set forth in Section 4.8(b) of the Parent Disclosure Schedule, neither the Parent nor its Subsidiaries has received any notice or communication from the FDA alleging noncompliance with any applicable FDA Law and Regulation. The Parent and its Subsidiaries are not subject to any enforcement, regulatory or administrative proceedings by the FDA or any comparable foreign regulatory or Governmental Authority and, to the knowledge of the Parent, no such proceedings have been threatened. There is no civil, criminal or administrative action, suit, demand, claim, complaint, hearing, investigation, demand letter, warning letter, proceeding or request for information pending against Parent or the Parent Subsidiaries, and, to the Parent sknowledge, the Parent and its Subsidiaries have no liability (whether actual or contingent) for failure to comply with any FDA Law and Regulation or any law or regulation of a comparable foreign regulatory or Governmental Authority. There is no act, omission, event, or circumstance of which the Parent or its Subsidiaries have knowledge that would reasonably be expected to give rise to or lead to any such action, suit, demand, claim, complaint, hearing, investigation, notice, demand letter, warning letter, proceeding or request for information or any such liability. There has not been any intentional violation of any FDA Law and Regulation or any law or regulation of a comparable foreign regulatory or Governmental Authority by the Parent or its Subsidiaries in their product development efforts, submissions, record keeping

A-37

Table of Contents

and reports to FDA or a comparable Governmental Authority that could reasonably be expected to require or lead to investigation, corrective action or enforcement, regulatory or administrative action that would result in a Material Adverse Effect. To the knowledge of Parent, there are no civil or criminal proceedings relating to the Parent or its Subsidiaries or any Parent or Subsidiary employee which involve a matter within or related to the FDA s jurisdiction or the jurisdiction of any comparable Governmental Authority.

- (c) To the extent applicable, each of the products and product candidates of the Parent and its Subsidiaries is being, and at all times has been, developed, tested, manufactured, labeled, promoted and stored, as applicable, in material compliance with FDA Law and Regulation and requirements of comparable foreign regulatory and Governmental Authorities, including those requirements relating to good manufacturing practice and good laboratory practice.
- (d) The animal studies and other preclinical tests conducted by the Parent or its Subsidiaries were, and if still pending, are, being conducted in all material respects in accordance with applicable FDA and foreign requirements including, but not limited to, good laboratory practices, and FDA Law and Regulation. Neither the Parent nor its Subsidiaries has received any written notice from the FDA or any other regulatory or Governmental Authority requiring the termination or suspension or material modification of any animal study or preclinical study conducted by or on behalf of the Parent or any Subsidiary.
- (e) Neither the Parent nor any of its Subsidiaries is subject to any pending or, to the knowledge of the Parent, threatened investigation by (A) the FDA, (B) Department of Health and Human Services Office of Inspector General or Department of Justice pursuant to the Federal Healthcare Program Anti-Kickback Statute (42 U.S.C. Section 1320a-7b(b) (known as the <u>Anti-Kickback Statute</u>) or the Federal False Claims Act (31 U.S.C. Section 3729), or (C) any equivalent state or foreign Law.
- (f) To the knowledge of the Parent, no officer, employee or agent of the Parent or the Parent Subsidiaries, has been convicted of any crime or engaged in any conduct for which such person or entity could be excluded from participating in the federal health care programs under Section 1128 of the Social Security Act or any similar law or regulation.
- (g) The Parent is not Debarred from any government program. To the knowledge of the Parent, no officer, employee or agent of the Parent has been convicted of any crime or engaged in any conduct for which debarment is mandated or permitted by 21 U.S.C. § 335a. To the knowledge of the Parent, no officer, employee or agent of the Parent has been convicted of any crime or engaged in any conduct for which such person or entity could be excluded from participating in the federal health care programs under Section 1128 of the Social Security Act or any similar Law. To the knowledge of the Parent, neither the Parent nor its Subsidiaries employs or uses the services of any individual or entity that is or, during the time when such individual or entity was employed by or providing services to the Parent or any of its Subsidiaries, Debarred.
- (h) With respect to all Third Party Manufacturers, the Parent has inspected all Third Party Manufacturers and to its knowledge and to the extent applicable, each such Third Party Manufacturer:
- (i) has complied and is complying, in each case in all material respects, with all Laws, including the FD&C Act and any similar state or foreign Laws;
- (ii) has all material permits to perform its obligations as a Third Party Manufacturer and all such permits are in full force and effect; and
- (iii) has not been Debarred from participation in any government program and does not employ or use the services of any individual or entity that is or, during the time when such person or entity was providing services as a Third Party Manufacturer to the Parent or any of its Subsidiaries, was Debarred.
- (i) To the knowledge of the Parent, no officer, employee or agent of the Parent or the Parent Subsidiaries has: made any untrue statement of material fact or fraudulent statement to the FDA or any other Governmental Entity; failed to disclose a material fact required to be disclosed to the FDA or any other Governmental Entity; or committed an act, made a statement, or failed to make a statement that would

A-38

Table of Contents

reasonably be expected to provide the basis for the FDA or any other Governmental Entity to invoke its policy respecting Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities, as set forth in 56 Fed.Reg. 46191 (September 10, 1991).

- (j) Neither the Parent nor its Subsidiaries has knowingly or willfully solicited, received, paid or offered to pay any remuneration, directly or indirectly, overtly or covertly, in cash or kind for the purpose of making or receiving any referral which violated any applicable anti-kickback or similar Law, including the Anti-Kickback Statute, or any applicable state or foreign anti-kickback Law.
- (k) Section 4.8(k) of the Parent Disclosure Schedule lists, as of the date of this Agreement, all written claims or statements (including all correspondence, other than immaterial correspondence, with Governmental Authorities, Medicare Administrative Contractors) concerning or relating to any health care program that involves, relates to or alleges (i) any material violation of any applicable rule, regulation, policy or requirement of any such program with respect to any activity, practice or policy of the Parent or its Subsidiaries or (ii) any violation of any applicable rule, regulation, policy or requirement of any such program with respect to any claim for payment or reimbursement made by the Parent or its Subsidiaries or any payment or reimbursement paid to the Company or its Subsidiaries. Except as set forth on Section 4.8(k) of the Parent Disclosure Schedule, there are no such violations nor, to the knowledge of the Parent, are there any grounds to reasonably anticipate the commencement of any investigation or inquiry, or the assertion of any claim or demand by, any government agency, Medicare Administrative Contractors with respect to any of the activities, practices, policies or claims of the Parent or its Subsidiaries, or any payments or reimbursements claimed by the Parent or its Subsidiaries, in each case concerning or relating to any federal or state government funded health care program. Neither the Parent nor its Subsidiaries is, as of the date hereof, subject to any outstanding audit by any Governmental Authority, intermediary or carrier and, to the knowledge of the Parent, there are no grounds to anticipate any such audit, except such audits in the ordinary course of review, in the foreseeable future.
- (1) To the knowledge of the Parent, neither the Parent nor any of its Subsidiaries has submitted any claim for payment in violation of any Laws relating to false claim or fraud, including without limitation the Federal False Claim Act, 31 U.S.C. § 3729, or any applicable state or foreign false claim or fraud Law.
- (m) Neither the Parent nor any of its Subsidiaries has failed to comply with any applicable security and privacy standards regarding protected health information under HIPAA, or any applicable state or foreign privacy Laws.
- (n) Neither the Parent nor its Subsidiaries nor, to the knowledge of the Parent, any of their respective officers, directors or employees, acting in their capacities as such, is or has been involved in any activities which are, or are alleged in writing by any qui tam relator or Governmental Authority to be, prohibited under the federal Medicare and Medicaid statutes, which are specifically defined as 42 U.S.C. §§ 1320a-7, 1320a-7a, 1320a-7b, 1395nn, 18 U.S.C. §1347, § 287, §1001, and § 1035, or the federal TRICARE statute, or the regulations promulgated pursuant to such federal statutes.
- (o) The Parent and each of its Subsidiaries has conducted its export transactions in accordance in all material respects with applicable provisions of United States export control Laws, including but not limited to the FD&C Act, Export Administration Act and implementing Export Administration Regulations. Without limiting the foregoing: (i) the Parent and each of its Subsidiaries has obtained all material export licenses and other approvals required for its exports of products and technologies from the United States; (ii) the Parent and each of its Subsidiaries is in compliance in all material respects with the terms of all applicable material export licenses or other approvals; (iii) there are no pending or, to the knowledge of the Parent, threatened claims against the Parent or any of its Subsidiaries with respect to such material export licenses or other approvals; (iv) there are no actions, conditions or circumstances pertaining to the Parent s or any of its Subsidiary s material export transactions that would reasonably be expected to give rise to any future claims and (v) no consents or approvals will be required in connection with the Merger.
- (p) Each of the Parent and its Subsidiaries has submitted to the FDA and all comparable foreign regulatory or Governmental Authorities for and received approval of all material FD&C Act Permits

A-39

Table of Contents

necessary to conduct the business of the Parent and its Subsidiaries as currently conducted. The Parent and its Subsidiaries are in material compliance with all such FD&C Act Permits. To the knowledge of the Parent, no suspension, revocation, cancellation or withdrawal of the FD&C Act Permits is threatened and there is no basis for believing that such FD&C Act Permits will not be renewable upon expiration or will be suspended, revoked, cancelled or withdrawn.

- 4.9 *Material Permits*. Each of Parent and its Subsidiaries holds all federal, state, local and foreign governmental licenses, permits, franchises and authorizations necessary for conduct of its business as presently conducted and the ownership and operation of its properties and other assets, including those that are required under all Environmental Laws, in each case (whether under Environmental Laws or otherwise) (such licenses, permits, franchises and authorizations, the Parent Material Permits) and the FDA Laws and Regulations. Each of Parent and its Subsidiaries has submitted to the FDA and all similar applicable state and local regulatory bodies for and received approval of all registrations, applications, licenses, requests for exemptions, permits and other regulatory authorizations (the FD&C Act Permits) necessary to conduct the business of Parent and its Subsidiaries as currently conducted. Parent and its Subsidiaries are in compliance with all such Parent Material Permits and FD&C Act Permits. To the knowledge of the Parent, no suspension, revocation, cancellation or withdrawal of the Material Permits or FD&C Act Permits is threatened and there is no basis for believing that such Material Permits or FD&C Act Permits will not be renewable upon expiration or will be suspended, revoked, cancelled or withdrawn.
- 4.10 *Litigation and Product Liability*. Except as set forth in Section 4.10 of the Parent Disclosure Schedule, there is no suit, action, arbitration, claim, governmental or other proceeding before any Governmental Authority pending or, to the knowledge of Parent, threatened against Parent or any of its Subsidiaries. No material product liability claims have been asserted in writing or, to the knowledge of Parent, threatened in writing against Parent in respect of any product or product candidate tested, researched, developed, manufactured, marketed, distributed, or sold by, or on behalf of, or in cooperation with Parent.
- 4.11 Labor and Employment Matters.
- (a) To the knowledge of Parent, there are no material labor grievances pending or, to the knowledge of Parent, threatened in writing between Parent or its Subsidiaries, on the one hand, and any of their respective employees or former employees, on the other hand. Parent has not received written notice of any pending or threatened charge, other than any that are immaterial, of (i) an unfair labor practice as defined in the National Labor Relations Act, as amended; (ii) safety violations under the Occupational Safety and Health Act; (iii) wage or hour violations; (iv) discriminatory acts or practices in connection with employment matters or (v) claims by governmental agencies that Parent has failed to comply with any Law relating to employment or labor matters. To the knowledge of Parent, Parent is not currently and has not been the subject of any threatened or actual whistleblower or similar claims by past or current employees or any other persons.
- (b) To the knowledge of Parent, Parent is currently in substantial compliance with all Laws relating to employment, including those related to wages, hours, collective bargaining and the payment and withholding of taxes and other sums as required by the appropriate Governmental Authority and has withheld and paid to the appropriate Governmental Authority all amounts required to be withheld from Parent employees and is not liable for any arrears of wages, taxes withheld, penalties or other sums for failing to comply with any of the foregoing.
- (c) Since the enactment of the WARN Act, neither Parent nor any of its Subsidiaries has effectuated (i) a plant closing affecting any site of employment or one or more facilities or operating units within any site of employment or facility of Parent or any of its Subsidiaries, or (ii) a mass layoff (as defined in the WARN Act) affecting any site of employment or facility of Parent or any of its Subsidiaries, nor has Parent or any of its Subsidiaries been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar state or local law, in each case that would reasonably be expected to have a Parent Material Adverse Effect.

A-40

Table of Contents

- 4.12 Offer Documents; European Prospectus; Parent Circular; Information to be Supplied; Merger Registration Statement.
- (a) Each of the Schedule TO, the Offer Registration Statement and the other documents required to be filed by the Parent with the SEC in connection with the Offer, the Merger and the other transactions contemplated hereby will comply as to form in all material respects with the requirements of the Exchange Act and the Securities Act, as the case may be. Each of the Schedule TO, the Offer Registration Statement and the other documents required to be filed by the Parent with the SEC in connection with the Transactions and any of the information supplied or to be supplied by the Parent or its Subsidiaries or their representatives for inclusion or incorporation by reference in the Schedule 14D-9 will not, at the time the Offer is commenced or at the Acceptance Date, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- (b) In the event that a vote of the Company s stockholders is required to consummate the Merger, the information to be supplied by Parent for inclusion (or incorporation by reference, as the case may be) in the Proxy Statement (if applicable) shall not on the date the Proxy Statement is first mailed to the stockholders of the Company, at the time of the Company Special Meeting (if applicable) and at the Effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or otherwise necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading, or omit to state any material fact necessary to correct any statement in any earlier written communication constituting a solicitation of proxies by the Company for the Company Special Meeting which has in the interim become false or misleading in any material respect.
- (c) The European Prospectus shall not, at the time the European Prospectus is made publicly available, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading.
- (d) The Parent Circular shall not, at the time the Parent Circular is made publicly available and at the time of the Parent Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements contained therein, in light of the circumstances under which they were made, not misleading.
- (e) The registration statement on Form F-4 to be filed with the SEC by Parent in connection with the issuance of Parent Common Shares in the Merger, as amended or supplemented from time to time (as so amended and supplemented, the Merger Registration Statement) will comply as to form, in all material respects, with the requirements of the Exchange Act and the Securities Act. The Merger Registration Statement will not, on the date of its filing or mailing, at the time of the Company Stockholder Meeting or at the effective Time, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.
- (f) Notwithstanding the foregoing provisions of this Section 4.12, no representation or warranty is made by Parent with respect to statements made or incorporated by reference in the Offer Documents, the European Prospectus, the Parent Circular or the Merger Registration Statement based on information supplied by or on behalf of the Company and its Subsidiaries for inclusion or incorporation by reference therein or based on information which is not made in or incorporated by reference in such documents but which should have been disclosed pursuant to Section 3.16.
- 4.13 *Properties and Assets*. Other than properties and assets disposed of by Parent and its Subsidiaries in the ordinary course of business since the Parent s Most Recent Balance Sheet Date, Parent and its Subsidiaries have good and valid title to or, with respect to leased property, valid leasehold interests in, all of their respective properties, interests in material properties and assets, real and personal, in each case free and clear of Liens.

A-41

Table of Contents

- 4.14 Taxes.
- (a) Each of Parent and its Subsidiaries has timely filed all material federal, state, local and foreign Returns relating to any and all Taxes concerning or attributable to Parent or any of its Subsidiaries or to their operations, and all such Returns are complete and correct in all material respects.
- (b) Each of Parent and its Subsidiaries (i) has paid all Taxes it is obligated to pay as reflected on the Returns or otherwise to the extent such payment was legally due and (ii) has withheld all federal, state, local and foreign Taxes required to be withheld with respect to its employees or otherwise.
- (c) Neither Parent nor any of its Subsidiaries has any material liability for unpaid Taxes that has not been properly accrued for under GAAP and reserved for on the Parent s Most Recent Balance Sheet, whether asserted or unasserted, contingent or otherwise or which accrued after the Parent s Most Recent Balance Sheet Date in the ordinary course of business.
- (d) Taking into account the application of the special rules set forth in Treasury Regulation Section 1.367(a)-3(c)(3)(ii), (i) Parent (or any Subsidiary that is a qualified subsidiary (as defined in Treasury Regulation Section 1.367(a)-3(c)(5)(vii)) or a qualified partnership (as defined in Treasury Regulation Section 1.367(a)-3(c)(5)(viii)) is currently engaged, and will have been engaged for the entire 36-month period immediately preceding the Closing Date, in an active trade or business outside the United States within the meaning of Treasury Regulation Section 1.367(a)-2T(b)(2) and (3) (the Oualified Trade or Business); and (ii) Parent (and, if applicable, the Subsidiary engaged in the Qualified Trade or Business) does not have any plan or intention to substantially dispose of or discontinue the Qualified Trade or Business.
- (e) Parent has not been a controlled foreign corporation as defined in Section 957(a) of the Code (a _CFC) on any day of its current taxable year and has no plan or intention to become a CFC on any day subsequent to the Closing Date.
- (f) Parent has not been a passive foreign investment company, as defined in Section 1297(a) of the Code (a <u>PFI</u>C), for its taxable year in which the Closing Date occurs and has no plan or intention to become a PFIC for any subsequent taxable year.
- (g) To the knowledge of Parent, neither Parent nor any of its Subsidiaries has taken any action or failed to take any action which action or failure would (i) prevent the qualification of the Offer and the Merger as a reorganization within the meaning of Section 368(a) of the Code or (ii) cause the Offer and the Merger to fail to satisfy the requirements of Treasury Regulation Section 1.367(a)-3(c)(i), (ii) or (iv) following the Merger.
- 4.15 Environmental Matters. Except as set forth on Section 4.15 of the Parent Disclosure Schedule:
- (a) Parent is in compliance with all Environmental Laws. Parent possesses Material Permits required under all Environmental Laws and is in material compliance with the terms and conditions thereof.
- (b) Parent has not received any written communication, whether from a Governmental Authority or other Person, that alleges that either Parent or any of its Subsidiaries is not in compliance with any Environmental Laws or any Material Permit required under any applicable Environmental Law, or that it is liable under any Environmental Law, or that it is responsible (or potentially responsible) for the remediation of any Materials of Environmental Concern at, on or beneath its facilities or at, on or beneath any land adjacent thereto or any other property.
- (c) There are no past or present facts, circumstances, activities, practices or conditions existing at the facilities currently or formerly owned or operated by Parent, including the release of any Materials of Environmental Concern, that would reasonably be expected to give rise to any liability or result in a claim against Parent or any of its Subsidiaries under any Environmental Law or that would interfere with or prevent compliance with any Material Permits.

A-42

Table of Contents

- (d) Neither Parent nor any of its Subsidiaries has used, generated, manufactured, discharged, assembled, processed, stored, released, disposed of, arranged for the disposal of or transported any Materials of Environmental Concern except in material compliance with all Material Permits and Environmental Laws
- (e) Neither Parent nor any of its Subsidiaries is a party to any federal, state, local or private litigation, proceedings, administrative action, or, to the knowledge of Parent, is the subject of any investigation involving a demand for damages or other potential liability under any Environmental Laws, and neither Parent nor any of its Subsidiaries has received or, to the knowledge of Parent, is subject to any order or decree of any Governmental Authority relating to a violation of or liability under Environmental Laws.
- (f) To the knowledge of Parent, no underground storage tanks or surface impoundments exist on any property currently or formerly owned or leased by Parent or its Subsidiaries.
- 4.16 Intellectual Property.
- (a) To the knowledge of Parent, each of Parent and its Subsidiaries owns, is licensed or otherwise possesses the rights to use and license (subject to any existing licenses or other grants of rights to third parties) all patents (including any registrations, continuations, continuations in part, divisionals, renewals, reexaminations, reissues and applications therefor), copyrights, trademarks, service marks, trade names, Uniform Resource Locators and Internet URLs, designs, slogans, computer programs and other computer software, databases, technology, trade secrets and other confidential information, know-how, processes, formulae, algorithms, models, user interfaces, customer lists, inventions, source codes and object codes, methodologies, architecture, structure, display screens, layouts, development tools, instructions, templates, trade dress, logos and all documentation and media constituting, describing or relating to each of the foregoing, together with all goodwill related to any of the foregoing, in each case as is necessary to conduct their respective businesses as presently conducted (collectively, the Parent Intellectual Property Rights).
- (b) Except as set forth in Section 4.16(b) of the Parent Disclosure Schedule, (i) to the knowledge of Parent, all Parent Intellectual Property Rights that have been registered with any Governmental Authority are valid and subsisting and (ii) as of the Closing Date, in connection with such registered Parent Intellectual Property Rights, all necessary registration, maintenance and renewal fees will have been paid and all necessary documents and certificates have been filed with the relevant Governmental Authorities.
- (c) To the knowledge of Parent, neither Parent nor any of its Subsidiaries is, or will as a result of the consummation of the Merger or other transactions contemplated by this Agreement be, in breach in any material respect of any license, sublicense or other agreement relating to the Parent Intellectual Property Rights, or any licenses, sublicenses and other agreements as to which Parent or any of its Subsidiaries is a party and pursuant to which Parent or any of its Subsidiaries uses any patents, copyrights (including software), trademarks or other intellectual property rights of or owned by third parties material to the conduct of the business of Parent and its Subsidiaries (the Parent Third Party Intellectual Property Rights).
- (d) Except as set forth in Section 4.16(d) of the Parent Disclosure Schedule, neither Parent nor any of its Subsidiaries has been named as a defendant in any suit, action or proceeding which involves a claim of infringement or misappropriation of any Parent Third Party Intellectual Property Right. Except as set forth in Section 4.16(d) of the Parent Disclosure Schedule, neither Parent nor any of its Subsidiaries has as of the date of this Agreement received any written notice of any actual or alleged infringement, misappropriation or unlawful or unauthorized use of any Parent Third Party Intellectual Property. With respect to the Parent Identified Products, Parent does not infringe any third party intellectual property rights other than Parent Third Party Intellectual Property Rights. With respect to the Parent Identified Products that are not yet being marketed, to the knowledge of Parent, after the same are marketed, such marketing would not infringe any Parent Third Party Intellectual Property Rights.
- (e) As of the date hereof, to the knowledge of Parent, no other Person is materially infringing, misappropriating or making any unlawful or unauthorized use of any Parent Intellectual Property Rights. As of the Closing Date, to the knowledge of Parent, no Person will be infringing, misappropriating or making any unlawful or unauthorized use of any Parent Intellectual Property Rights.

A-43

Table of Contents

- 4.17 *Brokers*. No broker, financial advisor, investment banker or other financial intermediary is entitled to any fee, commission or expense reimbursement in connection with the Merger or other transactions contemplated by this Agreement based upon arrangements made by or on behalf of Parent, other than Goldman, Sachs & Co.
- 4.18 *Interim Operations of Merger Sub*. Merger Sub was formed solely for the purpose of engaging in the transactions contemplated by this Agreement, has engaged in no other business activities and has conducted its operations only as contemplated in this Agreement.
- 4.19 Ownership of Company Common Stock. Neither Parent nor any of Parent's Affiliates or Associates directly or indirectly owns, and at all times since August 1, 2002, neither Parent nor any of Parent's Affiliates directly or indirectly has owned, beneficially or otherwise, 15% or more of the outstanding Company Common Stock, as those terms are defined in Section 203 of the DGCL.
- 4.20 *Financing*. At the Closing, Parent, NAH and Merger Sub will have sufficient funds to perform all of their respective obligations under this Agreement and to consummate the Merger.
- 4.21 *Solvency*. As of and immediately following the Effective Time, (a) Parent shall be able to pay its debts as they become due and (b) Parent shall have adequate capital to carry on its business.
- 4.22 Capital Resources. Parent has cash on hand and committed financing that, together, will at the Effective Time be sufficient to pay the aggregate Per Share Cash Consideration. Parent has made available to the Company a copy of the commitment letter, dated June 1, 2007, setting forth the terms of the committed financing available to Parent in connection with the transactions contemplated by this Agreement (the Commitment Letter), which Commitment Letter is in full force and effect. All commitment and other fees required to be paid under the Commitment Letter prior to the date hereof have been paid. To the knowledge of Parent, there is no state of facts existing as of the date of this Agreement that will preclude satisfaction as of the Effective Time of the conditions contemplated by the Commitment Letter.
- 4.23 *Opinion of Financial Advisor*. The Parent Board has received the opinion of its financial advisor, Goldman, Sachs & Co., dated as of the date of this Agreement, to the effect that the consideration to be paid in the Offer and the Merger is fair, from a financial point of view, to Parent.

5. CONDUCT OF BUSINESS PENDING THE MERGER

- 5.1 Conduct of Business by Company Pending the Merger. The Company covenants and agrees that, beginning on the date hereof and the ending at the earlier to occur of the Closing or such earlier time as this Agreement is terminated in accordance with Section 8 (such period being hereinafter referred to as the Interim Period), except as expressly provided or permitted by this Agreement or set forth in Section 5.1 of the Company Disclosure Schedule or unless Parent shall otherwise consent in writing (which consent shall not be unreasonably withheld, conditioned or delayed), the Company shall, and shall cause each of its Subsidiaries to, use reasonable best efforts to (i) conduct its business in the ordinary course and (ii) preserve intact its business organization, properties and assets, including keeping available the services of their officers, employees and consultants, maintaining in effect all Company Material Contracts and preserving its relationships, customers, licensees, manufacturers, suppliers and other Persons with which it has material business relations. Except as expressly provided or permitted by this Agreement or as set forth in Section 5.1 of the Company Disclosure Schedule, during the Interim Period, the Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, do any of the following without the prior written consent of Parent (which consent shall not, in the case of any matter described in clauses (k), (o) or (p) and, to the extent relating to such clauses, clause (u), be unreasonably withheld, conditioned or delayed):
- (a) amend their Certificate of Incorporation, Bylaws or other equivalent organizational documents;
- (b) issue, sell, transfer, pledge, dispose of or encumber any shares of capital stock of any class, or any options, warrants, convertible securities or other rights of any kind to acquire any shares of capital stock, of

A-44

Table of Contents

the Company or any of its Subsidiaries (except for the issuance of shares of Company Common Stock pursuant to the Company Equity Awards outstanding on the date of this Agreement and listed on Section 3.3(e) of the Company Disclosure Schedule;

- (c) redeem, repurchase or otherwise acquire, directly or indirectly, any shares of capital stock of the Company or any equity interest in or securities of any of its Subsidiaries, other than (i) forfeitures of Company Common Stock pursuant to any right of forfeiture pursuant to Restricted Stock Unit Awards and Performance Shares Awards under the Company Equity Plans; (ii) in connection with any cashless exercise of any Company Stock Options in accordance with the terms of the Company Equity Plans; or (iii) any net settlement of Restricted Stock Units or Performance Shares Awards in accordance with the Company Equity Plans;
- (d) sell, transfer, pledge, dispose of or encumber any material properties, facilities, equipment or other assets, except for sales of inventory or equipment in the ordinary course of business;
- (e) declare, set aside or pay any dividend or other distribution (whether in cash, stock or other securities or property, or any combination thereof) in respect of any of its capital stock or other equity interests (except that a wholly owned Subsidiary of the Company may declare and pay cash dividends to the Company or any of its Subsidiaries);
- (f) split, combine or reclassify any shares of its capital stock or other securities or equity interests, or issue any other securities in respect of, in lieu of or in substitution for shares of its capital stock or equity interests;
- (g) sell, transfer, lease, license, sublicense, mortgage, pledge, encumber, grant or otherwise dispose of any Company Intellectual Property Rights, or amend or modify in any material respect any existing Company Material Agreements with respect to any Company Intellectual Property Rights;
- (h) acquire (by merger, consolidation, acquisition of stock or assets or otherwise) an interest in any corporation, limited liability company, partnership, joint venture or other business organization or division thereof provided this shall not prevent the Company or its Subsidiaries from investing its cash and cash equivalents in short-term investments consistent with its investment policy as disclosed to Parent prior to the date hereof;
- (i) incur any indebtedness for borrowed money or issue any debt securities or assume, guarantee (other than guarantees of bank debt of the Company's Subsidiaries entered into in the ordinary course of business) or endorse or otherwise as an accommodation become responsible for the obligations of any Person, or make any loans, advances or enter into any financial commitments, except in each case (A) in the ordinary course of business and (B) as otherwise permitted under any loan or credit agreement to which the Company or any of its Subsidiaries is a party as of the date of this Agreement;
- (j) authorize any capital expenditures outside of the Company s budget in excess of \$2,000,000 individually or \$5,000,000 in the aggregate;
- (k) except as required to comply with Law or agreements, plans or arrangements existing on the date hereof (including this Agreement), (A) increase the compensation payable to its officers or employees, except for increases in salary or wages in the ordinary course of business consistent with past practice; (B) grant any additional severance or termination pay to, or enter into any employment or severance agreements with, its officers; (C) pay any severance or termination pay to any employee, except in the ordinary course of business and consistent with past practice; (D) enter into any collective bargaining agreement; (E) establish, adopt, enter into or amend in any material respect any bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, trust, fund, policy or other enforceable arrangement for the benefit of any of its directors, officers or employees, other than, in the ordinary course of business or (F) enter into any employment arrangement, except in the ordinary course of business; provided, however, nothing in this subsection (k) shall prohibit the Company from awarding cash bonuses (i) payable after the Closing, (ii) in

A-45

Table of Contents

an aggregate amount not to exceed \$3,000,000, (iii) with no individual bonus (other than the bonus payable to the Company s chief executive officer) in excess of \$180,000 and (iv) with no individual bonus in excess of 50% of the recipient s current base salary;

- (l) make any material changes to the personnel or business policies of the Company;
- (m) materially change any accounting policies or procedures (including procedures with respect to reserves, revenue recognition, payments of accounts payable and collection of accounts receivable), unless required by statutory accounting principles or GAAP;
- (n) create, incur, suffer to exist or assume any material Lien on any of its material properties, facilities or other assets, other than any Lien for Taxes not yet due;
- (o)(A) enter into any new Company Material Contract pursuant to which the Company and its Subsidiaries will spend or receive (or are reasonably expected to spend or receive) in the aggregate more than \$200,000 during the current or next fiscal year; (B) amend in any material respect or terminate (other than in accordance with its terms) any Company Material Contract or waive, release or assign any material rights or claims thereto or thereunder; (C) enter into or extend any material lease with respect to real property or (D) initiate or participate in any new clinical trials or clinical trial or clinical development program;
- (p) enter into any agreement, or amend or waive the terms of any existing agreement, which grants to any Person exclusive supply, manufacturing, warehousing, production, marketing or distribution rights with respect to any of its products or technologies or enter into any material collaboration agreement, material license, co-marketing or co-promotion agreement or any other such material agreement with respect to the Company s Intellectual Property Rights;
- (q) make any material Tax election or settle or compromise any material federal, state, local or foreign Tax liability, or agree to an extension of a statute of limitations with respect thereto except as made in connection with the end of Company s fiscal year;
- (r) pay, discharge, satisfy or settle any material litigation or waive, assign or release any material rights or claims with respect thereto, other than settlements in the ordinary course of business requiring no obligation other than the payment of cash not in excess of \$250,000 in the aggregate and no admission being made with respect to (A) any criminal wrongdoing or (B) the invalidity or unenforceability of, or any infringement with respect to, any Company Intellectual Property Rights;
- (s) except as contemplated by Section 2.18 or as required by the Company Equity Plans, accelerate or otherwise amend the terms of any outstanding options under the Company Equity Plans;
- (t) fail to maintain in full force and effect all material insurance policies currently in effect, or permit any of the coverage thereunder to lapse, in each case, without simultaneously securing replacement insurance policies which will be in full force and effect and provide coverage substantially similar to or greater than under the prior insurance policies; or
- (u) enter into any agreement or contract to do any of the foregoing.
- 5.2 No Solicitation of Transactions.
- (a) The Company shall, and shall cause its officers, directors, employees, auditors, attorneys and financial advisors and any other agents (each, a Representative) to, immediately cease any discussions, negotiations or written communications (other than communications solely directed at informing other parties of the restrictions contained in this Section 5.2 and only in response to an inquiry from such other parties) with any party or parties that commenced prior to the execution of this Agreement with respect to any Competing Proposal. As used in this Agreement, a Competing Proposal means any proposal, offer or indication of interest (other than this Agreement, the Offer and the Merger), whether in writing or otherwise, from any Person or group (as defined in Section 13(d)(3) of the Exchange Act) other than Parent, NAH,

A-46

Table of Contents

Merger Sub or any affiliates thereof (a <u>Third Party</u>), relating to (i) any acquisition or purchase, directly or indirectly, of more than 20% of the consolidated total assets of the Company and its Subsidiaries or more than 20% of any class of equity or voting securities of the Company; (ii) any tender offer (including a self-tender offer) or exchange offer that, if consummated, would result in any Third Party beneficially owning more than 10% of any class of equity or voting securities of the Company or (iii) a merger, consolidation, share exchange, business combination, sale of substantially all the assets, reorganization, recapitalization, liquidation, dissolution or other similar transaction involving the Company.

(b) During the Interim Period, the Company shall not, nor shall it authorize or permit any Representative of the Company or its Subsidiaries to (i) solicit, initiate or knowingly encourage, or otherwise knowingly facilitate, directly or indirectly, any inquiries relating to, any Competing Proposal; (ii) directly or indirectly initiate or participate in any discussions, negotiations or communications (other than communications solely directed at informing other parties of the restrictions contained in this Section 5.2 and only in response to an inquiry from such other parties) regarding any Competing Proposal; (iii) furnish to any Third Party any nonpublic information or data for the purpose of encouraging or facilitating, or, except where failure to do so would cause the Company Board to breach its fiduciary obligations under applicable Laws, provide access to the properties, offices, books, records, officers, directors or employees of the Company for the purpose of encouraging or facilitating, any Competing Proposal; or (iv) release any party from or waive any provision of, any confidentiality or standstill agreement to which it is a party. Without limiting the generality of the foregoing, it is understood that any violation of any of the restrictions set forth in this Section 5.2 by any Representative of the Company or any of its Subsidiaries shall be and be deemed to be a breach by the Company of this Section 5.2 by the Company. Notwithstanding the foregoing and anything to the contrary contained in this Agreement, if, prior to the closing of the polls at the Company Special Meeting on the proposal to adopt this Agreement, the Company Board determines (after consultation with outside counsel and its financial advisor) that a Competing Proposal (other than one made following a breach by the Company of this Section 5.2) constitutes or is reasonably likely to lead to a Superior Competing Proposal, then the Company may, but only to the extent that the Company Board determines in good faith after consultation with the Company s outside counsel that failure to do so would cause the Company Board to breach its fiduciary obligations under applicable Laws and subject to the Company providing prompt (but in any event within 12 hours) written notice to Parent of its decision to take such action and compliance by the Company with Section 5.2(d), (A) furnish information or data with respect to the Company and its Subsidiaries to such Third Party (and the Representatives of such Third Party); (B) participate in discussions and negotiations directly or through its Representatives with such Third Party, subject to, in the case of clause (A) and, to the extent such discussions or negotiations include the disclosure by the Company, any of its Subsidiaries, or any of their Representatives of any material nonpublic information, clause (B), a confidentiality agreement not less favorable to the Company than the Confidentiality Agreement, provided, that all such material nonpublic information not already provided or made available to Parent is provided to Parent as soon as reasonably practicable (but in any event within 12 hours) after it is provided to such Third Party and (C) amend, or grant a waiver or release under, any standstill or similar agreement with respect to any Company Common Stock, but only to the extent necessary to permit the making and subsequent consummation of a Superior Competing Proposal by a Third Party. For purposes of this Agreement, <u>Superior Competing Proposal</u> shall mean a bona fide written proposal or offer made by a Third Party to acquire, directly or indirectly, including pursuant to a tender offer, exchange offer, sale of shares of stock, sale of assets, merger, consolidation, business combination, recapitalization, liquidation, dissolution or similar transaction, more than 50% of the capital stock of the Company then outstanding or more than 50% of the consolidated total assets of the Company and its Subsidiaries (i) on terms the Company Board determines in good faith (after consulting the Company s outside legal counsel and financial advisor) are more favorable to the holders of Company Common Stock than the Transactions, taking into account, among other things, relevant legal, financial, regulatory, timing and other aspects of the offer and the Third Party making the offer and the terms and conditions of this Agreement and (ii) which is reasonably capable of being consummated.

A-47

(c) Neither the Company Board nor any committee thereof shall (i) withdraw or modify, or publicly propose or publicly resolve to withdraw or modify, in a manner adverse to Parent, NAH or Merger Sub, the Recommendation; (ii) approve or recommend, or publicly propose or publicly resolve to approve or recommend, any Competing Proposal; (iii) approve or recommend, or execute or enter into, or publicly propose or publicly resolve to approve or recommend or execute or enter into, any letter of intent, agreement in principle, merger agreement, stock purchase agreement, asset purchase agreement, acquisition agreement, option agreement or similar agreement relating to a Competing Proposal (other than a confidentiality agreement referred to in Section 5.2(b) entered into in the circumstances referred to in Section 5.2(b) (an Acquisition Agreement); (iv) approve or recommend, or execute or enter into, or publicly propose or publicly resolve to approve or recommend or execute or enter into, any agreement requiring it to terminate this Agreement or abandon or fail to consummate the Offer, the Merger or the other Transactions contemplated hereby or (v) take any action necessary to render the provisions of any moratorium, control share, fair price, transaction, business combination, or other anti-takeover Laws and regulations of any state or other jurisdiction, including the provisions of Section 203 of the DGCL, inapplicable to any Competing Proposal. Notwithstanding the foregoing or anything to the contrary set forth in this Agreement, prior to obtaining the Company Requisite Stockholder Approval, the Company Board may, in response to a Superior Competing Proposal (other than one made following a breach by the Company of this Section 5.2), (1) withdraw or modify the Recommendation or (2) approve or recommend the Superior Competing Proposal, but in the case of any action described in clause (1) or any recommendation described in clause (2), only (x) if the Company Board determines in good faith, after consultation with the Company s outside counsel, that failure to do so would violate its fiduciary obligations under applicable Laws, and in each case, only (y) at a time that is after 3 Business Days following Parent s receipt of written notice advising Parent that the Company Board intends to take such action (during which period the Company shall make its chief executive officer, chief financial officer and its counsel available to Parent to meet on at least two separate occasions for a total of at least 12 hours, and shall consider in good faith any amendment of the terms of the Offer and/or the Merger proposed by Parent, NAH or Merger Sub or any proposal by Parent, NAH or Merger Sub to amend the terms of this Agreement, the Offer or the Merger), specifying therein the material terms and conditions of such Superior Competing Proposal and identifying the Person or group making such Superior Competing Proposal, provided that such 3 Business Day period shall be extended for an additional 3 Business Days following each material modification of such Competing Proposal occurring after the receipt of Parent s proposal, and (z) if, after the end of such 3 Business Day period (plus each applicable extension), the Company Board determines in good faith (after consultation with the Company s outside legal counsel and its financial advisor) that such proposed transaction continues to be a Superior Competing Proposal, after taking into account any bona fide proposal by Parent, NAH or Merger Sub to amend the terms of this Agreement and the Merger. The Company shall not during the term of this Agreement release any Third Party from, or agree to amend or waive any provision of, any confidentiality agreement with any Third Party with respect to a Competing Proposal entered into pursuant to this Section 5.2 other than the standstill provision as provided above, and the Company shall use its reasonable best efforts to enforce, to the fullest extent permitted by Law, each confidentiality agreement entered into pursuant to this Section 5.2. In addition, notwithstanding the foregoing or anything to the contrary set forth in this Agreement, prior to obtaining the Company Requisite Stockholder Approval, the Company Board may withdraw or modify the Recommendation if the Company Board determines, after consultation with outside counsel, that failure to do so would violate its fiduciary obligations under applicable Law, but only at a time that is after 3 Business Days following Parent s receipt of written notice advising Parent that the Company Board is prepared to take such action, specifying the reasons therefor.

affiliate

(d) In addition to the obligations set forth in Sections 5.2(a), (b) and (c), the Company shall notify Parent orally and, if requested by Parent, in writing, within 12 hours of the receipt by the Company of any Competing Proposal, which notification shall include the material terms and conditions of such Competing Proposal and the identity of the person or group making or sending the Competing Proposal. The Company shall promptly (but in any event within 12 hours) advise Parent of any material change in the terms or conditions of a Competing Proposal or any other material development with respect thereto.

A-48

Table of Contents

- (e) Nothing contained in this Section 5.2 or any other provision hereof shall prohibit the Company or the Company Board from (i) taking and disclosing to the Company s stockholders pursuant to Rules 14d-9 and 14e-2 promulgated under the Exchange Act a position with respect to a tender or exchange offer by a Third Party or (ii) making any disclosure to the Company s stockholders, if the Company Board determines, after consultation with its outside counsel, that failure to so disclose would cause it to violate its fiduciary obligations under applicable Law.
- (f) Nothing in this Section 5.2 shall permit the Company to terminate this Agreement (except as expressly provided in Section 8).
- 5.3 *No Control of Other Party s Business*. Nothing contained in this Agreement is intended to give Parent, directly or indirectly, the right to control or direct the Company s or its Subsidiaries operations prior to the Effective Time, and nothing contained in this Agreement is intended to give the Company, directly or indirectly, the right to control or direct Parent s or its Subsidiaries operations.

6. ADDITIONAL AGREEMENTS

- 6.1 Offer Registration Statement; European Prospectus; Parent Circular.
- (a) As soon as reasonably practicable after the execution hereof Parent shall prepare and file the Offer Registration Statement with the SEC and prepare and file with the AFM a draft of a listing disclosure document (as amended or supplemented, the <u>European Prospectus</u>) under the European Union Prospectus Directive and any rules and regulations relating thereto (the <u>EU Prospectus Directive</u>) and applicable Dutch securities Laws, in each case with respect to the Parent Common Shares issuable and deliverable pursuant to this Agreement. The Company and Parent shall cooperate with each other in the preparation of the Offer Registration Statement and the European Prospectus. Parent shall prepare the European Prospectus to comply as to form in all material respects with the EU Prospectus Directive and applicable Dutch securities Laws. Each of Parent and the Company shall use its reasonable best efforts to, as promptly as reasonably practicable after the filing of the European Prospectus, have the European Prospectus approved by the AFM. Parent shall advise the Company, promptly after it receives notice, of the time when each of the Offer Registration Statement and the European Prospectus has become effective or been approved, as applicable, or any supplement or amendment to either such document has been filed, the issuance of any stop order, or the suspension of the qualification of the Parent Common Shares issuable and deliverable in connection with the Offer and the Merger for offering or sale in any jurisdiction. Each of Parent and the Company shall, as soon as reasonably practicable, notify the other of the receipt of any comments from the SEC or the AFM with respect to the Offer Registration Statement or the European Prospectus, as applicable, or for additional information.
- (b) The Company shall, upon request of Parent, furnish Parent with all information concerning itself, its Subsidiaries, directors, officers and stockholders, as applicable, and any other matters as may be necessary or advisable in connection with Offer Documents, the European Prospectus (including information required to be set forth therein as required by the EU Prospectus Directive), any stockholders circular required to be prepared and distributed by Parent in connection with the Parent Stockholders Meeting (the Parent Circular or any other filing, notice, statement, registration, submission of information or application required to be made by or on behalf of the Company or Parent or any of their respective Subsidiaries to any third-party and/or Governmental Authority in connection with the Offer, the Merger and the other Transactions. If at any time prior to the Effective Time any information relating to the Company or Parent, or any of their respective affiliates, officers or directors, should be discovered by the Company or Parent which should be set forth in an amendment and/or a supplement to the Offer Documents, the European Prospectus or the Parent Circular, so that the Offer Documents, European Prospectus and the Parent Circular would not, in the case of the Offer Documents, at the time the Offer is commenced and at the Acceptance Date, in the case of the European Prospectus, at the time the European

A-49

Table of Contents

Prospectus is made publicly available, or, in the case of the Parent Circular, at the date it is made publicly available and at the time of the Parent Stockholders Meeting, include any misstatement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading, the party which discovers this information shall promptly notify the other party and, to the extent required by Law, an appropriate amendment or supplement describing that information shall be promptly filed with the AFM or any other applicable Governmental Authority and, to the extent required by Law, disseminated to Parent s stockholders.

- (c) Preparation of Merger Registration Statement and Proxy Statement.
- (i) If required by applicable Laws, promptly after the acceptance for exchange or payment of shares of Company Common Stock pursuant to the Offer, Parent and the Company shall prepare and file with the SEC, the Merger Registration Statement and the Proxy Statement. Parent shall provide the Company and its counsel with any comments it may receive from the SEC or its staff with respect to the Merger Registration Statement as promptly as reasonably practicable after receipt of such comments and the parties shall cooperate to prepare appropriate responses to the SEC to such comments and make such modifications to the Merger Registration Statement as shall be reasonably appropriate. The Company shall provide the Parent and its counsel with any comments it may receive from the SEC or its staff with respect to the Proxy Statement as promptly as reasonably practicable after receipt of such comments and the parties shall cooperate to prepare appropriate responses to the SEC to such comments and make such modifications to the Proxy Statement as shall be reasonably appropriate. Each of the Company and Parent shall use all reasonable efforts to have the Merger Registration Statement declared effective under the Securities Act as promptly as reasonably practicable after the acceptance for payment or exchange of shares of Company Common Stock pursuant to the Offer and to keep the Merger Registration Statement effective as long as is necessary to consummate the Merger. The Company shall furnish all information concerning the Company as Parent may reasonably request in connection with the preparation of the Merger Registration Statement. The Parent shall furnish all information concerning the Parent and the Company may reasonably request in connection with the preparation of the Proxy Statement. If required by applicable Laws, the Company shall use its reasonable best efforts to mail the Proxy Statement to its stockholders as promptly as reasonably practicable after the Merger Registration Statement is declared effective under the Securities Act and, if necessary, after the Proxy Statement shall have been so mailed, promptly circulate amended, supplemental or supplemented proxy material, and, if required in connection therewith, resolicit proxies. No amendment or supplement to the Merger Registration Statement or Proxy Statement will be made by Parent or the Company without the approval of the other party, which will not be unreasonably withheld or delayed. Each party will advise the other party promptly, after it receives notice thereof, of the time when the Merger Registration Statement is declared effective or any supplement or amendment thereto has been filed, of the issuance of any stop order, of the suspension or qualification of Parent Common Shares issued in connection with the Merger for offering or sale in any jurisdiction, or of any request by the SEC for amendment of the Proxy Statement or comments thereon and responses thereto or requests by the SEC for additional information. If, at any time prior to the Effective Time, the Company or Parent discovers any information relating to either party, or any of their respective affiliates, officers or directors, that should be set forth in an amendment to either the Merger Registration Statement or the Proxy Statement so that such document would not contain any misstatement of material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party that discovers that information shall promptly notify the other party and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC and, to the extent required by law or regulation, disseminated to the stockholders of the Company.

(ii) Notwithstanding the foregoing, if Parent or Merger Sub shall acquire at least 90% of the outstanding shares of Company Common Stock pursuant to the Offer or otherwise, the parties hereto

A-50

Table of Contents

agree, subject to the satisfaction or (to the extent permitted hereunder) waiver of all conditions to the Merger, to take, or cause to be taken, all necessary and appropriate action to cause the Merger to be effective as soon as reasonably practicable after the acceptance for payment or exchange of shares of Company Common Stock pursuant to the Offer without the Company Stockholder Meeting.

- (d) Parent will use its reasonable best efforts to cause the Parent Circular and the European Prospectus to be made publicly available as promptly as reasonably practicable after the Offer Registration Statement is declared effective.
- (e) As promptly as reasonably practicable after the date of this Agreement, the Parties shall prepare and file any other filings required under the Exchange Act, the Securities Act or any other federal or state securities Laws relating to the Merger and the other transactions contemplated by this Agreement (collectively, the Other Filings).
- (f) The Parties shall use their respective reasonable best efforts to cause the Schedule TO, the Schedule 14D-9, the Offer Registration Statement, the Parent Circular, the European Prospectus, the Merger Registration Statement, the Proxy Statement and the Other Filings to comply in all material respects with all requirements of Laws. Whenever any event or circumstance occurs which is required under the Securities Act, the Exchange Act or other Laws to be set forth in an amendment or supplement to the Schedule TO, the Schedule 14D-9, the Offer Registration Statement, the Parent Circular, the European Prospectus, the Merger Registration Statement, the Proxy Statement or any Other Filing, each Party shall promptly upon becoming aware thereof inform the other of such occurrence and the relevant Party shall promptly prepare an amendment, supplement or filing, as appropriate, accurately describing such event or circumstance and provide the other Party reasonable opportunity under the circumstances to review and comment, and cooperate in filing with the SEC, its staff or any other Governmental Authority, and/or mailing to stockholders of the Company, such amendment or supplement.
- (g) Subject to Section 5.2(c), the Proxy Statement shall include the Recommendation.
- 6.2 Stockholders Meetings.
- (a) If required by applicable Laws, the Company, acting through the Company Board, shall, in accordance with applicable Laws, duly call, convene and hold the Company Special Meeting, as soon as reasonably practicable after the acceptance for payment or exchange of shares of Company Common Stock pursuant to the Offer, for the purpose of voting upon this Agreement and the Merger, and the Company agrees that this Agreement shall be submitted at such meeting. The Company shall (i) take all action necessary to duly call, give notice of, convene and hold the Company Special Meeting, (ii) subject to Section 5.2(c), include in the Company Proxy Statement the Recommendation and (iii) use its reasonable best efforts to obtain the Company Requisite Stockholder Approval. The Company shall solicit from stockholders of the Company proxies in favor of the adoption of this Agreement and shall take all other action reasonably necessary or advisable to secure the Company Requisite Stockholder Approval.
- (b) As soon as reasonably practicable following the date of this Agreement, the effectiveness of the Offer Registration Statement and the approval of the European Prospectus by the AFM, Parent, acting through its Supervisory and/or Managing Boards, shall (i) take all action necessary to duly call, give notice of, convene and hold a general meeting of its stockholders (such meeting, including any adjournments or postponements, the Parent Stockholders Meeting) for the purposes of, among other things, obtaining the Parent Requisite Stockholder Approval and (ii) use its reasonable best efforts to obtain the Parent Requisite Stockholder Approval.
- 6.3 Access to Information; Confidentiality.
- (a) Upon reasonable notice, during normal business hours and in a manner that does not disrupt or interfere with business operations, Parent and the Company shall (and shall cause each of their respective

A-51

Table of Contents

Subsidiaries to) afford to the officers, employees, accountants, counsel and other representatives of the other Party reasonable access, during the Interim Period, to all its properties, books, contracts, commitments and records and, during such period, furnish promptly to the other Party such information concerning its business, properties and personnel as the other Party may reasonably request. Parent and the Company shall make available to the other, upon reasonable notice, during normal business hours and in a manner that does not disrupt or interfere with business operations, the appropriate individuals for discussion of its business, properties and personnel as the other may reasonably request.

- (b) The Parties shall keep all information obtained pursuant to Section 6.3(a) confidential in accordance with the Confidentiality Agreement dated May 7, 2007 (the <u>Confidentiality Agreement</u>), between Parent and the Company.
- 6.4 Reasonable Best Efforts; Further Assurances.
- (a) Parent and the Company shall use their reasonable best efforts to satisfy or cause to be satisfied all of the conditions precedent that are set forth in Section 7, as applicable to each of them. Each Party, at the reasonable request of the other, shall execute and deliver such other instruments and do and perform such other acts and things as may be reasonably necessary and consistent with this Agreement to effect the consummation of the Merger and other transactions contemplated by this Agreement.
- (b) Subject to the terms and conditions hereof, the Company and Parent agree to use their respective reasonable best efforts to take, or cause to be taken, all action and to do, or cause to be done, all things necessary, proper or advisable to promptly consummate and make effective the Merger and other transactions contemplated by this Agreement, including using their respective reasonable best efforts: (i) to effect all necessary registrations and filings required by any Governmental Authority (in connection with which Parent and the Company shall cooperate with each other in connection with the making of all such registrations and filings, including providing copies of all such documents to the non-filing party and its advisors prior to the time of such filing and, if requested, will consider in good faith reasonable additions, deletions or changes suggested in connection therewith) and (ii) to furnish to each other such information and assistance as reasonably may be requested in connection with the foregoing.
- (c) The Company shall use its reasonable best efforts to promptly obtain all consents, approvals, authorizations, and waivers of, and to give all notices to, each third party that may be necessary for the performance of its obligations under this Agreement and the consummation of the transactions contemplated hereby, including obtaining all consents, approvals, authorizations and waivers, and giving all notices listed in Section 3.4(c) of the Company Disclosure Schedule; provided, however, that the Company shall not amend or agree to amend, or waive any material right or material economic benefit under, any Company Material Contract in connection with obtaining such consents, approvals, authorizations and waivers without Parent s prior written consent (which consent shall not be unreasonably withheld or delayed); and provided, further, that, in connection with obtaining such consents, approvals, authorizations and waivers, or the giving of such notices, the Company shall not incur any out of pocket costs or any other obligation or liability without Parent s prior written consent (which consent shall not be unreasonably withheld or delayed), unless such costs, obligation or liability is *de minimis* in nature.
- (d) Parent, on the one hand, and the Company, on the other hand, shall, in connection with the efforts referenced in Section 6.4(b) to obtain all requisite approvals and authorizations for the transactions contemplated by this Agreement under the HSR Act, ECMR or any other applicable Antitrust Law, use its reasonable best efforts to: (i) cooperate in all respects with each other in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private party; (ii) keep the other Party reasonably informed of the status of matters related to the transactions contemplated by this Agreement, including furnishing the other with any written notices or other communications received by such Party from, or given by such Party to, the Federal Trade Commission (the <u>FTC</u>), the Antitrust Division of the Department of Justice (the <u>DOJ</u>), the European Commission or any other Governmental Authority and of any notices or other communications received or

A-52

Table of Contents

given in connection with any proceeding by a private party, in each case regarding any of the transactions contemplated hereby and (iii) permit the other Party to review any communication given by it to, and consult with each other in advance of any meeting or conference with, the FTC, the DOJ or any other Governmental Authority or, in connection with any proceeding by a private party, with any other person, and to the extent permitted by the FTC, DOJ or such other applicable Governmental Authority or other person, give the other Party the opportunity to attend and participate in such meetings and conferences in accordance with Antitrust Law. For purposes of this Agreement, <u>Antitrust Law</u> means the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, the ECMR, and all other federal, state and foreign, if any, statutes, rules, regulations, orders, decrees, administrative and judicial doctrines and other Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition through merger or acquisition.

- (e) In furtherance and not in limitation of the covenants of the parties contained in Sections 6.4(b) and (d), each Party hereto shall use its reasonable best efforts to resolve such objections, if any, as may be asserted with respect to the transactions contemplated by this Agreement under any applicable Antitrust Law. Notwithstanding anything to the contrary in this Agreement, in connection with any filing or submission required or action to be taken by either Parent or the Company to consummate the Merger, in no event shall Parent or any of its Subsidiaries or affiliates be obligated to propose or agree to accept any material undertaking or condition, to enter into any consent decree, to make any material divestiture or to accept any material operational restriction.
- (f) In the event that any administrative or judicial action or proceeding is instituted (or threatened to be instituted) by a Governmental Authority or a private party challenging any transaction contemplated by this Agreement, each of the parties (i) shall cooperate with the others and use its respective reasonable best efforts to defend, contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any order that is in effect (until all appeals are exhausted, if necessary) that prohibits, prevents or restricts consummation of the transactions contemplated by this Agreement and (ii) shall defend, at their costs and expense, any action or actions, whether administrative or judicial, in connection with the transactions contemplated by this Agreement.
- (g) Notwithstanding the foregoing or any other provision of this Agreement, nothing in this Section 6.4 shall limit a Party s right to terminate this Agreement pursuant to Section 8.1 so long as such Party has up to then complied in all material respects with its obligations under this Section 6.4.
- (h) Notwithstanding the foregoing or any other provision of this Agreement, the Company shall not take any action, agree to take any action or consent to the taking of any action pursuant to this Section 6.4 (including with respect to selling, holding separate or otherwise disposing of assets or conducting its business in a specified manner) without the prior written consent of Parent.
- (i) Parent has advised the Company that Parent intends to seek financing of a portion of the aggregate Per Share Cash Consideration (<u>Financing</u>). The Parties agree that such Financing is not a condition to any Party s obligation to effect the Merger and consummate the other transactions contemplated hereby in accordance with this Agreement, and that Parent, NAH and Merger Sub shall be required to proceed with the Closing notwithstanding Parent s inability or failure to obtain such Financing. The Company agrees that it will use its reasonable best efforts to cooperate with Parent in connection with Parent s preparation of offering materials in connection with the Financing (provided that the Company shall not be required to assume any liability therefor). Subject to consent by the Company s independent accountants, Parent may use the Company s financial statements in connection with the Financing. In connection therewith, in each case at Parent s expense, the Company shall (i) provide any source of Financing with access to the Company s books and records and provide such party with such other information reasonably requested for purposes of verifying the Company s financial statements (subject to a confidentiality undertaking reasonably acceptable to the Company, and upon reasonable notice, during normal business hours and in a manner that does not disrupt or interfere with the Company s business operations) and (ii) do such other acts and things, as shall be reasonably requested by Parent or any source of Financing in connection with the

A-53

Table of Contents

same. Parent shall indemnify and hold harmless the Company and each person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Effective Time, a director, officer or Representative of the Company or any of its Subsidiaries, against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or relating to the Financing.

6.5 *IFRS Financials*. As promptly as reasonably practicable after the date hereof, the Company shall prepare and deliver to Parent such financial information as may be required by the AFM, BaFin, Frankfurt Stock Exchange and/or as reasonably requested by Parent s German counsel, including financial information prepared in accordance with International Financial Reporting Standards (IFRS), in the form required by the AFM, BaFin, Frankfurt Stock Exchange and/or as reasonably requested by Parent s German counsel, pursuant to the Commission Regulation 809/2004 implementing Directive 2003/71/EC (as further supplemented in the FMSA, and the rules and regulations promulgated thereunder).

6.6 Employee Benefits.

- (a) For the period of time commencing at the Effective Time and ending on the first anniversary of the Effective Time, Parent shall, or shall cause its Subsidiaries or the Surviving Company to, ensure that all employees and officers of the Company who are employed by Parent or the Surviving Company or their respective Subsidiaries upon and after the Effective Time (<u>Continuing Employees</u>) receive compensation and benefits that, in the aggregate, are substantially similar to the compensation and benefits received by such individuals immediately prior to the date hereof under applicable Company Employee Plans or other applicable compensation arrangements provided by the Company or any of its Subsidiaries. The provisions of this Section 6.6(a) shall not create in any current or former employee of the Company or its Subsidiaries any rights to employment or continued employment with the Parent, the Surviving Company or any of their respective Subsidiaries.
- (b) Parent shall provide each Continuing Employee full credit for prior service with the Company or any of its Subsidiaries or predecessors for purposes of eligibility and vesting under any benefit plans, policies and arrangements maintained by Parent or Surviving Company or their respective Subsidiaries (Parent Plans) to the same extent recognized under similar plans or arrangements of the Company or such Subsidiary in which any such Continuing Employee participated prior to and, to the extent applicable, following, the Effective Time. Further, Parent shall provide each Continuing Employee full credit for prior service with the Company or any of its Subsidiaries or predecessors for purposes of benefit accruals under Parent Plan providing for vacation or severance benefits. Parent shall maintain the group health plan (providing for group medical, vision and dental benefits) currently maintained by the Company through December 31, 2008. In addition, with respect to any group health plan maintained by Parent, the Surviving Company or any of their respective Subsidiaries, Parent shall waive, or cause to be waived, any waiting periods and pre-existing condition limitations for a Continuing Employee or a dependent of a Continuing Employee who is covered under the Company s group health plan, but only to the extent such waiting periods or pre-existing condition limitations were not otherwise applicable under the Company s group health plan. Parent shall give Continuing Employees credit towards any lifetime deductibles or limitations on out of pocket expenses to the extent a Continuing Employee incurred the same with respect to a Company Employee Plan and such deductibles or limitations are applicable under the Parent Plan.
- (c) The Company and Parent shall provide each other with such information as may be reasonably requested and take such action as may be necessary or appropriate to effect the matters set forth in this Section 6.6. All such information and notices and schedules to be provided hereunder shall be true, correct and complete as of the date provided.
- (d) Parent shall take all reasonable actions to effectuate the terms and conditions of this Section 6.6 including, without limitation, amending or causing to be amended the Parent Plans.

A-54

Table of Contents

- (e) Nothing herein expressed or implied shall (i) confer upon any of the Continuing Employees any rights or remedies (including, without limitation, any right to employment or continued employment for any specified period) of any nature or kind whatsoever under or by reason of the Agreement, or (ii) obligate Parent, the Surviving Company or any of their respective Subsidiaries to maintain any particular employee benefit plan or limit the ability of Parent, the Surviving Company or any of their respective Subsidiaries to terminate or amend any of such employee benefit plan to the extent permitted thereunder in accordance with their terms. None of the provisions of this Agreement are intended to constitute an amendment to any employee benefit plan and no Continuing Employee (or other party or person which is not a signatory to this Agreement) shall have the right to enforce or compel the enforcement of any provisions of this Section 6.6 or this Agreement.
- 6.7 Notification of Certain Matters. The Company shall give prompt notice to Parent, and Parent shall give prompt notice to the Company, upon becoming aware of (i) the occurrence, or failure to occur, of any event, which occurrence or failure to occur would reasonably be expected to result in any representation or warranty contained in this Agreement to be untrue or inaccurate such that the condition in Section 7.2(a) or 7.3(a) would not be satisfied and (ii) any failure of the Company, Parent, NAH or Merger Sub, as the case may be, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder such that the condition in Section 7.2(b) or 7.3(b) would not be satisfied; provided, however, that the delivery of any notice pursuant to this Section 6.7 shall not limit or otherwise affect the remedies available hereunder to the Party receiving such notice.
- 6.8 *Public Announcements*. Except as otherwise required by Law or the rules of the NGSM, or as provided elsewhere herein, prior to the Closing or the earlier termination of this Agreement pursuant to Section 8, (a) the press release announcing the execution of this Agreement shall be issued only in such form as shall be mutually agreed upon by the Company and Parent and (b) Parent and the Company shall each use its reasonable best efforts to consult with the other before issuing any other press release or otherwise making any public statement with respect to the Merger or this Agreement.
- 6.9 Accountant s Letters.
- (a) The Company shall use its reasonable best efforts to cause to be delivered to Parent a comfort letter, in customary form, of Ernst & Young dated within two (2) Business Days before each of: the date on which the Offer Registration Statement shall become effective, the date on which the Merger Registration Statement shall become effective, the date of the Acceptance Notice and as of the Effective Time, and addressed to Parent and the Company, in form and substance reasonably satisfactory to Parent.
- (b) Parent shall use its reasonable best efforts to cause to be delivered to the Company a comfort letter, in customary form, of Ernst & Young, dated a date within two (2) Business Days before each of: the date on which the Offer Registration Statement shall become effective, the date on which the Merger Registration Statement shall become effective, the date of the Acceptance Notice and as of the Effective Time, and addressed to Parent and the Company, in form and substance reasonably satisfactory to Parent.
- 6.10 Directors and Officers Insurance/Indemnification.
- (a) Parent shall negotiate (with the assistance of the Company or its representative) and purchase tail insurance coverage from the Company s existing directors and officers liability insurers, or from other insurers, that provides for a period of six (6) years that is substantially equivalent to the Company s existing directors and officers liability insurance program, or if substantially equivalent insurance coverage is not available, the best available coverage (D&O Tail Coverage); provided however that the aggregate cost for the purchase of such D&O Tail Coverage (for the entire six (6) year tail coverage period) shall not exceed more than 200% of the aggregate premium paid by the Company and its Subsidiaries for the existing directors and officers liability insurance program, provided, further, that should the cost of D&O Tail Coverage exceed the 200% cap, Parent shall instead purchase the best available coverage for 200% of the aggregate premium paid by the Company and its Subsidiaries for the existing directors and officers liability insurance program.

A-55

Table of Contents

- (b) Parent and the Surviving Company shall, until the sixth (6th) anniversary of the Effective Time, jointly and severally, indemnify and hold harmless, with respect to claims or events existing or occurring at or prior to the Effective Time, each person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Effective Time, a director or officer of the Company or any of its Subsidiaries (the Indemnified Parties), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or by reason of the fact that the Indemnified Party is or was an officer or director of the Company or any of its Subsidiaries, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent permitted under the DGCL for officers and directors of Delaware corporations.
- (c) The Certificate of Formation and Operating Agreement shall contain, and Parent shall cause the Certificate of Formation and Operating Agreement to so contain, provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of present and former directors and officers of the Company and its Subsidiaries than are presently set forth in the Certificate of Incorporation and Bylaws of the Company, and such commitments will be included and maintained in the Certificate of Incorporation and Bylaws of the Surviving Company upon its conversion to a corporation.
- 6.11 Stockholder Litigation. The Company shall promptly notify Parent of any stockholder litigation brought, or threatened in writing, against the Company and/or the members of the Company Board relating to the Merger or the transactions contemplated by this Agreement, and shall provide Parent with updates and such information as Parent shall reasonably request with respect to the status of the litigation and discussion between the parties thereto. The Company shall give Parent the opportunity to participate in the defense of and settlement discussions with respect to (but, in each case, not control) such litigation and the Company shall not make any payment or settlement offer prior to the Effective Time with respect to any such litigation unless Parent shall have consented in writing to such payment or settlement, which consent shall not be unreasonably withheld, conditioned or delayed.
- 6.12 Nasdaq Listing. Parent shall use reasonable best efforts to cause the Parent Common Shares to be issued as part of the Consideration to be approved for listing on the NGSM, on or prior to the Effective Time.
- 6.13 Affiliates. The Company shall use its reasonable best efforts to obtain an executed affiliate agreement substantially in the form attached hereto as Exhibit A, with such changes therein as are agreeable to Parent (each, a Company Affiliate Agreement), from each of the Persons identified in Section 6.13 of the Company Disclosure Schedule concurrently with the execution of this Agreement and thereafter from any other Person who is an officer or director of the Company or its Subsidiaries and who the Company hereafter determines to be an Affiliate of the Company regarding compliance with Rule 145 under the Securities Act.
- 6.14 Section 16(b). The Company and Parent shall take all steps reasonably necessary to cause the transactions contemplated by this Agreement and any other dispositions of equity securities of the Company (including derivative securities) in connection with the transactions contemplated by this Agreement by each individual who is a director or executive officer of the Company to be exempt under Rule 16b-3 promulgated under the Exchange Act.
- 6.15 *Tax Compliance*. Neither Parent nor NAH nor Company shall take or fail to take (and, following the Merger, Parent shall cause NAH and Surviving Company to take or not to fail to take) any action which action (or failure to act) would reasonably be expected to (i) cause the Offer and the Merger to fail to qualify as a reorganization within the meaning of Section 368(a) of the Code or (ii) cause the Merger to fail to satisfy the requirements of Treasury Regulation Section 1.367(a)-3(c)(i), (ii) or (iv) following the Merger. With respect to the Merger, Parent will (and following the Merger will cause its Subsidiaries to) file all required information with all relevant Returns and maintain all records required for Tax purposes, including, without limitation, the reporting requirements of the Company described in Treasury Regulation Section 1.367(a)-3(c)(6).

A-56

7. CONDITIONS OF MERGER

- 7.1 Conditions to Obligation of Each Party to Effect the Merger. The obligations of each Party to effect the Merger and consummate the other transactions contemplated hereby shall be subject to the satisfaction at or prior to the Closing of the following conditions, any of which may be waived in writing by the Party entitled to the benefit thereof, in whole or in part:
- (a) Effectiveness of the Offer Registration Statement. The Offer Registration Statement shall have become effective and no stop order or other order or directive suspending the effectiveness or approval of the Offer Registration Statement or the use of the Proxy Statement shall have been issued or initiated by the SEC.
- (b) Stockholder Approval. If required by applicable Law, this Agreement shall have been adopted by the Company Requisite Stockholder Approval and the Parent Requisite Stockholder Approval.
- (c) Nasdaq Listing. The Parent Common Shares issuable to the stockholders of the Company pursuant to this Agreement shall have been listed on the NGSM.
- (d) Antitrust. (i) The waiting period (and any extension thereof) applicable to the Merger under the HSR Act shall have been terminated or shall have expired; (ii) all required approvals by the European Commission applicable to the Merger under applicable Law shall have been obtained or any applicable waiting period thereunder shall have been terminated or shall have expired; (iii) all required approvals of the competent authority of any member state of the European Union applicable to the Merger under applicable Law shall have been obtained or any applicable waiting period thereunder shall have been terminated or shall have expired and (iv) all other required approvals of any Governmental Authority under applicable Law (including any Antitrust Law) shall have been obtained or any applicable waiting period thereunder shall have been terminated or shall have expired, in each case without requiring any divestiture, hold separate or any material operating condition.
- (e) No Injunctions or Restraints; Illegality. No Governmental Authority of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any order, executive order, stay, decree, judgment or injunction (preliminary or permanent) or statute, rule or regulation which is in effect and which has the effect of making the Merger illegal or otherwise prohibiting consummation of the Merger or the other transactions contemplated by this Agreement.
- (f) Relative Values of Parent and Company. On the Closing Date, the fair market value of Parent, computed according to the special rules contained in Treasury Regulation Section 1.367(a)-3(c)(3)(iii)(B), will be at least equal to the fair market value of Company.
- 7.2 Additional Conditions to Obligations of Parent. The obligations of Parent to effect the Merger are also subject to the following conditions, any and all of which may be waived in writing by Parent, in whole or in part:
- (a) Representations and Warranties. (i) Each of the representations and warranties of the Company set forth in Section 3.3 shall be true and correct in all material respects and (ii) each of the other representations and warranties of the Company contained in this Agreement (disregarding all qualifications and exceptions contained therein regarding materiality or a Company Material Adverse Effect or any similar standard or qualification) shall be true and correct except to the extent that all failures of such representations and warranties to be so true and correct (disregarding all qualifications and exceptions contained therein regarding materiality or a Company Material Adverse Effect or any similar standard or qualification), individually or in the aggregate, have not had, and would not reasonably be expected to have, a Company Material Adverse Effect, in each case of clauses (i) through (ii) above as of the date of this Agreement and as of the Closing Date as if made at and as of such time (except for representations and warranties made only as of a specified date, which shall be true and correct only as of the specified date). Parent shall have received a certificate to such effect signed on behalf of the Company by the Chief Executive Officer or Chief Financial Officer of the Company.

A-57

Table of Contents

- (b) Agreements and Covenants. The Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date. Parent shall have received a certificate to such effect signed on behalf of the Company by the Chief Executive Officer or Chief Financial Officer of the Company.
- (c) *Third Party Consents*. All consents, approvals and authorizations listed in Section 7.2(c) of the Company Disclosure Schedule shall have been obtained and evidence thereof, in form reasonably satisfactory to Parent, shall have been delivered to Parent and shall be in full force and effect as of the Closing.
- (d) Federal Tax Opinion. Parent shall have received the opinion of its counsel, Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C., in form and substance reasonably satisfactory to Parent, dated as of the Closing Date, substantially to the effect that, on the basis of facts, representations and assumptions set forth in such opinion that are consistent with the state of facts existing at the Effective Time, the Offer and the Merger together will be treated as a reorganization within the meaning of Section 368(a) of the Code. In rendering such opinion, counsel may require and rely upon customary representations contained in certificates of the Company, NAH and Parent and of their officers, directors and principal stockholders as are customary for such opinions.
- 7.3 Additional Conditions to Obligations of the Company. The obligation of the Company to effect the Merger is also subject to the following conditions, any and all of which may be waived in writing by the Company, in whole or in part:
- (a) Representations and Warranties. Each of the representations and warranties of Parent set forth in Section 4.3 shall be true and correct in all material respects and (ii) each of the other representations and warranties of Parent contained in this Agreement (disregarding all qualifications and exceptions contained therein regarding materiality or a Parent Material Adverse Effect or any similar standard or qualification) shall be true and correct except to the extent that all failures of such representations and warranties to be so true and correct (disregarding all qualifications and exceptions contained therein regarding materiality or a Parent Material Adverse Effect or any similar standard or qualification), individually or in the aggregate, have not had, and would not reasonably be expected to have, a Parent Material Adverse Effect, in each case of clauses (i) through (ii) above as of the date of this Agreement and as of the Closing Date as if made at and as of such time (except for representations and warranties made only as of a specified date, which shall be true and correct only as of the specified date). Company shall have received a certificate to such effect signed on behalf of Parent by the Chief Executive Officer or Chief Financial Officer of Parent.
- (b) Agreements and Covenants. Parent, NAH and Merger Sub shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by them on or prior to the Closing Date, and the Company shall have received a certificate to such effect signed on behalf of Parent by the Chief Executive Officer or Chief Financial Officer of Parent.
- (c) Federal Tax Opinion. The Company shall have received the opinion of its counsel, Ballard Spahr Andrews & Ingersoll, LLP, in form and substance reasonably satisfactory to Parent, dated as of the Closing Date, substantially to the effect that, on the basis of facts, representations and assumptions set forth in such opinion that are consistent with the state of facts existing at the Effective Time, (i) the Offer and the Merger together will be treated as a reorganization within the meaning of Section 368(a) of the Code and (ii) the exchange of Company Common Stock for Parent Common Shares will not result in gain recognition to the shareholders of the Company pursuant to Section 367(a) of the Code. In rendering such opinion, counsel may require and rely upon such representations contained in certificates of officers of the Company, NAH and Parent and of their officers, directors and principal stockholders as are customary for such opinions. Such opinion of counsel may assume that any shareholder of the Company that is a U.S. person and a five percent transferee shareholder as defined in Treasury Regulation Section 1.367(a)-3(c)(5)(ii) will enter into a five-year gain recognition agreement described in Treasury Regulation Section 1.367(a)-8T.

(d) Directors and Officers Insurance. Parent shall have purchased the D&O Tail Coverage in accordance with Section 6.10.

A-58

Table of Contents

8. TERMINATION, AMENDMENT AND WAIVER

- 8.1 *Termination*. This Agreement may be terminated and the Offer, the Merger and other Transactions may be abandoned prior to the Effective Time upon written notice to the other Party, notwithstanding approval thereof by the stockholders of the Company:
- (a) by mutual written consent of the Parties;
- (b) by either Parent or the Company if the Merger shall not have been consummated on or before the date that is 270 days from the date hereof (the <u>Termination Date</u>); provided, however, that the right to terminate this Agreement under this Section 8.1(b) shall not be available to a Party whose failure to fulfill any material obligation under this Agreement has been the principal cause of, or resulted in, the failure of any condition of the Offer or the Merger to have been satisfied on or before such date;
- (c) by either Parent or the Company, if a Governmental Authority of competent jurisdiction shall have issued an order or taken any other action, in each case, which has become final and non-appealable and which permanently restrains, enjoins or otherwise prohibits the acceptance for payment or exchange of Company Common Stock pursuant to the Offer or the Merger;
- (d) by either Parent or the Company, if, (i) the Offer shall have expired or been terminated in accordance with the terms of this Agreement without Parent, NAH or Merger Sub having accepted for payment or exchange any shares of Company Common Stock pursuant to the Offer or (ii) at the Parent Stockholders Meeting the Parent Requisite Stockholder Approval shall not have been obtained;
- (e) by Parent if, at any time prior to the Acceptance Date (i) the Company Board (or a duly authorized committee of the Company Board) shall have withdrawn or modified the Recommendation in a manner adverse to Parent or fails to include its recommendation of this Agreement and the Merger in the Schedule 14D-9, (ii) the Company Board has failed to reaffirm the Recommendation within 3 Business Days after Parent has requested in writing that it do so at any time when a Competing Proposal shall have been publicly proposed and not rejected by the Company Board, provided that such 3 Business Day period shall be extended for 3 Business Days following any material modification of such Competing Proposal occurring after the receipt of Parent s written request, (iii) the Company Board (or a duly authorized committee of the Company Board) shall have recommended to the Company stockholders that they approve or accept a Competing Proposal or (iv) the Company shall have breached any of its obligations under Section 5.2(b), (c) or (d), or Section 6.2;
- (f) by Parent, if neither Parent, NAH nor Merger Sub is in material breach of its obligations under this Agreement, and if, at any time prior to the Acceptance Date, (i) there has been a breach of any of the representations and warranties of the Company herein, which breach would cause the condition set forth in Section 7.2(a) not to be satisfied, or (ii) there has been a breach on the part of the Company of any of its covenants or agreements contained in this Agreement, which breach would cause the condition in Section 7.2(b) not to be satisfied, and, in both case (i) and case (ii), such breach (if curable) has not been cured within 30 days after notice thereof to the Company by Parent, provided such cure period shall not extend beyond the period set forth in Section 8.1(b);
- (g) by the Company, if it is not in material breach of its obligations under this Agreement, and if, at any time prior to the Acceptance Date, (i) there has been a breach of any of the representations and warranties of Parent, NAH or Merger Sub herein, which breach would cause the condition set forth in Section 7.3(a) not to be satisfied, or (ii) there has been a breach on the part of Parent, NAH or Merger Sub of any of their respective covenants or agreements contained in this Agreement, which breach would cause the condition in Section 7.3(b) not to be satisfied, and, in both case (i) and case (ii), such breach (if curable) has not been cured within thirty (30) days after notice thereof to Parent from the Company, provided such cure period shall not extend beyond the period set forth in Section 8.1(b);
- (h) by the Company, if, at any time prior to the Acceptance Date, the Company Board shall have approved or recommended a Competing Proposal; provided, however, that this Agreement may not be so

A-59

Table of Contents

terminated unless (i) the Company Board shall have complied in all material respects with the procedures set forth in Sections 5.2(c) and (d) and (ii) all of the payments required by Section 8.3 as a result of the operation of this paragraph (h) have been made in full to Parent; or

- (i) by Parent or the Company, if with respect to antitrust matters, HSR approval has not been obtained before the expiration of 120 days after the date of the HSR filing; provided, however, that such 120 day period shall be extended for an additional 120 day period in the event that facts and circumstances existing at such time indicate that there is a reasonable possibility that HSR approval will be obtained within such additional 120 day period with the cooperation in good faith of both Parent and the Company in pursuing such approval.
- 8.2 Effect of Termination. Except as provided in this Section 8.2, in the event of the termination of this Agreement pursuant to Section 8.1, this Agreement (other than this Section 8.2 and Sections 6.3(b), 6.8, and 8.3, each of which shall survive such termination) will forthwith become void, and there will be no liability on the part of Parent, Merger Sub or the Company or any of their respective officers or directors to the other, and all rights and obligations of any Party will cease, except that nothing herein will relieve any Party from liability for any willful breach, prior to termination of this Agreement in accordance with its terms, of any representation, warranty, covenant or agreement contained in this Agreement.

8.3 Fees and Expenses.

- (a) Except as set forth in this Section 8.3 and in Sections 6.4(i) and 6.5, all fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the Party incurring such fees and expenses, whether or not the Offer or the Merger is consummated; provided, however, that Parent shall pay all fees and expenses, other than accountants and attorneys fees, incurred in relation to the printing, mailing and filing of the Proxy Statement (including financial statements and exhibits) and any amendments or supplements thereto and all filing fees payable in connection with filings made under any Antitrust Laws or any other Law.
- (b) In the event that Parent or the Company, as the case may be, terminates this Agreement pursuant to Section 8.1(b) or Section 8.1(d) following the public announcement of a Competing Proposal and within 12 months after the date of such termination the Company enters into an agreement with respect to any Competing Proposal or such Competing Proposal is consummated, then the Company shall pay the Termination Fee to Parent, simultaneously with the execution of the agreement with respect to such Competing Proposal, by wire transfer of immediately available funds to an account specified by Parent.
- (c) In the event that Parent terminates this Agreement pursuant to Section 8.1(e) or the Company terminates this Agreement pursuant to Section 8.1(h), then the Company shall pay to Parent, simultaneously with such termination of this Agreement, the Termination Fee, which Termination Fee shall be payable by wire transfer of immediately available funds to an account specified by Parent.
- (d) If this Agreement is terminated pursuant to Section 8.1(g), then Parent shall pay to the Company the Termination Fee by wire transfer of immediately available funds to an account specified by the Company not later than two Business Days after the effective date of such termination.
- (e) If this Agreement is terminated pursuant to Section 8.1(f), then the Company shall pay to Parent the Termination Fee by wire transfer of immediately available funds to an account specified by Parent not later than two Business Days after the effective date of such termination.
- (f) Payment of the fees and expenses described in this Section 8.3 shall constitute the sole and exclusive remedy of the parties in connection with any termination of this Agreement, other than as a result of any willful breach by any Party.

9. GENERAL PROVISIONS

9.1 *Amendment.* This Agreement may be amended by the Parties by action taken by or on behalf of their respective Boards of Directors at any time prior to the Closing; provided, however, that, after approval of the Merger by the stockholders of the Company, no amendment may be made which would require stockholder approval unless such approval is obtained. This Agreement may not be amended except by an instrument in writing signed by all of the Parties.

A-60

- 9.2 Waiver. At any time prior to the Closing, Parent, NAH and Merger Sub, on the one hand, and the Company, on the other hand, may extend the time for the performance of any of the other s obligations or other acts required hereunder, waive any inaccuracies in the other s representations and warranties contained herein or in any document delivered pursuant hereto, and waive compliance with any of the other s agreements or conditions contained herein. Any such extension or waiver shall be valid only if set forth in an instrument signed by the Party to be bound thereby.
- 9.3 Survival of Representations and Warranties. The representations and warranties in this Agreement shall terminate at the Closing; provided, however, this Section 9.3 shall in no way limit any covenant or agreement of the Parties which by its terms contemplates performance after the Closing.
- 9.4 *Notices*. All notices or other communications which are required or permitted hereunder shall be in writing and sufficient if delivered personally or sent by nationally-recognized overnight courier or by registered or certified mail, postage prepaid, return receipt requested, or by electronic mail, with a copy thereof to be delivered by mail (as aforesaid) within 24 hours of such electronic mail, or by telecopier, with a copy thereof to be delivered by mail (as aforesaid) within 24 hours of such telecopy, in each case addressed as follows:

(a)	If to Parent, NAH or Merger Sub:
Ì	
	QIAGEN North American Holdings, Inc.
	19300 Germantown Road
	Germantown, MD 20874
	Fax: (240) 686-7303
	Attention: Chief Financial Officer
	With copies to:
	Dr. Philipp von Hugo
	QIAGEN GmbH
	QIAGEN Strasse 1
	40724 Hilden
	Germany
	and
	Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C.
	One Financial Center
	Boston, MA 02111
	617 542-2241 fax
	Attention: Jonathan L. Kravetz, Esq.
(b)	If to the Company:
(0)	in to the company,
	DIGENE CORPORATION
	1201 Clopper Road
	Gaithersburg, Maryland 20878
	Attention: Chief Executive Officer
	With a copy to:
	Ballard Spahr Andrews & Ingersoll, LLP
	1735 Market Street, 51st Floor
	Philadelphia, PA 19103
	215 864-8999 fax
	Attention: Mary J. Mullany, Esq.

A-61

Table of Contents

or to such other address as the Party to whom notice is to be given may have furnished to the other Party in writing in accordance herewith. All such notices or communications shall be deemed to be received (a) in the case of personal delivery, on the date of such delivery, (b) in the case of nationally-recognized overnight courier, on the next Business Day after the date when sent, (c) in the case of facsimile transmission or telecopier or electronic mail, upon confirmed receipt, and (d) in the case of mailing, on the third Business Day following the date on which the piece of mail containing such communication was posted by registered or certified mail, postage prepaid, return receipt requested.

- 9.5 Interpretation. When a reference is made in this Agreement to Sections, subsections, Schedules or Exhibits, such reference shall be to a Section, subsection, Schedule or Exhibit to this Agreement unless otherwise indicated. The words include, includes and including when used herein shall be deemed in each case to be followed by the words without limitation. The word herein and similar references mean, except where a specific Section or Section reference is expressly indicated, the entire Agreement rather than any specific Section or Section. The table of contents and the headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. References in this Agreement to a contract or agreement mean such contract or agreement as amended or otherwise modified from time to time. References in this Agreement to a Law include any rules, regulations and delegated legislation issued thereunder.
- 9.6 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.
- 9.7 Entire Agreement. This Agreement (including all exhibits and schedules hereto), and other documents and instruments delivered in connection herewith constitute the entire agreement and supersede all prior agreements and undertakings (other than the Confidentiality Agreement), both written and oral, among the Parties with respect to the subject matter hereof.
- 9.8 Assignment. Neither this Agreement nor any of the rights, interests or obligations under this Agreement may be assigned by operation of Law or otherwise, except that Merger Sub may assign all or any of its rights hereunder to another wholly owned Subsidiary of Parent, provided that no such assignment shall relieve Parent, NAH or Merger Sub of its obligations hereunder. Subject to the preceding sentence, this Agreement shall be binding upon, and inure to the benefit of and be enforceable by, the parties hereto and their respective successors and assigns.
- 9.9 Parties in Interest. Except as set forth in Section 6.10(a) and Section 6.10(b), this Agreement is not intended, and shall not be deemed, to confer upon any person who is not a Party hereto any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 9.10 Failure or Indulgence Not Waiver; Remedies Cumulative. No failure or delay on the part of any Party in the exercise of any right hereunder will impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty or agreement herein, nor will any single or partial exercise of any such right preclude other or further exercise thereof or of any other right. All rights and remedies existing under this Agreement are cumulative to, and not exclusive to, and not exclusive of, any rights or remedies otherwise available.
- 9.11 Governing Law; Enforcement. This Agreement and the rights and duties of the Parties hereunder shall be governed by, and construed in accordance with, the law of the State of New York (except to the extent to which, under the internal affairs doctrine as applied under New York law, the DGCL applies by reason of being the Law of the Company s and/or Merger Sub s state of organization). The Parties agree that irreparable damage

A-62

Table of Contents

would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, this being in addition to any other remedy to which they are entitled at law or in equity. In addition, each of the Parties: (a) consents to submit itself to the exclusive personal jurisdiction of the U.S. District Court for the Southern District of New York, in the event of any dispute related to or arising out of this Agreement or any transaction contemplated hereby; (b) agrees not to commence any action, suit or proceeding related to or arising out of this Agreement or any transaction contemplated hereby except in such courts; (c) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court; (d) waives any right to trial by jury with respect to any action related to or arising out of this Agreement or any transaction contemplated hereby and (e) consents to service of process by delivery pursuant to Section 9.4 hereof.

- 9.12 *Counterparts*. This Agreement may be executed in one or more counterparts, and by the different Parties in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.
- 9.13 Definitions. For purposes of this Agreement, unless the context otherwise requires, the following terms shall have the following meanings:
- (a) Acceptance Date. Acceptance Date means the date on which Parent, NAH or Merger Sub, in compliance with Section 1.6(b), accepts for payment or exchange all shares of Company Common Stock validly tendered and not withdrawn pursuant to the Offer.
- (b) *Company Certificate*. Company Certificate means a certificate which previously represented any share or shares of Company Common Stock or shares of Company Common Stock in uncertificated form. Company Certificates shall have a correlative meaning.
- (c) Fully-Diluted Shares. Fully-Diluted Shares means (A) the shares of Company Common Stock then outstanding plus (B) the shares of Company Common Stock which the Company may be required to issue pursuant to Company Equity Awards then outstanding.
- (d) *Knowledge*. knowledge of the Company , and all variations thereof, means the actual knowledge of the individuals identified in Section 9.13 of the Company Disclosure Schedule or the knowledge that would reasonably be expected to be obtained upon appropriate inquiry with respect to such matter of any of such individuals. For the purposes of this Agreement, knowledge of Parent , and all variations thereof, means the actual knowledge of the individuals identified in Section 9.13 of the Parent Disclosure Schedule or the knowledge that would reasonably be expected to be obtained upon appropriate inquiry with respect to such matter of any of such individuals.
- (e) Law. Law means any federal, state, local or foreign statute, law, regulation, legal requirement, or rule, ordinance or code of any Governmental Authority (Law if in the singular or Laws if in the plural).
- (f) Merger Cash Pool shall mean the product of (x) the Per Share Cash Consideration multiplied by (y) the Merger Cash Election Number.
- (g) Minimum Condition. Minimum Condition means 50.1% of the Fully-Diluted Shares.
- (h) Offer Cash Pool. Offer Cash Pool means the product of (x) the Per Share Cash Consideration multiplied by (y) the Cash Election Number.
- (i) Termination Fee. Termination Fee means cash in an amount of Fifty-Nine Million Dollars (\$59,000,000).

A-63

(j) As used in this Agreement, the following terms shall have the respective meanings ascribed thereto in the respective sections of this Agreement set forth opposite each such term below:

Term	Section
Acceptance Date	9.13(a)
Acceptance Notice	1.1(b)
Acquisition Agreement	5.2(c)
AFM	4.4(c)
Agreement	Preamble
Anti-Kickback Statute	3.10(f)
Antitrust Law	6.4(d)
Articles of Association	4.1(b)
Assumed Equity Awards	2.18(b)
BaFin	4.4(c)
Bankruptcy and Equitable Exceptions	3.4(b)
Recommendation	3.5(a)
Business Day	2.2
Cancelled Equity Awards	2.18(a)
Cash Election	1.1(a)
Cash Election Number	1.5(a)
Cash Election Shares	1.5(b)(iv)
Cash Fraction	1.1(a)
Certificate of Formation	2.8
Certificate of Merger	2.3
CFC	4.14(e)
Closing	2.2
Closing Date	2.2
Closing Average	2.20
COBRA Coverage	3.14(d)
Code	3.14
Commitment Letter	4.22
Company	Preamble
Company Affiliate Agreement	6.13
Company Board	1.2(a)
Company Certificate	2.5
Company Common Stock	1.1(a)
Company Disclosure Schedule	3
Company Employee Plans	3.14
Company Equity Awards	3.3(b)
Company Equity Plans	3.3(b)
Company Financial Statements	3.6(b)
Company Intellectual Property Rights	3.21(a)
Company Material Adverse Effect	3
Company Material Contracts	3.9(a)
Company Preferred Stock	3.3(a)
Company Requisite Stockholder Approval	3.4(a)
Company SEC Reports	3.6(a)
Company Special Meeting	3.16(e)
Competing Proposal	5.2(a)
Confidentiality Agreement	6.3(b)
Consideration	1.1(a)
Continuing Directors	1.4(a)

Term	Section
Continuing Employees	6.6
Continuing Plans	6.6
Debarred	3.10(h)
DGCL	Recitals
Dissenting Shares	2.12
DLLCA	Recitals
DOJ	6.4(d)
ECMR	3.4(d)
Effective Time	2.3
Election Deadline	1.5(b)(ii)
Environmental Law	3.20(g)
ERISA	3.14
ERISA Affiliate	3.14
EU Prospectus Directive	6.1(a)
European Prospectus	6.1(a)
Excess Cash Election Shares	1.5(b)(iv)
Excess Merger Stock Election Shares	2.5(c)(iv)
Excess Merger Cash Election Shares	2.5(c)(iii)
Excess Stock Election Shares	1.5(b)(v)
Exchange Act	3.3(d)
Exchange Agent	1.6
Exchange Fund	1.6(b)
Excluded Shares	2.10
FDA	3.10(a)
FDA Law and Regulation	3.10(a)
FD&C Act	3.4(d)
FD&C Act Permits	3.11
Financing	6.4(i)
FTC	6.4(d)
FMSA	4.4(c)
Form of Election	1.5(b)(ii)
Fully-Diluted Shares	9.13(b)
GAAP	3.6(b)
Governmental Authority	3.4(d)
HIPAA	3.10(n)
HSR Act	3.4(d)
IFRS	6.5
Indemnified Parties	6.10(b)
Insurance Policies	3.18(a)
Interim Period	5.1 5.1
Internal Investigation	3.12
IRS	3.14
knowledge of Parent	9.13
knowledge of the Company	9.13
Law	9.13(e)
Leases	3.17(b)
Liens	3.2(c)
Material Permits	3.11
Materials of Environmental Concern	3.20(g)
Maximum Cash Election Number	1.1(a)
Manager Cash Ecotor Number	1.1(a)

Term	Section
Merger	Recitals
Merger Cash Election	2.5(a)
Merger Cash Election Number	2.5(b)
Merger Cash Election Shares	2.5(c)(iii)
Merger Cash Pool	2.5(b)
Merger Election Deadline	2.5(c)(i)
Merger Exchange Fund	2.6(a)
Merger Form of Election	2.5(c)(i)
Merger Non-Election Shares	2.5(a)
Merger Registration Statement	4.12(e)
Merger Stock Election	2.5(a)
Merger Stock Election Shares	2.5(c)(iii)
Merger Stock Election Number	2.5(b)
Merger Sub	Preamble
Minimum Condition	9.13(g)
Most Recent Balance Sheet	3.7
Most Recent Balance Sheet Date	3.7
NAH	Preamble
NGSM	2.20
Non-Election	1.5(b)(i)
Non-Election Shares	1.5(b)(i)
Offer	Recitals
Offer Documents	1.1(c)
Offer Registration Statement	1.1(c)
Operating Agreement	2.8
Option Conversion Ratio	2.18(c)
Other Filings	6.1(e)
Parent	Preamble
Parent Circular	6.1(b)
Parent Common Shares	1.1(a)
Parent Convertible Note Shares	4.3(b)
Parent Disclosure Schedule	4
Parent Financial Statements	4.5(b)
Parent Financing Preference Shares	4.3(a)
Parent Intellectual Property Rights	4.16(a)
Parent Material Adverse Effect	4
Parent Material Permits	4.9
Parent Plans	6.6
Parent Policy	6.6
Parent Preference Shares	4.3(a)
Parent Requisite Stockholder Approval	4.4(a)
Parent SEC Reports	4.5(a)
Parent Stockholders Meeting	6.2(b)
Parent Third Party Intellectual	4.16(c)
Property Rights	
Parent Warrant Shares	4.3(b)
Parent s Most Recent Balance Sheet	4.5(c)
Parent s Most Recent Balance Sheet Date	4.5(c)
Parent Option/Award Shares	4.3(b)
Parties	Preamble

Term	Section	
Party	Preamble	
Per Share Cash Consideration	1.1(a)	
Per Share Stock Consideration	1.1(a)	
PFIC	4.14(f)	
Proxy Statement	3.16(e)	
Recommendation	1.2(a)	
reporting tail coverage	6.10(a)	
Representative	5.2(a)	
Returns	3.19(b)	
Schedule 14D-9	1.2(b)	
Schedule TO	1.1(c)	
SEC	3.2(d)	
Securities Act	3.2(c)	
SOX	3.6(c)	
SPAQ	4.3(h)	
Stock Election	1.1(a)	
Stock Election Number	1.5(a)	
Stock Election Shares	1.5(b)(iv)	
Subsidiary	3.2(e)	
Superior Competing Proposal	5.2(b)	
Surviving Company	2.1	
Tax	3.19(a)	
Termination Fee	9.13(i)	
Third Party	5.2(a)	
Third Party Intellectual Property Rights	3.21(d)	
Third Party Manufacturer	3.10(i)	
Top-Up Closing	1.3(c)	
Top-Up Exercise Event	1.3(b)	
Top-Up Exercise Notice	1.3(c)	
Top-Up Notice Date	1.3(c)	
Top-Up Option	1.3(a)	
Top-Up Option Shares	1.3(a)	
Top-Up Termination Date	1.3(b)	
Transactions		
WARN Act	3.15(d)	

(REMAINDER OF PAGE INTENTIONALLY LEFT BLANK)

A-67

IN WITNESS WHEREOF, Parent, NAH, Merger Sub and the Company have caused this Agreement and Plan of Merger to be executed as of the date first written above by their respective officers thereunto duly authorized.

QIAGEN N.V.

By: /s/ PEER M. SCHATZ
Name: Peer M. Schatz

Title: President and Chief Executive Officer

QIAGEN NORTH AMERICAN HOLDINGS, INC.

By: /s/ PEER M. SCHATZ
Name: Peer M. Schatz

Title: President and Chief Executive Officer

QIAGEN MERGER SUB, LLC

By: Qiagen North American Holdings, Inc., its Sole

Member

By: /s/ PEER M. SCHATZ
Name: Peer M. Schatz

Title: President and Chief Executive Officer

DIGENE CORPORATION

By: /s/ Daryl Faulkner
Name: Daryl Faulkner

Title: President and Chief Executive Officer

A-68

ANNEX I

CONDITIONS TO THE OFFER

The capitalized terms used in this <u>Annex I</u> have the meanings set forth in the attached Agreement and Plan of Merger, and the term Agreement shall be deemed to refer to the attached Agreement and Plan of Merger.

Notwithstanding any other provision of the Offer, Parent, NAH or Merger Sub, as applicable, shall not be required to accept for payment or, subject to any applicable rules and regulations of the SEC, including Rule 14e-l(c) under the Exchange Act (relating to Parent s, NAH s or Merger Sub s, as applicable, obligation to pay for or return tendered Shares promptly after termination or withdrawal of the Offer), pay for, and may delay the acceptance for payment of or, subject to the restriction referred to above, the payment for, any tendered shares of Company Common Stock (the Shares), and (subject to the provisions of the Agreement) may terminate the Offer and not accept for payment any tendered shares if (a) the Minimum Condition shall not have been satisfied at the expiration of the Offer, (b) any applicable waiting period under the HSR Act shall not have expired or been terminated prior to the expiration of the Offer, (c) the Offer Registration Statement shall not have become effective under the Securities Act prior to the expiration of the Offer or shall be the subject of any stop order or proceedings seeking a stop order at the expiration of the Offer, (d) the Parent Common Shares issuable in exchange for the Shares in the Offer shall not have been approved (if such approval is necessary), prior to the expiration of the Offer, for listing on the NGSM (subject to official notice of issuance), (e) the Company shall not have received, prior to first date on which Parent accepts for payment all Shares validly tendered and not withdrawn pursuant to the Offer, a written opinion of Ballard Spahr Andrews & Ingersoll, LLP (or other nationally recognized tax counsel reasonably acceptable to the Company) to the effect that the Offer and the Merger together will constitute a reorganization within the meaning of Section 368(a) of the Code (the <u>Company 368 Opinion</u>) (which opinion may rely on such assumptions and representations as such counsel reasonably deems appropriate), such Company 368 Opinion shall have been withdrawn or an event shall have occurred that prevents Company from relying on such Company 368 Opinion; provided that such condition may be waived by the Company in its sole discretion, (f) Parent shall not have received, prior to first date on which Parent accepts for payment all Shares validly tendered and not withdrawn pursuant to the Offer, a written opinion of Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C. (or other nationally recognized tax counsel reasonably acceptable to Parent) to the effect that the Offer and the Merger together will constitute a reorganization within the meaning of Section 368(a) of the Code (the Parent 368 Opinion) (which opinion may rely on such assumptions and representations as such counsel reasonably deems appropriate), such Parent 368 Opinion shall have been withdrawn or an event shall have occurred that prevents Parent from relying on such Parent 368 Opinion, (g) Parent shall not have received copies of the material consents identified in Section 7.2(c) of the Company Disclosure Schedule, (h) the Agreement shall not have been adopted by the Parent Requisite Stockholder Approval, (i) the Agreement shall have been terminated in accordance with its terms or the Offer shall have been terminated with the consent of the Company or (j) at any time on or after the date of the Agreement and prior to the expiration of the Offer, any of the following conditions shall exist:

(i) there shall be any injunction, judgment, ruling, order or decree issued or entered by any Governmental Authority that (A) restrains, enjoins, prevents, prohibits or makes illegal the making of the Offer by Parent, NAH or Merger Sub, as applicable, the acceptance for payment, payment for or purchase of some or all of the Shares by Parent, NAH or Merger Sub, as applicable, or the consummation of the transactions contemplated by the Agreement, (B) imposes material limitations on the ability of Parent or any of its affiliates effectively to exercise full rights of ownership of one hundred percent (100%) of the Shares, including, without limitation, the right to vote the Shares purchased by them on all matters properly presented to the Company s stockholders on an equal basis with all other stockholders (including, without limitation, the adoption of the Agreement and approval of the Merger), (C) restrains, enjoins, prevents, prohibits or makes illegal, or imposes material limitations on, Parent s or any of its affiliates ownership or operation of all or any portion of the businesses and assets of the Company and its Subsidiaries, or, as a result of the transactions contemplated by the Agreement, of Parent and its subsidiaries, (D) compels Parent

A-69

Table of Contents

or any of its affiliates to dispose of any Shares or, as a result of the transactions contemplated by the Agreement, compels Parent or any of its affiliates to hold separate any portion of the businesses or assets of the Company and its Subsidiaries, or of Parent and its subsidiaries or (E) imposes damages on Parent, the Company or any of their respective affiliates as a result of the transactions contemplated by the Agreements in amounts that are material with respect to such transactions;

- (ii) there shall be any Law enacted, issued, promulgated, amended or enforced by any Governmental Authority applicable to (A) Parent, the Company or any of their respective affiliates or (B) the transactions contemplated by the Agreement (other than the routine application of the waiting period provisions of the HSR Act) that results, or is reasonably likely to result, directly or indirectly, in any of the consequences referred to in paragraph (i) above;
- (iii)(A) there shall have occurred any events or changes that, individually or in the aggregate, have had or would reasonably be expected to have a Company Material Adverse Effect or (B) (1) the Company shall have breached or failed to perform in any material respect its obligations, covenants or agreements under the Agreement, (2) the representations and warranties of the Company set forth in Section 3.3 shall not have been true and correct in all material respects when made or as of the Acceptance Date as if made at or at and as of such time or (3) the other representations and warranties of the Company contained in the Agreement shall not have been true and correct (disregarding all qualifications and exceptions contained therein regarding materiality or a Company Material Adverse Effect or any similar standard or qualification) when made or as of the Acceptance Date as if made at or as of such time (other than representations and warranties which by their terms address matters only as of another specified date, which shall be true and correct only as of such date) except in the case of this clause (3) only, for such inaccuracies as have not resulted, or are not reasonably likely to result, individually or in the aggregate, in a Company Material Adverse Effect;
- (iv) the Company Board shall have withdrawn or modified, in any manner (other than non-substantive modifications), its approval or recommendation of the Offer or the Merger; or
- (v) there shall have occurred (A) any general suspension of trading in securities on the New York Stock Exchange, the American Stock Exchange or on the NASDAQ, for a period in excess of twenty-four (24) hours (excluding suspensions or limitations resulting solely from physical damage or interference with such exchanges not related to market conditions), (B) a declaration of a banking moratorium or any suspension of payments in respect of banks in the United States (whether or not mandatory), (C) any limitation or proposed limitation (whether or not mandatory) by any United States Governmental Authority that has a material adverse effect generally on the extension of credit by banks or other financial institutions, (D) the commencement of a war involving the United States that, in the reasonable judgment of Parent, materially affects Parent, the Company, Parent s ability to consummate the Offer or materially adversely affects securities markets in the United States generally or (E) in the case of any of the situations in clauses (A) through (D) of this paragraph existing at the time of the commencement of the Offer, a material acceleration or worsening thereof.

The foregoing conditions are for the sole benefit of Parent, NAH and Merger Sub and may be asserted by Parent, NAH or Merger Sub regardless of the circumstances giving rise to such conditions or may be waived by Parent, NAH or Merger Sub, in whole or in part at any time and from time to time in the sole discretion (except as provided in Section 1.1(b)(y) of the Agreement) of Parent, NAH and Merger Sub. The failure by Parent, NAH or Merger Sub at any time to exercise any of the foregoing rights will not be deemed a waiver of any right, the waiver of such right with respect to any particular facts or circumstances shall not be deemed a waiver with respect to any other facts or circumstances, and each right will be deemed an ongoing right which may be asserted at any time and from time to time prior to the expiration of the Offer.

If the Offer is terminated, all tendered Shares not theretofore accepted for payment shall forthwith be promptly returned to the tendering stockholders.

A-70